NASA CR-159722 GDC-CRAD-80-001

ORBITAL REFILL OF PROPULSION VEHICLE TANKAGE

N80-18076

ł,

「日田二日のかいたい」というという

1

(NASA-CR-159722) ORBITAL REFILL OF PROPULSION VEHICLE TANKAGE (General Dynamics/Convair) 256 p HC A12/MF A01

CSCL 22A Unclas G3/12 47310

GENERAL DYNAMICS

Convair Division



NASA CR-159722 GDC-CRAD-80-001

ORBITAL REFILL OF PROPULSION VEHICLE TANKAGE

February 1980

Prepared by F. Merino J.A. Risberg M. Hill

Prepared for National Aeronautics and Space Administration LEWIS RESEARCH CENTER 21000 Brookpark Road Cleveland, Ohio 44135

> Prepared Under Contract NAS3-21360

Prepared by GENERAL DYNAMICS CONVAIR DIVISION P.O. Bo), 80847 San Diego, California 92138

FOREWORD

The following final report summarizes the technical effort conducted under Contract NAS3-21360 by the General Dynamics Convair Division from May, 1978 to September 1979 — The contract was administered by the National Aeronautics and Space Administration, Lewis Research Center, Cloveland, Ohio,

NASA LOR	C Program Manager - F. P. Symons
Convate Pr	ogvani Manager – F. Mertno
Assisting	J. Risberg, Modification of HVPRS Computer Program M. Hill, Modeling Analysis

All data are presented with the International System of Units as the primary system and English Units as the secondary system. The English system was used for the basic calculations.

ι

TABLE OF CONTENTS

ά.

Section			Page
1	INTROD	DUCTION	1-1
	1.1	SCOPE	1-1
	1.2	GROUNDRULES	1-2
	1.2.1	Earth Storable Vehicle	1-2
	1.2.2	Cryogenic Vehicle	1-2
	1.2.3	Experiment Modeling	1-2
2	IDENTI	FICATION OF CANDIDATE RECEIVERS	2-1
	2.1	LITERATURE REVIEW	2-2
	2.2	IDENTIFICATION OF FLUID PARAMETERS AND	
		TANK GEOMETRY	2-3
	2.3	CATEGORIZATION	2-4
	2.4	BASE LINE CANDIDATE VEHICLES	2-5
	2.5	VEHICLE RECOMMENDED FOR ANALYSIS	2-5
	2.5.1	Class A-Earth Storable Vehicle With Partial	
		Acquisition Device	2-6
	2.5.2	Class B-Cryogenic Vehicle With Partial Acquisition	
		Device	2-6
	2.5.3	Class C-Cryogenic Vehicle Without Acquisition Device .	2-8
3	POTV O	RBITAL RESUPPLY	3-1
	3.1	MISSION SCENARIO	3-1
	3.1.1	Selected POTV Mission	3-2
	3.1.1.1	Timelines	3-2
	3.1.2	Orbiter Tanker Configuration	3-4
	3.1.3	POTV Configuration	3-5
	3.1.3.1	Subsystems Influenced by Mission Requirements	3-6
	3.1.3.2	Subsystems Influenced by Space-Basing Requirements	3-7
	3.2	ORBITAL PROPELLANT RESUPPLY TECHNIQUES	3-11
	3.2.1	On-Orbit Resupply Concepts	3-11
	3.3	SELECTED ORBITAL RESUPPLY METHOD	3-15
	3.3.1	Initial Vent	3-15
	3.3.1.1	Propellant Tank Helium Dilution	3-15
	3.3.1.2	Peak Pressure Reduction	3-17
	3.3.2	Receiver Tank Prechill	3-17
	3.3.2.1	Tank Over-Pressure	3- 21
	3.3.2.2	Modelling	3-23
	3.3.2.3	Liquid Venting	3-23
PRECEDING CAR		iii	
	•••	CTT - CM Syn (C)	

..

TABLE OF CONTENTS (CONT'D)

...

Section			Page
	3 3 2 4	Terminating Prechill	3-24
	3.2.2.5	Prochill Analysis	3-25
	3 3 2 6	Summary	3-21
	3.3.3	Receiver Tank Fill	3-31
	3.3.3.1	Tank Refill (Autogenous)	3-36
	3.3.3.2	Tank Fill Analysis	3-37
	3.3.3.3	Supply Tank Influence	3-47
	3.3.3.4	Alternative Refill Concept	3-48
	3.3.5	Mechanical Mixers to Assist Propellant Refill	3-52
	3.3.5.1	Mixer Power Relationship to Liquid-Ullage Heat Exchange	352
	3.3.5.2	Mixer Power/Fluid Power Equivalence	3-52
	3.3.6	Start Basket Refill	3-55
	3.3.6.1	Passive Method of Bubble Collapse	3-55
	3.3.6.2	Active Method of Bubble Collapse	3-58
	3.3.6.3	Summary	3-66
	3.4	ORBITAL PROPELLANT TANKING OPERATIONS	3-67
	3.4.1	Conceptual Design Modifications for On-Orbit Refill	3-67
	3.4.1.1	Propellant Tank Modifications	3-67
	3.4.1.2	Transfer Line Design	3-70
	3.4.1.3	Helium System Design	3-71
	3.4.2	Orbital Propellant Tanking Operations	3-74
	3.4.2.1	Subsystem Influence Upon Refill Procedures	3-74
	3.4.2.2	Shuttle Flight Influence Upon Refill Procedure	3-74
	3.4.2.3	Tank Fill Procedures	3-75
	3.4.3	Zero-G Mass Gauging	3-76
	3.4.3.1	Current Mass Gauging Devices	3-76
	3.4.3.2	Thermodynamic Mass Gauging	. 3-80
4	COTV OI	RBITAL RESUPPLY	4-1
	4 1	MISSION SCENARIO	4-1
	4.1 1	Selected COTV Missions	4-1
	4.1.1.1	Timolinos	4-2
	4.1.1.1	O-kital Danat Configuration	4-2
	4.1.2	Andillar Equipment	4-2
	4.1.2.1	COTV Configuration	4-4
	4.1.0	Advanced Attitude Control System	4-5
	4.1.0.1	Advanced Main Engine	4-5
	4 1 9 9	COTV /DOTV Procedural Differences	4-5
	4.1.J.J 4 2	POST MISSION DE-TANKING OPERATIONS	4-6

TABLE OF CONTENTS (CONT'D)

۰,

Section			Page
	4.2.1	Operations for Autogenous Pressurant	4-6
	4.2.1.1	The Cost of Propellant Dumping	4-7
	4.2.1.2	Propellant Reclamination	4-8
	4.2.1.3	Residuals for RCS Propellants	4-10
	4.2.2	Operations for Helium Pressurant	4-13
	4.2.2.1	The Cost of Propellant Dumping	4-13
	4.2.2.2	Propellant Reclamation	4-16
	4.2.2.3	Residual for RCS Propellants	4-16
	4.3	COTV ON-ORBIT RESUPPLY	4-17
	4.3.1	COTV Prechill · · · · · · · · · · · · · · · · · ·	4-18
	4.3.1.1	Prechill Procedures	4-18
	4.3.2	COTV Tank Fill	4-21
5	LTL OR	BITAL RESUPPLY	5-1
	5.1	MISSION SCENARIO	5-1
	5.1.1	Selected LTL Mission	5-1
	5.1.2	LTL Configuration	5-1
	5.1.2.1	Main Propulsion Tankage	5-1
	5.1.2.2	Reduction Control System	5-3
	5.1.3	LTL Vehicle Concepts	5-3
	5.1.3.1	Vehicle Concept One	5-3
	5.1.3.2	Vehicle Concept 2	5-6
	5.1.3.3	Fluid Systems for Concept 1	5-8
	5.2	ORBITAL PROPELLANT RESUPPLY TECHNIQUES	5-11
	5.2.1	Propellant Tank Refill Requirements.	5-12
	5.2.1.1	Vent Propellant Tanks Prior to Orbiter Rendezvous	5-12
	5.2.1.2	Minimize Liquid Vent Potential	5-12
	5.2.1.3	Helium Must Not Enter Screen Galleries	5-12
	5.2.2	Initial Fill	5-12
	5.2.2.1	Non-Equilibrium Fill	5-12
	5.2.2.2	Thermal Equilibrium Tank Fill	5-13
	5.2.3	On-Orbit Refill	5-14
	5.2.3.1	Thermodynamics of Propellant Tank Vent	5-15
	5.2.3.2	Propellant Tank Refill	5-17
	5.3	HELIUM VENTING	5-17
	5.3.1	Alternative Vent Procedures	5-18
	5.3.2	Selected Vent Procedure.	5-21
	5.4	PROPELLANT REFILL PROCEDURE.	5-23
	5.4.1	Earth Storable Propellant Disconnect	5-23
	5.4.2	Helium Bottle Resupply	5-28
	5.4.2.1	Helium Transfer from Orbiter ••••••••	5-28

۱.

TABLE OF CONTENTS (CONT'D)

Section

J

ι.

	5.4.2.2 Helium Modules	
	5.4.3 Zero-G Mass Gauging	5-28
	and and me	5-31
6	EXPERIMENTAL MODELING	
		6-1
	6.1 RECEIVER TANK SCALE	
	6.1.1 Receiver Tank Shape	6-2
	6.1.2 Test Scale	6-3
	6.1.2.1 Preliminary Test Tank Design	6-3
	6.1.2.2 Larger Test Tank Design	6-3
	6.2 PRECHILL MODELING	6-10
	6.2.1 Scaling Peak Droggunge	6-13
	6.2.1.1 Model Tank Size Influence	6-14
	6.2.2 Time Scaling	6-15
	6.2.3 Fluid Substitute	6-17
	6.2.4 Predicted Prophill Test Value	6-22
	6.2.4.1 Zero-G Tost Environment A	ნ-24
	6.2.4.2 One-G Test Environment Limitations	6-24
	6.2.5 Prechill Summer:	6-26
	6.3 TANK FULL MODELLING	6-26
	6.3.1 Vapor Bubble Density (27)	6-29
	6.3.1.1 Initial Fluid Tempore 1	6-29
	6.3.1.2 Fourilibrium Trans	6-31
	6.3.1.3 Dimonsionless mining	€-31
	6.3.1.4 Mixer Demon (Pl to a	6-33
	6.3.1.5 Model Texts Ut (25.5.5)	6-40
	6.3.2 Eluid Substation	6-43
	6.3.3 One C T + P	6-44
	6.3.4 Stort Del 4 Denis Comment Limitions	6-46
	6.4 SDACDI AD DEST Considerations	6-47
	6.5 MODELLAB EXPERIMENT INTEGRATION	6-48
7	MODELLING OF LTL REFUELING OPERATIONS	6-49
1 A mu a - 11	REFERENCES	- 10
Appendix		7-1
Α	Identification of Condidate William	
В	Spacecraft Accommodation	A-1
С	Distribution List	B-1
		C-1

LIST OF FIGURES

).): •

Figure		
2, 1		Page
2-1	Three Vehicle Configurations Were Selected With Concurrence from NASA/LeRC	9 0
2-2	A Straightforward Review of all Upper Stage Concepts was Employed to Yield Representative Study-Candidates	2-2 '} 0
3-1	Orbiter Tanker Configuration	2-J
3-2	Operations for 5-Day Manned GEO Sortie Mission	3-1
3-3	Timeline for 5-Day Manned GEO Sortie Mission (One Orbiter Tanker	J- 2
3-4	Tanker Flight 1 and 2 Operations Timeline	3-3
3-5	Features of An Orbiter Tanker Kit	3-4
3-6	A Representative POTV Was Selected for this Study	3-5
3-7	Influence of MLI External Shield Radiative Properties and Orientation Upon Propellant Tank Equilibrium Temperature	3-8
3-8	Time for POTV Liquid Residuals to Boiloff in LEO	3-8
3-9	Transient Time for POTV LO ₂ Tank to Attain Temperature Equilibrium in LEO	
3-10	Transient Time for POTV LH ₂ Tank to Attain Temperature Equilibrium in LEO)- y
3-11	Liquid Oxygen Tank Helium Partial Pressure Following Refueling Operation	-10
3-12	Liquid Hydrogen Tank Helium Partial Pressure Following Refueling Operation	-10

allander de la constante de la La constante de la constante de

Figure		Page
3-13	Oxygen Tank Blowdown for Helium Expulsion	3-1 8
3-14	POTV LH ₂ Tank Pressures Could Exceed Tank Allowables During Prechill •••••••••••••••••••••••••••••••••••	3-22
3-15	Precision Metering of LH ₂ Is Not Needed to Avoid Over-Pressure During Prechill	3-22
3-16	A Prechill Procedure Can Be Identified to Eliminate Excessive Tank Pressures Due to Wall Boiling	3-23
3-17	Tank Pressure Increases Will Yield Total Energy Removal DuringPrechill	3-26
3-18	Hydrogen Tank Prechill Vent Mass Is Not Excessive Even at Large Tank Wall to Ullage Temperature Differences •••••••••	3-26
3-19	POTV Liquid Hydrogen Tank Pressure History During Prechill •••	3-29
3-20	POTV Liquid Hydrogen Tank Temperature History During Prechill	• 3-29
3-21	Mass Flow Rate and Velocity Influence Upon Liquid Hydrogen Tank Prechill Duration	• 3-30
22	Influence of Tank 11-to-Vapor Temperature Difference Upon Prechill Duration	• 3-30
3-23	Charge Mass Influence Upon Liquid Hydrogen Tank Peak Pressures	. 3-32
3-24	Final LH ₂ Tank Pressures for Thermodynamic Equilibrium Fill Process (POTV and COTV Tanks)	• 3-35
3-25	Final LO ₂ Tank Pressures for Thermodynamic Equilibrium Fill Process (POTV and COTV Tanks)	• 3 -3 5

۰.

.

Figure		Page
3 -26	Entering Liquid Hydrogen Vapor Pressure Required to Maintain a Constant Vapor Pressure in Tank During Fill	3-38
3-27	Flowrate Influences Tank Pressure During Fill	5-39
3-28	Influence of Liquid Spray Droplet Diameter Upon Liquid Hydrogen Tank During Fill	3-40
3-29	Influence of Liquid Spray Volume in Ullage Upon Liquid Hydrogen Tank Pressure During Fill	3-40
3-30	Mass Flow Rate Influence Upon Liquid Hydrogen Tank Pressure During Fill	3-41
3-31	Required Average Ullage-to-Liquid Heat Transfer Rate for Liquid Hydrogen Tank Refueling Operation	3-43
3-32	Influence of Vapor-Bubble Heat Exchange Mechanism Upon Hydrogen Tank Fill Pressures for Range of Liquid Spray Droplet Diameter	3-46
3-33	Influence of Vapor-Bubble Heat Exchange Mechanism Upon Hydrogen Tank Fill Pressures for Range of Liquid Spray Droplet Diameter .	3-46
3-34	Influence of Vapor Bubble Heat Exchange Mechanism Upon Hydrogen Tank Fill Pressures for Range of Tanking Flow Rates	3-47
3-3 5	Supply Tank Liquid Temperature During POTV Refill	3-49
3-36	Supply Tank Liquid Vapor Pressure During POTV Refill	3-49
3-37	Supply Tank LH ₂ Temperature Influence Upon POTV Tank Pressure During Refill	3-50

•/

.

•

<u>Figure</u>		81. t
3-38	Mixer Power Influence Upon Entrained Vapor-to-Liquid Hydrogen Heat Transfer Rate	
3-39	Mixer Power Influence Upon Entrained Vapor-to-Liquid Oxygen Heat Transfer Rate	3-53
3-40	Mixer Power Influence Upon Hydrogen Bubble Diameter During Tank Fill	3-54
3- 11	Mixer Power Influence Upon Oxygen Bubble Diameter During Tank Fill	3-54
3-42	Fluid Power Input Equivalence to Mixer Power During OTV LH ₂ Tank Fill	3-56
3-43	Fluid Power Input Equivalence to Mixer Power During OTV LO ₂ Tank Fill	3-56
3-44	Start Basket Schematic	3-57
3-45	Collapse Times for Spherical Bubbles in Liquid Hydrogen	3-59
3-46	Collapse Times for Spherical Bubbles in Liquid Oxygen	3-59
3-47	Active Method for Start Basket Bubble Collapse	3-60
3- 48	Hydrogen Vapor Bubble Diameter During Start Basket Refill • • • • •	3-61
3-49	Oxygen Vapor Bubble Diameter During Start Basket Refill	3-61
3- 50	Maximum Allowable Bubble Diameter for Condensing Hydrogen Vapor in Start Basket During POTV Propellant Tank Fill	3-62
3-51	Maximum Allowable Bubble Diameter for Condensing Oxygen Vapor in Start Basket During POTV Propellant Tank Fill	3-62

Figure		Page
3-52	Influence of Liquid Subcooling Upon Start Basket Refill Flow Parameters	- 450
3-53	Dropollant pl	3-65
0.00	Fill Operations	3-66
3-54	POTV Tankage Systems	3-68
3-55	POTV Tankage Systems	3-69
3-56	Typical Vehicle Propellant Disconnect Arrangement	3-71
3-57	Modular Pressurization System (Applicable to POTV, COTV and LTL Vehicle)	3-73
3- 58	Propellant Tank Conditions for Thermodynamic Mass Gauging Operations	2_ 01
3-59	Thermodynamic Mass Gauging Tanking Error	3-84
3-60	Final Helium Partial Pressures Resulting from LH ₂ Tank Mass Gauging Operations at 80 Percent Fill	3-85
3-61	Final Helium Partial Pressures Resulting from LH ₂ Tank Mass Gauging Operations at 88 Percent Fill • • • • • • • • • • • • • • • • • •	9.05
4-1	Orbital Propellant Depot	4-3
4-2	Schematic: ET Propellant Depot	4-3
4-3	COTV Characteristics	1
4-4	Cost of Replacing Dumped COTV Residuals	⊷ . ⊬7
4-5	Alternative Methods of Reclaiming COTV Residual Propellants During Vehicle Post Mission Operations	9

•

Figure		Page
4-6	Reliquefier Capacity-Time Requirements for Reliquefying COTV Hydrogen Residuals	• 4-10
4-7	Cost of Dumping CCTV Hydrogen Residuals Compared to Cost of Reclaiming Vapor Residual	4-11
4-8	Estimated Vapor Feed Performance for Oxygen/Hydrogen RCS Engines	4-11
4-9	Schematic of a Propellant Residual Reclamation Process	4-17
4-10	COTV and POTV Peak Prechill Pressures Will Be The Same	4-18
4-11	COTV Liquid Hydrogen Tank Pressure History During Prechill .	4-19
4-12	COTV Liquid Hydrogen Tank Temperature History During Prechili	4-20
4-13	Excellent Correlation Exists Between COTV and POTV LH ₂ Tank Prechill Predictions	4-20
4- 14	Supply Tank Temperature Influence Upon COTV Pressure During Refill	4-22
4-15	Predicted Supply Tank Liquid Temperature During COTV Refill.	i -23
4-16	Predicted Supply Tank Liquid Vapor Pressure During COTV Refill	4-23
5-1	LTL Earth Storables Vehicle Configuration	5-2
5-2	OMS Propellant Tank	5-4
5-3	Low Thrust Liquid (LTL) Vehicle Concept One	5-5

-

.

<u>rure</u>	- <u>P</u> :	age
5-4	Low Thrust Liquid (LTL) Vehicle Concept 2 5	5-7
5-5	LTL Concept 1 Fluid System Plumbing Arrangement	-5
5-6	LTL Concept 1 Fluid Systems Schematic •••••• 5	-10
5-7	Maximum OMS Propellant Tank Pressure During Initial Fill With N_2O_4 · · · · · · · · · · · · · · · · · · ·	-13
5-8	Final Storage Tank Pressures for N ₂ O ₄ Thermodynamic Equilibrium Fill Process 5	-15
5-9	Propellant Tank Venting Can Occur Without Losing Liquid From Screen Galleries	-16
5-10	OMS Propellant Tank Pressures During Refill	-17
5-11	Expansion Angle of Ideal N ₂ O ₄ -Helium and MMH-Helium Mixtures as a Function of Helium Partial Pressure	-19
5-12	Helium Recovery System 5-	-20
5-13	Earth Storable Propellant Disconnect Valve	-25
5-14	Operation Sequence of Earth Storable Propellant Disconnect Valve 5-	- <u>30</u>
6-1	Orbital Refill Experiment Installation in Spacelab Standard Double Rack	·4
6-2	Test Tank Design Details	6
6-3	Test Tank Design Details (0.2 Scale)6-	11
6-4	Test Tank Design Details (0.3 Scale)	12

.

Figure	Page
6-5	A Prechill Procedure Can Be Identified to Eliminate Excessive Tank Pressures Due to Wall Boiling
6-6	Maximum Pressure During OTV Propellant Tank Prechill ••••••6-14
6-7	PV, M is a Parameter for Scaling Peak Pressures During Prechill+ 6-15
6-8	Test Tank Scale Influence on V*, M*
6-9	Peak Prechill Pressures May Be Excessive for Small Scale Experiments
6-10	(hA. m Cp)t is Applicable as Time Scale Parameter for Prechill Process
6-11	Predicted LH ₂ OTV Tank Temperature Histories from "HYPRES" Computer Simulation of 0.108 Scale Model Prechill Tests 6-25
6-12	Predicted LH ₂ OTV Tank Pressure Histories from "HYPRES" Computer Simulation of 0,108 Scale Model Prechill Tests 6-27
6-13	A Normal Gravity Environment will Influence OTV Model Tank Prechill Test Results
6-14	(hA $ au$ m Cp) is Applicable as Time Scale Parameter for Tank Fill Process
6-15	Average Vapor to Liquid Heating Rate Needed to Achieve Thermal Equilibrium
6-16	Mixer Power Influence Upon Entrained Vapor to Liquid Hydrogen Heat Transfer Rate
6-17	Mixer Power Influence Upon Entrained Vapor to Liquid Oxygen Heat Transfer Rate

•

Figure		Page
6-18	Mixer Power Influence Upon Hydrogen Bubble Diameter During Tank Fill	6-41
6-19	Mixer Power Influence Upon Oxygen Bubble Diameter During Tank Fill	6-41
6-20	Fluid Power Input Equivalence to Mixer Power During OTV LH ₂ Tank Fill	6-42
6-21	Fluid Power Input Equivalence to Mixer Power During OTV 1.0_2 Tank Fill $\dots \dots $	6-42
6-22	V*/M* Influence Upon POTV LH ₂ Tank Pressure Following Thermal Equilibrium Fill Process	6-43

LIST OF TABLES

۰

Tab	ble	Page
2-1	Categorization Limits	2-4
2-2	Initial Selection of Representative Vehicles	2-6
2-3	Representative Candidate Receivers	2-7
2-4	Additional Characteristics for Selected Representative Vehicle Tanks •	2-7
3-1	A Minimum Set of On-Orbit Refueling Criteria is Required for Acceptable Operation	3- 12
3-2	Typical Example of a Resupply Concept Screening Procedure ••••••	3-13
3-3	Propellant Tank Refill with Liquid Spray was Previously Identified as a Viable Concept	3-14
3- 4	Helium Expended for Pressurization of Dual Stage POTV · · · · · · · · · · · · · · · · · · ·	3-17
3-5	Propellant Transporation Efficiency for On-Orbit Resupply of Dual Stage POTV · · · · · · · · · · · · · · · · · · ·	3-27
3-6	Baseline Conditions Selected for Liquid Hydrogen Tank Prechill Procedure	3-33
3-7	Baseline Conditions Selected for Liquid Hydrogen Propellant Tank Fill Analysis	3-42
3-8	Ullage Cooling Required to Attain Thermal Equilibrium During POTV LH ₂ Task Fill	3-42
3-9	Supply Tank Conditions During LH ₂ Tank POTV Refill	8-50
3-10	POTV Start Basket Characteristics	-57

PRECEDING PAGE BLANK NOT FR.MED

LIST OF TABLES (CONT'D)

•

Table

.

3-:	11 Subsystem Influence Upon Refill Procedures
3-1	2 Propellant Transfer Can Be Accomplished With a minimum
3-1	3 Propellant Transfer Timelines (Flights 1 & 2)
3-14	4 LH ₂ Propellant Transfer Flow Conditions Selected for POTV Refill
3-15	Propellant Transfer Timeline (Flight 3)
4-1	COTV Propellant Tanks Final MECO Residuals
4-2	Propellant Depot Facility Cost Estimate
4-3	On-Orbit Refueling of a COTV (Helium Pressurization Q
4-4	Conditions Selected for COTV Liquid Hydrogen Tank Prechill
5-1	LTL Vehicle Tanks Venting Procedure Prior to Orbitor D
5-2	LTL Vehicle Propellant Tanks Refill Procedure (Applicable to N_2O_4 and MMH).
5-3	LTL Vehicle Helium Re-Supply Options
6-1	OTV LH ₂ Model Test Tank Weight Summary
6-2	Model Test Tank Volume-To-Mass Ratio Comparison With OTV LH ₂ Prototype Tank
6-3	Model Tank Scale Influence Upon Test Variables
6-4	Scale Model Test Variables for Simulating OTV LH2 Tank Prechill 6-28

LIST OF TABLES (CONT'D)

,

<u>Table</u>		Page
6-5	Model Tank Scale Influence Upon Fill Tank Variables $d^* = f (m^* v^{*2} / V_L^*) \cdots$	6-37
6-6	Model Tank Scale Influence Upon Fill Tank Variables (d*=1)	6-37
6-7	Model Tank Scale and Fluid Substitute Influence Upon Fill Test Variables $d^* = f (m^* v^* V_L^*)$, for LN ₂	6-47

SUMMARY

This study was concerned with three main areas: 1) identification of representative receiver-fluid combinations for propulsion vehicle tankage, 2) on-orbit refill analysis of each of three candidate receivers selected for further evaluation. and 3) modelling analysis to determine experimental conditions necessary for verifying the filling characteristics of each receiver vehicle configuration.

Initially, an evaluation was made of candidate propulsion vehicle system tankage for on-orbit resupply. Various NASA, Convair, and industry studies were reviewed to identify alternative concepts for orbital transfer vehicles (OTV), space platforms/ space stations, and spacecrafts. STAR and International Aerospace Abstracts were consulted to insure adequate coverage of representative concepts. All document sources consulted in this review are referenced in Appendix A. After reviewing applicable documentation, vehicles were listed and categorized (Appendix A).

Following the literature review and vehicle documentation process, one vehicle was selected, in accordance with the statement of work, from each of the following categories: an earth-storable vehicle with partial screen acquisition device, a cryogenic vehicle with partial screen-acquisition device, and a cryogenic vehicle without a screen device. The selected vehicle configurations, given in Figure 2-1, were:

- Earth Storable Vehicle A low thrust (LTL) concept that employs MMH and N₂O₄ propellants. This vehicle is representative of a configuration that can be constructed from existing hardware.
- Cryogenic Vehicle Without Acquisition Device The Personnel Orbital Transfer Vehicle (POTV) consists of two cryogenic stage for delivering payloads (20,000 kg) to geostationary orbit (GEO) and back to low-earth-orbit (LEO). This vehicle uses LH₂ and LO₂ propellants and will be available in the near term (1980's).
- Cryogenic Vehicle With Partial Acquisition Device The Cargo Orbital Transfer Vehicle (COTV) is capable of delivering 250,000 kg to GEO and and returning to LEO. This vehicle is contemplated for the 1990's and beyond in the era of space-basing. Again, main tank propellants are LH₂ and LO₂.

Vehicle missions were defined for each candidate receiver to encompass the key issues of orbital-refueling operations. These operations included major activities from post-mission "storage" in the LEO parking orbit through resupply. A re-fueling analysis of each mission was then performed to determine such variables as propellant losses, propellant tank pressures, fill levels and total refill time.

Analysis of POTV and COTV orbital refill indicated that key issues (or concerns) were the same for each vehicle. These were: a) how to avoid excessive tank pressures, b) how to avoid liquid venting, and c) how to perform refill with the limited resources available in space. It was also found that the same refill technique and procedures were applicable to each vehicle. This was a surprising result in light of the limited orbiter-tanker resources available for POTV refueling, which contrasts with the substantially greater orbital depot resources assumed for COTV refill operations.

The refueling analysis showed that problems could be circumvented by introducing the processes of initial tank vent, prechill and fill.

Initial tank vent is required only if helium is present. Tank vent (or blowdown) to a low pressure will expel sufficient helium that concern for excessive tank pressure (due to helium partial pressure) or helium trapped within a screen device is eliminated.

The prechill process is required to reduce tank temperature to an acceptably low level prior to initiating tank fill. Prechill will consist of a series of charge and vent cycles, where either liquid or vapor is introduced during the charge cycle. Vapor only will be expelled during each vent cycle because the elevated tank temperatures will preclude the possibility that liquid is present at vent initiation. Significant analysis results are listed below:

- 1. Liquid oxygen tank prechill is not required because under no circumstance will excessive tank pressures occur during refueling operations. Thus emphasis was directed at the liquid hydrogen tank.
- 2. Rapid prechill of the hydrogen tank does not appear to be an important consideration. Figure 3-4 indicates that up to 64 hours of activity is required to support a single orbiter/POTV rendezvous and transfer operation, five percent of which may be required for propellant transfer. It seems evident that propellant transfer operations could be increased to 10 percent of the total timeline without significant impact. This is nearly two orders of magnitude more time than the approximate 200 second prechill time indicated by Figure 3-20.

3. Liquid hydrogen consumed for the tank prechill process will have an insignificant influence upon overall efficiency and cost of transporting propellants into space for POTV refueling. As a result, propellant transfer efficiency should not be an important consideration in the prechill process selection.

It is concluded that the prechill process described and analyzed in Section 3.3.2.5 will satisfy the requirements of simplicity, reliability and safety.

At the completion of prechill, the tank is locked up and liquid introduced through one or more spray nozzles to accomplish tank fill. A fill condition of 90 percent or greater will be achieved without the need for venting if near-thermal equilibrium conditions are present. It was determined that sufficient bulk fluid agitation will be created by the entering liquid to provide near-thermal equilibrium during fill. Together, tank prechill and bulk fluid agitation should provide a no-vent fill or refill.

Propellant transfer timelines were developed for a POTV refueled by an orbiter-tanker. Tables 3-13 and 3-15 show that this transfer operation can be performed in three hours by over-lapping LH_2 and LO_2 transfer.

The primary requirements for LTL refueling operations are:

- 1. Minimize propellant tank venting in the vicinity of the orbiter because N_2O_4 and MMH are corrosive. Liquid venting must be avoided.
- 2. Prevent helium entry to the screen galleries because vapor-free liquid flow from each propellant tank must be assured.

Refueling will include the initial vent and tank fill processes, but not prechill, because tank and propellant temperatures will be approximately the same. Propellant tank fill pressures will remain below the vent pressure levels if the initial vent (or blowdown) process reduces tank pressure to approximately one to two atmospheres.

A procedure was identified that would satisfy the above requirements during initial vent. Basically the approach is to rely upon procedures and added propellant plumbing to transfer propellant between tanks. In this way a tank may be drained of excess propellant prior to the initial vent process that expels helium.

The single potential concern of the selected refueling procedure is that propellant contained within the screen devices might boil during tank vent. Boiling will be avoided if sufficient liquid residual is maintained in contact with the screen to replenish liquid lost through evaporation. Orbital experiments were not recommended because such tests would be configuration sensitive and have limited applicability. An important result of this study is that zero-g mass gauging devices will be required for on-orbit refueling operations of earth storable and cryogenically fueled vehicles. A survey was conducted of existing radiation and kF type devices to identify the state of the art.

An analysis was also conducted which indicates that propellant mass gauging is feasible through thermodynamic means of measuring tank pressure increases resulting from a fixed helium mass addition.

The processes selected for further evaluation (i.e. modeling) were prechill and fill. The initial tank vent process was judged to be sufficiently well defined to preclude experimentation. Prechill and fill are similar in one important aspect; it is intended that heat and mass transfer be dominated by forced convection in order that these processes remain independent of acceleration environment. Consequently, a modelling analysis was performed to identify conditions under which these processes can be simulated with a 45.7 cm (18 inch) diameter test tank (the largest size that can be contained within a spacelab doublerack). Per NASA/LeRC directive, the Spacelab was groundruled as the orbital experimental test facility.

It was concluded from computer simulations that results could not be directly extrapolated to a full scale OTV, even for tests conducted in a zero-g environment with LH₂. This discrepancy between model and prototype behavior is influenced by the substantial difference between prototype and model tank volume-to-mass ratio, which is an important test parameter. It is expected, however, that the heat transfer phenomena involved in the prechill and fill processes can be evaluated. Empirical coefficients obtained from such tests could be applied to an analytical model such as HYPRES, which would then be employed for full scale vehicle predictions.

Assessments were also made of the influence of a fluid substitute (LN_2) and a one-g test environment on test results. It was concluded that one-g test results would not be applicable to prototype vehicle predictions but that tests with LN_2 would provide useful data.

Finally, discussions with NASA/MSFC safety personnel yielded the following comments regarding the proposed receiver tank experiments to be conducted in the Space'ab environment:

- 1. A waiver would be required by the experiment integration safety review board to allow the anticipated LN_2 quantities for refueling tests.
- 2. Light hydrogen is unacceptable under any condition.

1

INTRODUCTION

The United States is on the threshold of a space industrialized era. Some of the ambitious space programs conceived by the NASA and industry include the construction of large antenna structures, solar powered satellites, and propellant depots. A common element of these programs is the requirement of effectively transferring propellants in space.

This area of orbital propellant transfer, or propellant management, has long been identified as a technology area by the NASA-LeRC and Convair. A previous study, "Orbital Refill of Fluid Management Systems", Reference 1-1, dealt with the problems of refilling small cryogenic and earth-storable systems. Convair has performed independent studies in the area for several years, Reference 1-2 and 1-3. Experience gained in the previous studies has served as a starting point for this study on orbital refuelling of vehicle tankage.

The objectives of this study were to 1) develop techniques for such necessary orbital propellant transfer and, 2) to identify experimental programs to verify these techniques.

1.1 SCOPE

A number of future missions have been defined which require orbital propellant transfer capability. In near-term, space programs such as the manned-geosynchronous-sortie, and very high energy probes to other planets will require the transfer of propellant quantities in the order of 30 to 300 metric tons per year. Earlier studies have shown that performance and life of operational spacecraft can be increased by resupplying attitude control propellants, fuel cell reactants, sensor coolants, or chemical laser fluids. Beyond the year 2000, large space industrialization programs may require propellant quantities that are several orders of magnitude greater than for the nearterm. The most ambitious program now being considered is the Solar Power Satellite Program requiring Heavy Lift Launch Vehicle (HLLV), space construction bases, and both electric and chemical Orbital Transfer Vehicle (OTV).

Although there are many potential orbital refuelling applications, the scope of this study was limited to analysis of and experimental modeling techniques for propellant transfer between supply tanks and receiver OTVs. The three OTV configurations selected for orbital refill analysis were identified using the selection procedure described in Section 2.

1.2 GROUNDRULES

Guidelines were established for selecting three vehicle configurations representative of those contemplated for various types of future space programs. It was required that one vehicle would be selected from each of the following categories: an earth storable vehicle with partial-screen acquisition device; a cryogenic vehicle with a partialscreen acquisition device; and a cryogenic vehicle without a screen-device. Representative vehicle selections were to be made following a literature review of previously conducted NASA and industry studies, as well as current Convair studies on future space programs.

1.2.1 EARTH STORABLE VEHICLE. An additional requirement was imposed upon this selection process; that of identifying hardware either from existing programs, or from previous study efforts. Because the data base for earth storable vehicles and missions was considerably smaller than for cryogenic OTVs, no attempt was made to optimize the vehicle configuration. Rather, the intent was to select a configuration which would be representative of its vehicle class.

1.2.2 CRYOGENIC VEHICLE. Of the two cryogenic vehicle classes selected, one was assumed to be available in the near-term (1980's) and the other was selected for application in the late 1990's and beyond. The vehicle for near-term application was assumed to have subsystems consistent with its early development period. Consequently, this vehicle will not have a screen acquisition device nor any subsystem requiring considerable technology. The more advanced OTV will be comprised of more sophisticated subsystems, such as a partial screen acquisition and an advanced engine system requiring no pre-pressurization.

Different methods of propellant supply will also be available to each OTV. Space programs for the 1980's will rely upon propellant resupply from an orbiter tanker. Programs contemplated for the 1990's and beyond were assumed to have orbital propellant depots available for OTV resupply.

1.2.3 EXPERIMENT MODELING. Experimental modeling techniques of receiver-tankresupply were developed during the study. These techniques were employed to determine the usefulness of simulant fluids and scale model testing. Both ground based facilities, including drop towers, and the Spacelab on-board the shuttle were assumed to be available for conducting the experiments. Scale model size was limited to the largest experiment test package that could be installed within a Spacelab double rack enclosure. This restriction confined analysis to that of a relatively small scale test tank (approximately one-tenth scale or less).



IDENTIFICATION OF CANDIDATE RECEIVERS

The purpose of task I was to select three potential vehicle concepts for subsequent orbital-refucling analysis. These concepts were to be representative of those vehicles expected to be designed for the 1980's and 1990's; that is, representative in terms of equivalent subsystems, orbital staytimes, thermal requirements, legistics and refurbishment requirements.

Following the literature review, one vehicle was selected, in accordance with the statement of work, from each of the following categories:

- 1. An earth-storable vehicle with partial acquisition device
- 2. A cryogenic vehicle with partial acquisition device
- 3. A cryogenic vehicle without an acquisition device.

The three vehicle concepts selected are shown in Figure 2-1.

Convair's approach for reviewing these conceptual designs and for determining representative configurations to be further analyzed in Task II, is shown in Figure 2-2. First, a literature review was accomplished. In order for it to be complete, all potential propulsive vehicle receiver tanks were included to show the wide variety of uses for propellant transfer technology. The types of receiver tanks identified include the following:

Orbital Transfer Vehicles

Mini-maneuvering (e.g., teleoperator) High and low thrust chemical Nuclear and solar electric (OMS, RCS)

Orbital Maintenance and RCS Tankage

Space station Propellant depots Large space structures Automated satellites (include cooling propellant)

After reviewing applicable documentation, these vehicles were listed and categorized according to fluids used, flow rates, tank geometry and pressure, acceleration-



PROPELLANT CAPACITY, KGM (LBM)

 $LO_2 = 44,407 (97,900) EA.$ $LH_2 = 8,256 (18,200) EA.$



 $LO_2 = 198, 817 (438, 038) EA.$ $LH_2 = 37,055 (81,692) EA.$

LTL



 $N_2O_4 = 3,524$ (7,768) MMH = 2,129 (4,693)

Figure 2-1. Three Vehicle Configurations Were Selected With Concurrence From NASA/LeRC

environment, and total quantities of fluids consumed as directed by the statement of work. Baseline vehicle characteristics were derived and candidates which had appropriate requirements for orbital resupply were selected.

2.1 LITERATURE REVIEW

Various NASA, Convair, Aerospace Corp. and industry studies have been reviewed to identify alternative concepts for orbital transfer vehicles (OTV), space platforms/ space stations, and automated and manned spacecraft. The Convair space data banks



Figure 2-2. A Straightforward Review of all Upper Stage Concepts was Employed to Yield Representative Study-Candidates and personnel were also reviewed to obtain available information for identification and description of orbital transfer, space platforms and spacecraft expected to have tanks refilled in space. STAR and International Aerospace Abstracts were consulted to insure adequate coverage of representative concepts. All document sources consulted in this review are referenced in Appendix A.

All available tanks that could conceivably be involved in fluid transfer were also considered. This included the STS Reaction Control System (RCS), the Orbital Maneuvering System (OMS) propellant tanks, and the Shuttle External Tanks (ET) as depot or OTV configuration. As information was extracted for each item, the source document identify number and pertinent page numbers were referenced.

Seventy-nine candidates were identified as a result of this literature review, and are tabulated in Tables A-1 through A-4, Appendix A.

The first 19 items (Table A-1) include the STS orbiter, space tugs, and orbital-transfer type vehicles and stages derived from previous upper-stage programs

such as reusable Agena, Centaur, and the transtage. Early large-scale orbital-transfer vehicles based on the Saturn V S-II stage tanks and engines are also described. The first nineteen items also include early space stations, LEO observatories, and a space taxi. Items 20 through 39 (Table A-2) include the various OTV concepts identified to support and move large space structures as solar power satellites, large radar platforms, propellant depots, space stations, and manufacturing facilities. Some duplication may exist due to inclusion of competitive concepts and designs. Items 40 through 69 (Table A-3) include concepts for propellant depots, the various development phase versions of solar power satellites, the supporting space stations, staging depots, construction space bases, radar platforms, earth observation platforms, antenna farms (for communication and power relay), and logistics tanks. The emphasis was on showing space station/platform concepts likely to use refillable tanks. Items 70 through 79 (Table A-4) include automated spacecraft likely to include refillable tanks; items 80-82 (Table A-4) include manned-spacecraft concepts.

2.2 IDENTIFICATION OF FLUID PARAMETERS AND TANK GEOMETRY

Vehicle programs and concepts identified from the literature search were further defined according to propellant fluid, tank geometry, operational and venting pressure ϵ , celeration environment, fluid temperature and propellant expulsion rate, as required by the

Statement of Work. Data not currently existing was derived from conceptual design data. For instance, tank geometry was assumed to be constrained by STS Orbiter cargo-bay dimensions, where STS was the designated delivery vehicle. Similarly, acceleration forces were matched to the mission. For example, delivery of large space structures from LEO to GEO requires low-thrust acceleration less than 1G; consequently, propulsion tanks of associated vehicles were assumed to operate in a less-than 1G acceleration environment.

2.3 CATEGORIZATION

A generalized classification of space vehicle tanks into fluid/acquisition classes was accomplished using the following groupings as stipulated by the proposal.

Class A	Storable with Acquisition Device
Class B	Cryogenic with Acquisition Device
Class C	Cryogenic without Acquisition Device

Additional categorization was accomplished in terms of tank size, tank geometry, operating pressure, vent pressure, temperature and flow rates as defined by limits given in Table 2-1. These categories, together with background in mission analysis, enabled the selection of representatives for each tank size.

SIZE:	WEIGHT:							
Largo	> 45400 kg (100, 000 I	(sı)						
Medium Small	45400 kg (100, 000 1.ba) - 4 4540 kg (10, 000 1.ba) - 4	45400 kg (100, 000 Lbs) - 4540 kg (10, 000 Lbs), (45000 kg - 2, 000 kg for LH ₂) 4540 kg (10, 000 Lbs) - 454 kg (1000 Lbs)						
SHAPE (Geometry Spheroidal Cylindrical Elipsoidal Toroidal	y);	OPERATING PRESSURE, $kN/m^2(psia)$: Low < 200 (29) Medium 200-2000 (29 - 290) High > 2000 (290)						
ACQUISITION SYS Acceleration Capillary Bladder Other: Pross Pump ACCELERATION I TEMPERATURE:	STEM/METHOD; n ure LEVE1.; <or> 1g</or>	VENT PRESSURE, $kN/m^2(psia)$: Low < 200 (29) Medium 200-2000 (29 - 290) High > 2000 (290) FLOW RATE, $kg/m^2/soc$ Low <10 (2) Medium 10 - 100 (2 - 20) High > 100 (20)						
Cryogente Avg High Hot	<200K (Boths Off) 200 - 320K (Storable) 320 - 400F (Cools Off) > 400K (Rapid Cooling)							

Table 2-1.	Categorization	Limits
------------	----------------	--------

2.4 BASELINE CANDIDATE VEHICLES

Baseline candidate vehicles were selected from the total listing in Tables A (Appendix A). This list of candidate vehicles is comprised of representative and realistic designs that are most likely to require propellant transfer in the next two decades.

Initially, likely candidates were screened based on those thought to be applicable for missions projected in this time period. On-orbit propellant transfer will operate in two general mission arenas. One is the transportation of payloads from LEO to a high energy orbit, e.g., LEO to GEO, and LEO to lunar orbit. The second is moving systems within an orbit. The former requires high impulse eryogenic propellants lifting heavy payloads. The latter operates near a base location for servicing and maneuvering payloads. These vehicles may require long orbit stay times between refill and are better suited for earth storable propellants. Generally these vehicles operate within a few hundred miles and in a range of orbit inclinations from the propellant base.

In this timeframe propellant transfer technology will first be used for topping off eryogenic vehicles which cannot be carried full to orbit due to STS payload limitations. Toward the latter part of the 1990s both cryogenic and earth storable vehicles are expected to be space based.

In the process of vehicle selection, those stages based on existing expendable vehicles were eliminated quickly. These vehicles if used in the Shuttle/Orbiter will be flown in one flight; therefore, not requiring propellant transfer. Also eliminated were receivers for RCS propellant. These generally require small amounts of propellant; a better solution might be total receiver tank changeout versus propellant transfer.

In the past two years, much effort has been concentrated defining the Solar Power Satellite (SPS) and its transportation system. Vehicles have been defined in those studies which depend on propellant transfer either at LEO or GEO. The SPS vehicles appear representative of those required for future space needs. From these-defined vehicles, Table 2-2 details the vehicle selected based on propellant type and acquisition device, and tank shape and size.

To enable a more thorough screening, each of the three major categories were broken down into three sub-categories (large, medium, and small tank). An attempt was made to obtain a representative candidate for each major category and sub-category as shown in Table 2–3. This comparative matrix was used to determine the final three vehicles selected.

2.5 VEHICLE RECOMMENDED FOR ANALYSIS

The determination of which three vehicles should be selected from Table 2-3 was primarily based on usage and configurations most likely to be produced in the 1980s

Representativo Vahialo	1							
reprosentative veniere	Justification							
Low thrust liquid orbital vehicle (Item 37, 39) (Two options shown)	Maximum number of refills expected							
Common stage OTV (117K/117K), (Item 22) (With acq. device) POTV, (Item 34) (Same without acq. device	Required for SPS program, or most manned GEO operations							
Pilot plant SPS (Item 44) Liquid— gaseous argon for orbit main- tenance + RCS (Periodically re- filled) during assembly at LEO	Requires least propellant delivery to LEO during assembly							
Representative Victor								
the predentativ	e veniele							
LH_2 tank for common stage OTV (Item	n 22)							
LO ₅ tank for common stage OTV (Item 22)								
1/2 stage oxidizer & fuel tanks for OT	V-475 ST (Item 36) or satellite control							
On orbit assessing only to the term								
$\frac{1}{2}$ tank (Iten	n 29)							
OTV - 475T (520K/520K) Item 26)	LH ₂ Cylindrical							
Common stage OTV (117K/117K) Item	$\frac{LO_2^{\sim}}{2}$ Ellipsoidal 22 – LH ₂ Cylindrical							
• Small $1/2$ stage oxidizer $(1\Omega_2)$ & fuel $(1.H_2)$ tanks for OTV 475T (Itom 26) or Teleoperator retrieval system hydrazine tanks (N $_2H_4$ cylindrical)								
	Representative VehicleLow thrust liquid orbital vehicle(Item 37, 39) (Two options shown)Common stage OTV (117K/117K),(Item 22) (With acq. device)POTV, (Item 34) (Same withoutacq. devicePilot plant SPS (Item 44) Liquid—gaseous argon for orbit malin-tenance + RCS (Periodically re-filled) during assembly at LEORepresentativeLH2 tank for common stage OTV (ItemLO2 tank for common stage OTV (ItemLO2 tank for common stage OTV (Item00 orbit assembly OTV LO2 tank (ItemOTV - 475T (520K/520K) Item 26)Common stage OTV (117K/117K) ItemL/2 stage oxidizer (LO2) & fuel (LH2)Feleoperator retrieval system hydrazi							

Table 2-2. Initial Selection of Representative Vehicles

and 1990s. With NASA/LeRC concurrence, three vehicles shown in Figure 2-1 were selected. Table 2-4 contains a summary tabulation of the vehicle characteristics.

2.5.1 CLASS A - EARTH STORABLE VEHICLE WITH PARTIAL ACQUISITION DEVICE. A low thrust liquid (LTL) concept was selected as representative of this class. The usage of this vehicle would primarily occur near one altitude location, e.g., LEO, GEO, lunar orbit. However, during early years operation it could be considered for moving large space demonstration structures between LEO and GEO. Its primary function would be to service, inspect, and retrieve objects near its operating altitude base. At LEO its prime function would be to enhance the Shuttle capability by placing the Shuttle payload at altitudes and inclinations beyond the STS capability. At other altitudes payload propellant servicing or module replacement are uses which may enhance cost effectiveness. Orbital debris removal is another application for the LTL vehicle.

2.5.2 CLASS B - CRYOGENIC VEHICLE WITH PARTIAL ACQUISITION DEVICE. The Personnel Grbital Transfer Vehicle (POTV) consists of two cryogenic 117,000 pound propellant stages capable of delivering three or four men to geostationary orbit and back to LEO without GEO refueling. The vehicle with GEO refueling is capable of delivering a 75 man passive module plus two-man crew module plus 20,000 Kg (44,000 lb) Table 2-3. Representative Candidate Receivers

le fressente et	CASE 315 72 1012 19. 14 10 - 1574, 102 20 - 272	Table Act, 2835 A CHORDAN, U. A. CHORDAN, A. CHORDAN, A. CHORDAN, A CHORDAN, A CHORDAN, A CHORDAN, A CHORDAN, A	186 1136-1244 1210-124-1242	Ashiri tete ta . Jatenti, V. H	<u>1467.0</u> (5 m. ² - 17 1500 1156 - 5 - 10 10 M 18-2-51, M 18 - 2 - 5	THE A S. CHISPALL. PROC. AVERIA, 1923	ZSCIISE-, A.C. D. 1956 P.Mef-1040-010	Borrog Islaw Zullan. P. 3629 (es anna) <u>P. 9. p. 5</u> 0
	Multipurpose LECation Provider area managedation	Mittary + 16655 dates 2.4.12 Jayl - 16 Freesen schilt	Holleter OTV men alt	[152] E. E. C. Doge Hert, Lorge et aunder of a fille OSCOP de modernen an et an.	Propert acapter allows Its relating to Octomation	Te III di Eromole 15 Norsy plates for de lavery of Borda to 1754 controre 15 Li Cored GLG	The shall furgest much re-	Partick Construction 1.114 This orbit maintenance 2. This what is the groun of
<u>1.45, 0546 1965</u>	LEO Dame France Sciences - rection translation, cardy receiver to the second	Orbot Maint + attribute L.P.a.tribute - attribute	Le 19 (1995), e 9 noval	Liberto Gilo generalista congesta de la consta de consta de la consta de la const	Version attraction and address of the second	e en gan a parte 1977 E entre de facerack Martin (1923-1112	Alterate for EFG to GFG.	Lar Gapteletric accelletric of algebra
fis ign Arread		Yes	Ye	Y				¥
Area 24.054	turnor y	Capitlary. Bladder	Ca, Illary C q Illary	Cudibury Custibury	Capitlary Capitlary	Acol Acol	Ace I Ace I	
	- 16 	2015 100	Files S.F.	Ellips. Cyl.		Copyer/ FIL.p.c.	111.11 111.11 111.11	Ç,
	- Harden	Teal Leaven		Modern. Modern		Line.	M-dour M-dour	-
	- -				: : : : : : :			
1. 4 Free Hart	MMI 11,624 (25, 656)	8-21-4 - 3, 734 (7, 76-6) MET - 2, 122 (4, 953) 1-214 - 8-6 (1, 506)	1.02 950,040 1.03 600 1.05 - 05,040 (41,600)	1102 - 14, 110 (97, 000) 1 102 - 2.00 (15, 200)		1.4 ₂ = 60 3, 700 (1, 331, 9 0 1.1 ₂ - <u>100, 700 (222, 909</u>)	1472 - 40,500 89,3001 1.052 - 7,130 00,1001	$\mathbf{L}_{\mathbf{v}_{1}} = 0_{\mathbf{v}_{1}} \mathbf{v}_{\mathbf{v}_{2}} (\mathbf{e}, \mathbf{o}_{\mathbf{v}_{2}})$
Report Street	Low thread repud estantial which	1.31 Particials Tark of the 29 To be greated dress 20,	0-1V-4751-0596/52045 Fujbad - 250 T-69 Building 201	Comora Negr OT V 017K-037E-0 75 mora + 44K 01980-222	172 (Suge Lie OTV 475) 1520K (SucK) Risim 20)		Periorial rebit Treactor Senation (Fe.172) attena 34)	Rocconstant di Placerer 2005 Note for SPA e Roce, 140 Note - Communication - 100
[++/126	84 - 14 - 1			- ² Methoda				Ŧ.
Elunit Arthuart at Cire A Strable with	Artigo estas e De serv		te Catylogramic Matthe Catylogramic	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		Cryster 110, Brite Col		· · ·

Table 2-4. Additional Characteristics for Selected Representative Vehicle Tanks

<mark>./.m² (psi)</mark> Maximum	()		365	(53)	200	(29)	325	(12.1)	192	(27. %)
Pressure ky Operating	1724 (250)	1794 (950)	172	(25)	172	(22)	172	(25)	172	(25)
Tank Maternal	Titanium	Titanium	Welded	AL	Welded	AL	2219-787	AL	2219-TF	AL, ALY
Tank Mass KGM (LBM)	253 (559)	233 (558)	1150	(2536)	2224	(1161)	260	(573)	-147	(986)
Tank Volume m ³ (Ft ³)	2.5 (90)	2.5 (90)	153	(05F9)	515	(19, 363)	41.3	(1457)	116.1	(4100)
Tank Surface Area m ² (Ft ²)	3. f (39. 1)	3.5 (39.1)	159	(1710)	339	(3648)	59	(634)	125	(1378)
Fluid Mass Per Tank KGM (LBM)	3524 (7768)	2129 (4693)	19-, 817	(438, 305)	37, 055	(E1,652)	11,407	(97, 900)	±256	(14,290)
I'luid Type	2 0 2 4	HEEK	LC,	1	1.H ₃	1	ĽĆ,	1	LH,	
Selected Vehicle Name	LTL (With OMS Tarks ⁽¹⁾), Item 39		OT V (520K/520K)	Fith Acquisition Device	(Item 26)		Common Stage OTV (117K/	1 17 ⁴), Item 22 (With	Acquisition Device)	
Class Reference No.	A-3. J		B-1				5-2			

(1) $1, 25n; (19, 1-in.) diameter \times 2, 4m; (94, 3-in.) long.$

,

2-7

of payload to GEO and returning both manned modules. This is 48,500 Kg (107,000 lb) to GEO and returning 28,500 Kg (63,000 lb).

This vehicle is a very effective system for the SPS era when large manned requirements are required at GEO for repair and construction of these satellites. Further, this vehicle is not limited to the era of heavy lift launch vehicles (HLLVs), the vehicles can be carried in separate Shuttle flights and topped with propellants by an Orbiter/Tanker or by a propellant depot. Early uses of the POTV would be required for GEO payload servicing and repair. The function of man in space is to augment unmanned servicing tasks. Man would be used to diagnose and repair space structures; do the out-of-theordinary space functions.

2.5.3 CLASS C - CRYOGENIC VEHICLE WITHOUT ACQUISITION DEVICE. The Cargo Orbital Transfer Vehicle (COTV) is capable of delivering 250,000 Kg (550,000 lb) to GEO and returning to LEO. This vehicle would operate in an era of space basing. Present concepts consider use of an electric OTV. However, these vehicles require nearly a year to transfer payloads from LEO to GEO. Should mission duration requirements of electric OTV technology prove infeasible the COTV would be developed and represent the largest vehicle category.

3

POTV ORBITAL RESUPPLY

In this section a mission scenario will be developed for the POTV concept selected in Section 2. A realistic mission will be defined which encompasses the key issues of orbital-refueling operations. These operations will include all major activities from post-mission "storage" in the LEO parking orbit through resupply. Vehicle and

orbiter-tanker subsystem requirements needed for orbital refueling will be identified. Operational procedures and techniques for orbital propellant transfer will then be developed.

3.1 MISSION SCENARIO

In the early 1990's, with propellant depots not yet available, OTV orbital resupply could be provided by dedicated Orbiter tankers. Figure 3-1 illustrates an orbital refueling operation in which propellants are transferred from a tanker kit to an OTV which is docked to the Orbiter. The tanker kit, (consisting of an LH₂ and an LO₂ tank, transfer system, and pressurization system), is contained within the Orbiter payload bay.





3.1.1 SELECTED POTV MISSION. Our current study of orbital propellant handling and storage systems (Contract NAS9-15640, Reference 3-1) has defined the mission operations sequence for a manned five day sortie mission to GEO employing a space-based, two-stage OTV. Figure 3-2 illustrates the major orbital operations required, including rendezvous and docking, propellant transfer, mating of OTV stages and crew module, orbit transfer to GEO, staging, operation at GEO, orbit transfer to LEO, and crew-module retrieval. This mission has been selected as a representative scenario for POTV refueling operations.

3.1.1.1 <u>Timelines.</u> Timelines were developed to determine the impact of various operations. The timeline for the total five-day manned GEO sortie mission is presented in Figure 3-3. This timeline is based on one Orbiter vehicle, two shifts (no weekends) for Orbiter processing, and three shifts for launch processing at the pad. The Orbiter will be committed for 47 days of which the major contributor (73 percent) is ground turnaround time.

The operations timeline for Orbiter flights 1 and 2 is presented in Figure 3-4. The total flight operations time is seen to take less than three days. The first working day of operations is the launch, rendezvous, docking, and IVA inspection and checkout of the POTV. (This also allows time for the crew to adjust to zero-g conditions before EVA is attempted.) The second day is dedicated to performing POTV inspection and maintenance tasks. A space-based POTV would be designed for conditioned monitored maintenance whereby any subsystem degradation or failure would be recognized beforehand and the appropriate module would be aboard the Orbiter for replacement by EVA or RMS. A nominal allowance of 6 hours EVA activity plus $4^{1/2}$ hours pre- and post-EVA operations is considered appropriate for nominal inspection and maintenance tasks.



Figure 3-2. Operations for 5-day manned GEO sortie mission.



*2 SHIFTS; NO WEEKEND OPERATION **3 SHIFTS FOR PAD PROCESSING

Figure 3-3. Timeline for 5-day manned GEO sortie mission (one orbiter tanker)

EVENT	ELAPSED C TIME (HR)	UMULATIVE TIME (HR)	0	DAYS	2	3
ORBITER LAUNCH RENDEZVOUS PHASING* RENDEZVOUS CLOSING AND DOCKING POTV CHECKOUT/INSPECTION - IVA CREW SUSTAINING ACTIVITY POTV MAINTENANCE - EVA** CREW SUSTAINING ACTIVITY PROPELLANT TRANSFER SEPARATION THERMAL CONDITIONING + ENTRY PHASING REENTRY	$1 \\ 0 - 24 \\ 5 \\ 2 \\ 2 (EST) \\ 14 \\ 10 (EST) \\ 14 \\ 3 \\ < 12 - 15 \\ 0.5$	1 1 8 8 10 24 34 48 51 63 63		ar a george a training		

*ASSUMED ORBIT PHASED WITH LAUNCH SITE (31° INCLINATION, 478 km (258 N.MI.) ALTITUDE), OTHERWISE FENDEZVOUS PHASING IS 0-24 HR.

**NOMINAL ALLOWANCE FOR ROUTINE INSPECTION OF OTV WITH MMU AND REPLACEMENT OF MODULES REQUIRING MAINTENANCE (CONDITION MONITORED MAINTENANCE).

Figure 3-4. Tanker flight 1 and 2 operations timeline
It was estimated in Reference 3-1 that only three hours will be required for propellant transfer operations. Although this duration may not be correct, it is significant that propellant transfer may represent only 5 percent of the total flight operations timeline. It appears from Figure 3-4 that doubling this time will have virtually no impact upon the total flight operations. Thus, the capability for rapid propellant transfer became a minor element in this study.

3.1.2 ORBITER TANKER CONFIGURATION. The orbiter tanker selected for this scenario is the configuration defined in Reference 3-1.

The selected dewar pictured in Figure 3-5 features two separate propellant tanks equipped with hemispherical bulkheads. The LO_2 tank has a reversed bulkhead, so that the two tanks can be nested to reduce the overall length. The LH_2 tank is located forward and the LO_2 tank is positioned aft in the Orbiter payload bay. A single vacuum shell equipped with three girth rings and five intermediate stiffener rings encases the two tanks. The forward and aft girth rings serve as structural ties to the Shuttle, and all three girth rings provide support for the tanks. The primary structure for the vacuum shell will be aluminum alloy isogrid, semimonocoque, or a combination of both. Both tanks are suspended from the vacuum-shell girth rings, using low-conductive struts arranged in "V" patterns and oriented such that the load paths are directed tangentially into the aft bulkheads. For the LH_2 tank, these support struts are augmented with low conductive drag links located at the forward bulkhead. Multilayer insulation (MLI) blankets are applied over all surfaces of each tank.





Propellant acquisition is accomplished using a full-screen channel system, which maintains communication with propellants located anywhere in the tank without any special settling thrust. This concept was selected because it imposes no restraints upon Orbiter operations, since propellant transfer can be successfully performed for any orientation, even during maneuvers. There is a family of Orbiter interface lines including propellant ground fill and vent, nearly identical to a ground-based OTV. On top of the propellant supply tank are folded lines which can be extended to connect

3.1.3 POTV CONFIGURATION. The basic POTV configuration for performing the five day sortie-mission to GEO is given in Figure 3-6. Vehicle subsystems will be selected on the basis of mission and space-based requirements. A list of subsystems influenced by mission requirements will include tank size, pressurization system, propellant acquisition system, insulation system and vent system. Subsystems influenced by space-basing requirements will include insulation system and vent will be discussed first. This selection process will be followed by a description of a "typical" OTV mission for which potential problems can tidentified and subsystems



3.1.3.1 Subsystems Influenced by Mission Requirements.

<u>Pressurization System</u> - The selected pressurization system will require helium for propellant tank pre-pressurization for each main engine start. Pressurization requirements during main engine firing will be autogenous for the liquid hydrogen tank and helium for the liquid oxygen tank. Main engine-start helium usages will not be excessive because engine-start NPSP requirement will be approximately 3.45 kN/m^2 (0.5 psid) (LH₂ tank) and 6.9 kN/m² (1.0 psid) (LO₂ tank). Total mission helium usages will be relatively small for the LO₂ tank because helium will be bubbled through the liquid bulk. The tank pressure increase will be primarily due to propellant evaporation into the helium bubbles. Autogeneous pressurization was selected for the hydrogen tank because a) it is a simple and proven approach, and b) the alternative helium pressurization approach will be considerably heavier. This type of pressurization system was analyzed in contract NAS3-20092, Reference 3-2.

Helium within a propellant tank can complicate an orbital tanking procedure because of the need to expel most of the inert gas before propellant transfer can be initiated. Unfortunately, in the near-term, there is no viable alternative to helium pressurization for main engine start since main engine NPSP requirements must be satisfied. An advanced engine with "boot-strap" capability, i.e., with no NSPS requirements, may be developed in the future. A major benefit from this development will be a simplified refueling procedure. Until then, refueling operations must be capable of dealing with helium inside the propellant tanks.

<u>Propellant Acquisition System</u> - Analyses were performed in Contract NAS3-20092 to assess the benefits of a partial propellant acquisition system for OTV. The acquisition system combined with a thermal subcooler was analyzed to determine if these subsystems could replace helium pressurization and RCS subsystems. Although a final assessment has not been reported, it is likely that an acquisition system is not performance effective for a number of OTV missions. At this time, it is judged that a partial screen acquisition device will not be included as part of an OTV configuration. However, an exception to subsystem selection is made in this case and with the pressurization ystem, as explained below.

To provide a more thorough discussion and analysis of POTV orbital refueling operations, it was decided to include the helium pressurization and screen acquisition (start basket) subsystems. In this way the influence of each upon refueling techniques or procedures could be assessed.

<u>Insulation System</u> - A multilayer insulation (MLI) system was selected as being representative of thermal protection systems which may be employed for OTV. A single blanket consisting of twenty MLI layers was selected on the basis of a previous analysis conducted for Contract NAS3-20092. Radiative properties of the organically-coated aluminized Kapton Superfloc MLI will result in a maximum equilibrium temperature of 289K (~520R) (Figure 3-7) for the estimated α/ϵ of 0.3. This maximum temperature is based upon the worst case assumption of a tank surface continuously exposed to the sun. The lower temperature curve is for the assumed condition of a vehicle rotating at a rate sufficient to maintain uniform skin temperatures throughout.

The time required for the propellant tanks to attain equilibrium temperatures will be dependent upon the propellant mass remaining at POTV final MECO. These residuals must first boiloff and be vented overboard before the propellant tank temperatures can increase. Figure 3-8 gives the time required for residual liquids te boiloff as a function of initial liquid residual. Studies have not yet been conducted from which an acceptable liquid residual range can be determined. On the basis of Centaur vehicle flight experience, however, a residual of about 200 kg (441 lb) per tank appears reasonable. These quantities will boiloff in about 6 to 8 days. Beyond this time, the propellant tanks and remaining vapor will begin to increase in temperature as shown in Figures 3-9 and 3-10. The rate of temperature increase will depend upon the external shield temperature, which can be as high as 289K (520R), as indicated by Figure 3-7. Propellant tank transient time to steady state is given for three external temperatures to show how this transient will be influenced by vehicle roll-rate and α/ϵ . It is seen that liquid-oxygen tank equilibrium can be attained in a minimum of 8 to 12 days, and hydrogen tank equilibrium can occur in a minimum of 16 to 22 days.

<u>Vent System</u> - A thermodynamic vent-system will be required to provide vent capability for the proposed OTV mission profiles. This type of vent system can maintain vehicle tank pressure control in a zero-g environment regardless of fluid quality at the heat exchanger inlet. Vent system sizing will be keyed to the MLI thermal protection capability. Additional vent capability may be required to satisfy refueling procedures. Further discussion will be postponed until after space-basing requirements have been evaluated.

3.1.3.2 Subsystems Influenced by Space-Basing Requirements – Space-basing conditions are defined as those conditions affecting the OTV from post-mission storage of each stage until after resupply. The period where OTV and orbiter are docked is exempted, since it is part of the tanking duration. Any subsystem capability needed to maintain the OTV in a "safed" condition for subsequent refueling operations is considered to be a space-basing vehicle requirement. The insulation and vent systems selection will be influenced by space-basing considerations.

<u>Insulation System</u> -In addition to the mission requirements previously identified, the insulation system must provide thermal protection for propellants where multiple orbiter flights are needed to support a single OTV mission. For this scenario, it is likely that the OTV stages will reside in orbit for several weeks before tanking is complete. Too little insulation will result in excessive propellant boiloff prior to a mission. Ideally, a trade analysis should be conducted to balance mission versus space-based requirements. This study assumed twenty ML₄ layers was acceptable for both requirements.

- 1. α/ϵ = shield absorbitivity to emissivity ratio
 - = 0.3 (expected value)
- 2. Curve 1 assumes vehicle roll rate that maintains uniform shield temperature
- 3. Curve 2 assumes that shield is continuously exposed to solar radiation





NOTES:

- 1. Liquid and vapor initially saturated @ 103 kN/m² (15 psia)
- 2. 20 layers MLI
- 3. Propellant tank heating rates are: LH₂ tank = 0.18 kW (607 Btu/hr) LO₂ tank = 0.07 kW (232 Btu/hr)
- 4. Time includes ≈ 0.4 days for tank pressure to increase to 138 kN/m² (20) vent pressure



- 1. 20 layer MLI
- 2. Propellant tank initially filled with vapor at: $P = 138 \text{ kN/m}^2$ (20 psia) T = 111K (200 R)
- 3. Constant tank pressure maintained in crbit
- 4. Tank skin and ullage reside at same temperature
- 5. Expected external shield temperature = 278K (520R)



TIME, DAYS

Figure 3-9. Transient time for POTV LO_2 tank to attain temperature equilibrium in LEO

- 1. 20 layers MLI
- 2. Propellant tank initially filled with vapor at: $P = 138 \text{ kN/m}^2$ (20 psia) T = 26.7K (48 R)

3. Constant tank pressure maintained

4. Tank skin and ullage reside at same temperature

5. Expected external shield temperature = 289K (520R)



Figure 3-10. Transient time for POTV LH_2 tank to attain temperature equilibrium in LEO

Vent System - There may be an advantage in performing propellant tank blowdown to a low pressure prior to initiating orbital refill. This operation could be performed with the thermodynamic vent system except that its limited vent capability would result in an extremely long blowdown mode. The propellant mass vented during this orbital period is likely to be at least an order of magnitude greater than the mass vented during the mission. Furthermore, the long blowdown duration could delay rendezvous by several days. It appears, therefore, that a requirement exists for another system capable of vent rates at least an order of magnitude greater than those of the thermodynamic vent system. A non-propulsive vent system will be included in the vehicle

ORBITAL PROPELLANT RESUPPLY TECHNIQUES 3.2

Refueling operations of space-based OTVs, whether from an Orbiter tanker or space depot, will present obstacles not heretofore experienced by operational vehicles. The obstacles are: (1) the hostile space environment and (2) the limited resources available for space-based operations.

The space environment introduces three variables (zero-g, vacuum, and changing orbital conditions), all of which will complicate space-based operations. Liquid vapor distribution is not sufficiently well defined in space to enable pure vapor venting during the filling operation. Also, the cost of transporting propellants into space for refueling will be sufficiently high (>400 %/kg) that two-phase venting is undesirable. The combination of zero-g and vacuum environments will greatly limit freedom of movement during operations. In addition, the variations of heating and lighting environments in low earth orbit will further complicate refueling operations.

The limited personnel, ancillary equipment, and instrumentation resources available in space serve as a major constraint for OTV refueling operations. The flight crew will perform operations in space that require numerous personnel on the ground and the operations will be more complicated. Personnel could be the most severe constraint because the equipment and instrumentation made available for space-based operations must be limited to that which can be monitored and performed by the flight crew. These constraints of limited resources plus the space environment lead to the minimum set of on-orbit refueling criteria given in Table 3-1.

3.2.1 ON-ORBIT RESUPPLY CONCEPTS. Filling a vehicle on the ground can be easily accomplished because vapor is readily expelled as liquid is introduced. Even cooling a vehicle on the ground prior to cryogenic fill is a routine operation. However, low-g propellant fill cannot be satisfactorily accomplished with standard one-g techniques. This is because liquid-vapor distribution will not be well defined during the filling operation. Consequently, a routine ground operation such as venting can become a process requiring considerable care in orbit.

 Table 3-1.
 A minimum set of on-orbit refueling criteria is required for acceptable operation

Flexibility - This is an important ingredient required of refueling procedures because initial conditions may vary from one refueling operation to another. For example, the initial OTV temperature will be a function of time in orbit between refueling operations and liquid residuals at the end of the previous mission. It is conceivable that propellants may reside within the tanks as one extreme condition; the other extreme would be that of vapor-only inside warm propellant tanks. Tank temperature could have a profound influence on the transfer process.

- Simplicity Limited resources demand that a simple and straightforward procedure be devised. The few personnel available must be capable of connecting and disconnecting transfer lines and monitoring systems to avoid supply tank propellant depletion, receiver tank overfill, or overpressure. We must be able to rely on a limited number of measurements to describe propellant transfer conditions adequately.
- Safety Operations must be selected to eliminate any concern for tank over-pressure, and mixing of hydrogen and oxygen in a confined area must be avoided.
- Precision Some degree of precision will be required in this operation to support vehicle missions. Propellant tank pressure, temperatures and tank masses must be known with reasonable accuracy. However, propellant temperatures and pressures can be heavily influenced by the initial OTV thermal condition.
- Efficiency An efficient propellant-management system is necessary because of the high transporation costs of propellant delivery to orbit.

Propellant fill is further complicated by one of two conditions that may exist: (1) in empty storage tank residing at a substantially higher initial temperature than the cryogen prior to initial fill or (2) a partially full tank requiring that helium pressurant be vented before refill can be initiated. Two questions arise. First, how do we fill the storage tank without expending excessive propellants in the process and without exceeding structural allowable storage tank pressure? Second, how can we vent helium without losing liquid overboard in the process? A resupply concept must satisfactorily handle these conditions in addition to satisfying the requirements and constraints previously identified.

A comprehensive screening of potential resupply concepts was conducted on a previous study, Filling of Orbital Fluid Management Systems, NASA CR159404, Reference 3-3, to identify methods of refilling small-scale propellant-management systems in space.

Systems considered include those using pressurant to condense vapor, valving arrangements, pumping, capillary pumping, use of thermodynamic venting, vacuum refilling, inflow baffling, high pressure manifolding, shaping of channels, and propellant depot refilling stations. These concepts were developed into the most reasonable, or most likely, candidates for orbital refilling. The descriptions and comparisons made in that study are applicable to a wide range of receiver applications, including OTVs. Table 3-2 is an example of the data developed showing candidates, their operation, advantages and disadvantages, and other comments.

CANDIDATE CONCEPTS	METHOD OF OPERATION	ADVANTAGES	DISADVANTAGES
1. Standard one-g filling recomplet derive i whether derive i money construction construction The incomendation	Fill using normal ground filling Inlet.	No aditional hardware.	Liquid caper interface unknown during filling – Vapor trapped in device.
2 Hellum pressurant/liquid apray folet.	Fill tank with light using spray nozale to achieve thermodynamic equilibrium daving childron and full. Use before to contense any vapor trapped is the capillary device during filling. Holtaus could be acted just before transfer to acted any possibility of forming vapor in the capil- lary device after pressurfratton.	Patriy eimple hardware. Does not dopond upon position of the low-g interface and therefore can be tasked in earmal gravity with assurance that it will operate in low gravity. Thermal equilibrium results in low pressure rise during fill. No capillary device hardware modifica- tion required. Could be retrofitted to existing systems.	High pressure rise will occur during chilldown which may require venting to occur. Use of holium makes filling a partially full tank more complicated because the holium will probably taxe to be vented off to keep final pressures within limits. Nordown and purge required to replace sitte with Glig before fill of a tank containing only vapor.
3. Hellum pressurant /low pressure rise iniet. 	Fill tank with diffuser to anistave low pri suire rise during chilidiwn. Use hellum to condense any vapur trapped in the eccencilovice.	Fairly atopic hardware, Jow pressure the and therefore best charac of not conting during children). No conflictly desire hardware mode required, Could be "netrofitted" to existing systems.	Noncynffibrium conditions dai ing fill, thurotore high pressure ruse. Door britty of vagoer formation after helions addition if unixing and therefore pressure decay occurs. Some disadeantages as Concept 2 with respect to Offerennoval.

Table 3-2. Typical example of a resupply concept screening procedure

Each of the candidate concepts from the study was compared on the basis of the ten criteria shown in the comparison chart (Table 3-3). The evaluation was performed using hydrogen as the baseline fluid; however, the results are generally applicable to other cryogenic fluids with both condensible and noncondensible pressurant. A more detailed discussion of the concept evaluation can be found in Reference 3-3.

The selected concept (#2) for refilling small scale systems includes filling the tank through a spray nozzle to maintain pressure control, and employing helium pressurant to re-condense vapor trapped within a screen acquisition device. This was one of two concepts evaluated for POTV. The second concept assumed that propellant was introduced through a large diffuser in an effort to achieve extremely low entering velocities. An assessment of the first concept is given in Section 3.3.2. The second concept, which was 'ound to be inadequate, is discussed in Section 3.4.

					•							
Keyr hitghest - 10 Lowest - 0 Lowest - 0 Lowest - 0	Liquid Spray Inlet/ilcilum Pressurant	Low Pressure Rise Inlet/licitum Pressurant	Low Pressure Rise and High Chilldown Efficiency Inlets/ Heltum Pressurant	Autogenous Pressurization to Condense Vapor	Piping Instde Channels to Force Vapor Out of the Channels	Propellan Depot Approach	Thermodynamic Vent or Vapor Cooled Shield Intet to Remove Fluid From Channels	Open Loop Refrigeration to Condense Vapor	Internal Vanes to Direct Flow	Shaping Capillary Devices to Position and Eject Vapor	Standard One G Filling	Propellant Control Surfaces Vent Tube
Criteria	2	3	4	9	6	8	7	12	10	11	1	5
Technical Probability of Successful Operation	10	10	10	7	7	10	7	8	7	7	0	0
Adaptability to Existing Configura- tions	10	10	10	10	2	10	7	4	4	2		
Operatiosal Simplicity	8	8	6	6	10	1	8	6	9	6		
Versatility	10	10	10	10	10	10	7	7	5	4		
Gravity Insensitivity	9	7	7	6	10	7	з	6	4	1		
Ground Testability	10	7	7	6	10	7	3	6	4	1		
State of Development	9	8	7	5	6	3	5	6	5	6		
Cost Effectiveness/ Recurring	10	9	9	8	5	9	8	6	7	6		
Weight/Hardware	9	8	7	9	2	1	7	4	5	5		
Weight/Vented Fluid	10	9	10	10	4	10	7	з	3	4		
Reliability/Safety	10	10	9	9	10	7	9	8	10	10		
Total	105	96	92	86	76	75	71	64	63	52	N/A	N/A

Table 3-3. Propellant tank refill with liquid spray was previously identified as a viable concept

3.3 SELECTED ORBITAL RESUPPLY METHOD

For initial filling, the POTV is expected to reside at a temperature of about 289K (520R). This will be the approximate equilibrium temperature dictated by the multilayer insulation (MLI) radiation properties (Figure 3-7). Vehicle equilibrium conditions are also expected to occur during the approximate six- to eight-week period between the end of one mission and the beginning of refueling operations for another mission. Thus, propellant boiloff could complicate propellant transfer operations. The key issues to be resolved are: 1) how to fill the tanks without expending excessive propellants in the process and without exceeding structural allowable receiver tank pressure, and 2) how to assure that the start basket (if one is present) will be free of vapor at end of propellant tank refill. The approach selected for the POTV appears to resolve these issues. The following steps serve as the primary elements of an acceptable propellant transfer procedure: initial vent, prechill, fill. Each element will be analyzed in detail.

A recommended refueling procedure will be developed for a POTV having each of the following pressurization system-start basket combinations:

- no helium pressurization no start basket a)
- no helium pressurization start basket b)
- helium pressurization no start basket C)
- helium pressurization start basket d)

3.3.1 INITIAL VENT. Propellant tank venting is required whenever it will simplify the prechill and tank-fill operations. There are two occasions when venting is either desirable or mandatory; when helium is to be expelled prior to the refueling operation, and to reduce peak pressures that occur during prechill.

3.3.1.1 Propellant Tank Helium Dilution. Refueling of the empty or near-empty vehicle propellant tanks will be performed prior to a scheduled mission. It is expected that the propellant tanks will be refueled to the 95-97 percent level. Because helium is non-condensible, that quantity in the tank at the start of fill will also be present at the end of tanking. Consequently, the large ullage volume reduction experienced during refill can substantially increase helium partial pressure.

This fact is illustrated by Figure 3-11 which shows that as little as 0.5 kg (1.1 lb) helium in the LO₂ tank can provide a helium partial pressure in excess of 69 kN/m² (10 psia). A partial pressure of this magnitude is considered to be unacceptably high. It is estimated that a helium pressure no greater than 20.7 kN/m² (3 psia) would be acceptable. This assessment is based upon the need to know liquid-vapor-pressure conditions at the end of propellant tanking. Since only tank pressure can be measured, an uncertainty in the residual helium quantity directly affects our knowledge of propellant vapor



Figure 3-11. Liquid oxygen tank helium partial pressure following refueling operation.

Another important reason for expelling helium from the propellant tanks is the need to maintain relatively low tank pressures during the refueling operation. As a guide, propellant vapor pressures could be maintained between 103 kN/m² (15 psia) and 138 kN/m² (20 psia), with a maximum allowable tank pressure of about 172 kN/m² (25 psia). Should pressure variations of about 13.8 kN/m² (2 psi) be selected as a contingency during fill, then helium partial pressure should not exceed 20.7 kN/m² (3 psia). This partial pressure will convert to a maximum allowable helium mass of 0.13 kg (0.28 lb) at initiation of oxygen tank fill (Figure 3-11). Considerably more helium will be acceptable in the liquid hydrogen tank; in excess of 1.36 kg (3 lb) according to Figure 3-12.

Helium residuals at MECO were estimated from work performed on Contract NAS3-20092, which are given in Table 3-4. It is clear that the hydrogen tank does not have to be vented to satisfy the previously expressed propellant tank pressure criteria, whereas considerable oxygen tank venting is required. Two tank blowdowns are needed (Figure 3-13) to reduce the residual helium quantity to an acceptable level. Several hours may be required between vent periods to allow an oxygen tank pressure increase to the level indicated in Figure 3-13.



Figure 3-12. Liquid hydrogen tank helium partial pressure following refueling operation

 Table 3-4.
 Helium expended for pressurization of dual stage

 POTV

	Total H LO ₂ Ta	lelium To ink, kg (lb)	Total H LH ₂ Ta	elium To nk, kg (lb)
1st Stage	1.75	(3.86)	0.89	(1.97)
2nd Stage	1.66	(3.65)	1.44	(3.18)

Note: Helium quantities extracted from Tables 5 and 6, Reference 5.

3.3.1.2 <u>Peak Pressure Reduction</u>. Receiver tank prechill which follows the initial vent period must be conducted with care in order to avoid an excessive tank pressure buildup. Tank pressure during prechill will be influenced by several factors, including initial pressure at prechill initiation. Thus peak pressures can be controlled, in part, by first venting the tank before initiating prechill. The advantages of this vent process will be quantified in the next section.

3.3.2 <u>RECEIVER TANK PRECHILL</u>. Prechill is required whenever bitial tank temperature is such that the stored energy will result in excessive pressure during the tank fill mode. Prechill is accomplished by introducing liquid into the propellant tank at a velocity that provides good heat exchange between the high temperature walls and the cooling fluid. This procedure has the advantage of requiring little mass to effect tank cooling.

1. Initial Blowdown Conditions

 $P = 103 \text{ kN/m}^2 (15 \text{ psia})$ T = 111K (200 R) Helium Mass = 2.75 kg (3.86 lb) Vent Area = 12.9 cm² (2 in²)

2. Helium Mass, kg (lb):

at end 1st blowdown = .244 (.56) at end 2nd blowdown = .089 (.204)

3. It was assumed that ullage temperature increased to 222K (400R) between blowdowns. This could require several hours of heat exchange between the wall and ullage.

.



Figure 3-13. Oxygen tank blowdown for helium expulsion

The primary requirement for system prechill is to reduce tank temperatures sufficiently that the fill process will be accomplished with a locked up tank. Venting is unacceptable during the fill mode because of the possiblity that an unknown quantity of liquid will be lost overboard, since propellant control cannot be maintained during this process. Venting is acceptable during prechill, however, because the elevated tank temperatures will quickly evaporate all liquid propellant during this phase.

The key factor in determining prechill requirements is the theoretical maximum pressure that can occur during tank fill as a function of initial tank temperature. Maximum tank pressure, for a given mass addition, occurs when the tank vapor and tank wall reside at the same temperature. The analytical development for theoretical maximum pressure is given below.

The First Law expression for introducing liquid into a container is

$$dE_g + dE_w = h_L dm_L$$
(3-1)

 $dE_g = (u_g m_g)_2 - (u_g m_g)_1 = change in ullage energy$

If one assumes that the tank is initially evacuated

$$m = 0 \text{ and } dE = u m$$

$$g_1 \qquad g_2 \qquad g_2 \qquad (3-2)$$

Also for an initially evacuated container, $m_{g_2} = dm_L$ (3-3)

$$dE_w = (u_w m_w)_2 - (u_w m_w)_1 = change in tank wall energy$$

Since tank mass is constant, $dE_w = (u_w - u_w) m_w$ (3-4)

Combining Equations 3-1 through 3-4

$$u_{g_2}m_{g_2} + (u_{w_2} u_{w_1})m_{w} = h_{L}m_{g_2}$$
 (3-5)

$$(u_{g_2} - h_L) m_{g_2} = (u_{w_1} - u_{w_2}) m_w$$
 (3-6)

Finally

$$m_{g_2} = (u_{w_1} - u_{w_2}) m_w / (u_g - h_L)$$
(3-7)

where

$$u_{W_2}$$
 and u_{g_2} are evaluated at T_2
 $dEg = change in tank wall internal energy
 $h_L = enthalpy of liquid entering tank$
 $dm_L = differential liquid mass addition to tank$
 $u_g = internal energy of vapor in tank$
 $m_g = mass of vapor in tank$
 $u_w = internal energy of tank wall$
 $m_w = tank wall mass$
 $T = temperature$
subscript$

1 = conditions at beginning of interval

2 = conditions at end of interval

From the equation of state, gas pressure is

$$P_{g_2} = \left(\frac{m \ ZRT}{V_t}\right)_{g_2}$$
(3-8)

,

where

- Z = compressibility factor
- R = gas constant
- $V_t = tank volume$
- P_{g_2} = gas pressure

The theoretical maximum tank pressures during prechill, as described by equations 3-7 and 3-8 are plotted in Figure 3-14 as a function of initial tank temperature for the POTV. Of particular significance is the conclusion that LO_2 tank overpressure will not occur at any time during tank chill. LO_2 MLI radiative properties should maintain tank equilibrium temperatures below 289 K (520°R). This condition will result in a maximum tank pressure of 138 kN/m² (20 psia) which is well below the maximum allowable of about 345 kN/m² (E psia). For the hydrogen tank, however, the maximum allowable pressure of about 193 kN/m² (28 psia) dictates that the propellant tank be prechilled to a temperature less than 236 K (425°R). It is believed that a maximum pressure less than 138 kN/m² (20 psia) is acceptable for this phase of the operation. Therefore, the LH₂ tank will be prechilled to a temperature of about 200 K (360°R).

The following charge and vent procedure was selected for LH_2 tank prechill:

- a. Meter LH_2 into the tank at a high velocity to provide good heat exchange with the walls.
- b. Allow time for a tank pressure increase to 69 kN/m² (10 psia) (vapor temperature equals tank temperature at this time)
- c. Vent the tank to near zero pressure and repeat steps a and b as required to reduce tank temperature below 200K (360 °R)

There are several questions that can be asked about the selected prechill procedure. These are:

- 1. How can we be certain that tank over-pressure will not occur during prechill?
- 2. How can we analytically model a complicated process that includes liquid boiling at hot tank walls as a result of jet or spray impingement?
- 3. How can we be certain that liquid will not be present when tank venting is initiated?
- 4. How will we know when the propellant tank has been prechilled below 200K (360R)

Acceptable procedures or processes are described in the following discussion which satisfactorily answers these questions.

3.3.2.1 <u>Tank Over-Pressure</u>. First, tank over-pressure will be prevented by controlling propellant flow into the tank. Accuracy does not appear to be a critical item as Figure 3-15 indicates. For example, the initial LH₂ charge will require about 9.1 kg (20 lbm); this will create a peak pressure of about 69 kN/m² (10 psia). If 18.2 kg (40 lbm) of LH₂ is inadvertently introduced, peak tank pressure will be about 124 kN/m² (18 psia), which is well below the tank allowable of about 172 kN/m² (25 psia). The data of Figure 3-15 was obtained by solving equations 3-7 and 3-8.



Figure 3-14. POTV LH₂ tank pressures could exceed tank allowables during prechill



Figure 3-15. Precision metering of LH₂ is not needed to avoid over-pressure during prechill

3.3.2.2 Modelling. An important consideration in selecting a refueling technique is whether it can be subjected to model-scaling. A concern is the early prechill period that will be characterized by a complex thermodynamic and fluid mechanic process due to liquid impingement on the hot tank walls. The resulting forced-convectionnucleate and film-boiling-phenomena are extremely difficult to analytically model or scale. Fortunately, the complicated wall-boiling process can be resolved by sidestepping the issue. The wall-boiling-phenomenon need not be a critical part of prechill because peak pressures will not occur during this period. This is illustrated by Figure 3-16, which indicates that peak pressures should occur long after the LH₂ has evaporated and the vapor temperature increased to wall temperature. Peak pressures will occur only at maximum gas temperatures. This knowledge, plus the fact that tank temperature will be at about 200K (360 R) at prechill temperature, should greatly diminish the possibility that liquid will be present at vent initiation. Another factor to consider is that the heat exchange process during the limited boiling period represents only about ten percent of the total energy removed during prechill. This is additional support for the belief that the initial transient boiling period is not as important to the understanding of prechill as are the latter stages of this process.

3.3.2.3 <u>Liquid Venting</u>. Liquid venting will not occur during prechill-vent period because only vapor will be in the tank at vent initiation. This point is illustrated with a review of the selected prechill procedure (and referring to Figure 3-16):

1. About 9.1 kg (20 lb) LH_2 will be metered into the tank at a high velocity.



2. The peak pressure resulting from this mass addition will be about 69 kN/m² (10 psia).

Figure 3-16. A prechill procedure can be identified to eliminate excessive tank pressures due to wall boiling

- 3. Venting will be initiated as tank pressure peaks out (i.e., pressure riserate approaches zero). This should coincide with ullage and tank wall temperatures approaching the same value.
- 4. If liquid had been present in the tank prior to vent initiation, it is likely that
 - a. beiling would then occur due to the average wall temperature being greater than 200 K (360 R), and
 - b. tank pressure would not be leveling off due to boiloff.
- 5. It is highly unlikely that liquid hydrogen can reside in the tank for more than a short time interval, if the average tank temperature is 178K (320 R) greater than liquid temperature.

3.3.2.4 <u>Terminating Prechill.</u> One solution to the problem of determining propellant tank temperature at prechill termination is to monitor a large number of temperatures during this process. Many measurements will be needed because the tanks will not be prechilled at a uniform rate. It is likely that the various measurements could be integrated, with the aid of a computer, to arrive at an average propellant tank temperature. This approach may be unacceptable if telemetry requirements become excessive, or if the transient conditions yield an inaccurate average vehicle temperature.

A potentially useful method is to use the accumulated tank pressure increase during the charge and vent process for determining how much energy has been removed from the propellant tank.

The tank wall energy transferred to the propellant during prechill is approximately proportional to the tank pressure increase. Rearranging equation 3-7 provides the following relationship:

$$(uw_1 - uw_2) mw = mg_2 (ug_2 - h_L)$$

$$(3-9)$$

and

$$mg_{2} = \rho_{g_{2}} V_{T}$$

$$\rho_{g_{2}} = propellant tank vapor density$$
(3-10)

NOTE: This derivation assumes that the tank is initially evacuated.

Equation (3-9) shows that tank-wall energy-removal is a function of final vapor mass and vapor internal energy. However, both mass and internal energy are a function of final temperature and pressure. Consequently, tank energy removal can be related to these variables. This relationship is shown by Figure 3-17. There are two significant points to be made from this figure. First, energy removal is approximately independent of vapor temperature. Second, the energy removal is directly proportional to final tank pressure. It is concluded that the energy extracted during prechill can be approximated from the tank pressure increase experienced during the process, even if gas temperature is not known. The gas temperature uncertainty will **r**esult in an uncertainty in propellant vapor generated during prechill.

More propellant will be evaporated than the theoretical minimum required for tank prechill. This is true because gas temperature will be less than tank temperature when venting occurs. Fortunately, the additional vapor that may be vented does not appear to be excessive, as indicated by Figure 3-18. This figure indicates that even if the tank-to-gas temperature difference is as great as 50 K (90°R), hydrogen losses will be increased by only 5.4 kg (12 lb) per POTV stage, a small quantity compared to other losses that will be experienced during refueling operations.

Propellant transporation inefficiencies to be experienced during a POTV orbital refueling operation were determined during the Reference 1-1 study, and are given in Table 3-5. These inefficiencies were calculated for the orbiter-tanker configuration (Figure 3-1) in support of the five-day mission sortie. The listed prechill losses are two times greater than the theoretical minimum values. Even so, these quantities are insignificant and, as a result, it is concluded that a prechill procedure should not be selected on the basis of minimum fluid losses.

3.3.2.5 <u>Prechill Analysis</u>. A prechill subroutine, developed with IRAD funds, was employed to evaluate the details of a POTV prechill process. This subroutine which details are documented in Reference 3-3, was used to analyze fluid management system prechill for that study. The computer program is capable of evaluating prechill for the condition of vapor entry to the propellant tank, but not liquid entry. This limitation does not represent a handicap because, as explained in Section 3.3.2.2, liquid entry will have only a minimal influence upon the process.

The key to this analysis is in identifying the appropriate heat transfer coefficient between incoming vapor and the tank walls. Conventional forced convection expressions for flow over a flat plate or for jet impingement upon a surface, represent configurations that are too dissimilar to be applicable. Instead, a correlation developed for industrial mixing processes was selected as being representative of the heat exchange mechanism that will occur when vapor is continuously introduced into the propellant tank. The heat transfer correlation and required modifications are given as

$$\frac{h}{\rho c_{\rm p}} \left(N_{\rm PR} \right)^{2/3} = 0.13 \left(\frac{(P_{\rm i}/V) \,\mu}{\rho^2} \right)^{1/4} \quad (\text{Reference 3-3}) \quad (3-11)$$





Figure 3-17. Tank Pressure Increases Will Yield Total Energy Removal During Prechill



- Vent mass is based upon POTV tank prechill to 200 K (360°R).
- 2. $T_w = tank$ wall temperature at vent initiation.
- 3. T_{u} = ullage temperature at vent initiation.

Figure 3-18. Hydrogen Tank Prechill Vent Mass Is Not Excessive Even at Large Tank Wall to Ullage Temperature Differences

I 	nefficiency*, per _per_Shuttle_paylo	cent oad	
Tanker Kit			
Orbiter support	4.36		
Supply tanks	6.29		
Pressurization and transfer	2.63		
Liquid residuals	1.90	*Shuttle P/L capability =	
Vapor residuals	0.50	45, 454 kg	
Subtotal	15.68	**Dual stage DOTT tatal	
Transfer line chill	0.10	build stage POTV total boiloff losses are amortized over three Shuttle flights.	
POTV tanks prechill (tanks initially wa	rm) 0.28		
POTV boiloff** assume 29 days wait	0.74	5	
TOTAL	16.80		
Overall efficiency	83 ⁷ 7		

Table 3-5.Propellant transportation efficiency for on-orbit resupply
of dual stage POTV.

where

h	Ξ	heat transfer coefficient
ρ	=	fluid density
e p	H	constant pressure heat capacity
N _{PR}	=	Prandtl n um ber
P _i	H	mixer input power
V	=	tank volume
μ	=	fluid viscosity

Equation 3-11 was developed for liquids contained in cylinders. These liquids were continuously agitated with a mixing unit. Mixer input power was responsible for fluid agitation and is one of the variables of Equation 3-11. It is believed that fluid agitation during prechill will be the same (for equivalent power conditions) whether a mixer or fluid inflow is responsible. Since power output rather than power input will influence fluid agitation, equivalence will be between fluid power input and mixer power output.

This results in

$$P_{i}(EFF) = P_{o} = \dot{mv}^{2}$$
 (3-12)

where

 $P_0 = mixer power output$

EFF = mixer efficiency (conservatively assumed as 40 percent for this study) m = entering mass flow rate

v = entering fluid velocity

 \dot{mv}^2 = fluid power input

Substituting Equation 3-12 into 3-11 results in

$$\frac{h}{c_{\rm p}} (N_{\rm PR})^{2/3} = 0.163 \left(\frac{(m v^2/V) \mu}{\rho^2}\right)^{1/4}$$
(3-13)

Equation 3-13 indicates that heat transfer to the tank walls can be controlled by varying entering flowrate and velocity.

The following charge and vent procedure was selected for this POTV prechill analysis:

- 1. Charge the tank at a known vapor flowrate until the difference between wall and gas temperature has reached a specified value; 10 K (18°F) was the selected ΔT .
- 2. Vent the tank to a pre-determined low pressure. A reasonable level was selected as 6, 89 kN/m² (1.0 psia).
- 3. Charge and vent the tank as required to reduce tank temperature to the predetermined level.

Figures 3-19 and 3-20 give tank ullage pressure and wall temperature histories during the prechill period. These curves are based upon adding 9.08 kg (20 lb) hydrogen at 0.91 kg/sec (2.0 lb/sec) during the charge period. Note that the prechill, which includes two charge periods and one vent period, will be about 206 seconds in duration. Peak pressure for this procedure will not exceed 78.6 kN/m² (11.4 psia).

The influence of key prechill variables was assessed to determine if precisely known flow conditions would be required during this process. Figure 3-21 shows that velocity and flowrate variations of about 100% will alter prechill durations by about five to



Figure 3-19. POTV Liquid Hydrogen Tank Pressure History During Prechill







Figure 3-21. Mass flow rate and velocity influence upon liquid hydrogen tank prechill duration



Figure 3-22. Influence of tank wall-to-vapor temperature difference upon prechill duration

fifteen percent. Figure 3-22 indicates that large differences in tank wall-to-gas temperatures will have a minimal influence upon prechill duration. Finally Figure 3-23 shows that if the hydrogen charge mass were to be inadvertently increased from 9.1 kg (20 lb) to 27.3 kg (60 lb), peak pressures will still remain below the maximum allowable of about 172 kN/m^2 (25 psia).

3.2.3.6 <u>Summary.</u> Analysis of the propellant tank prechill process, in this and related studies, has led to several surprises, which are listed below:

- 1. Liquid oxygen tank prechill is not required because under no circumstance will excessive tank pressures occur during refueling operations. Thus emphasis was directed at the liquid hydrogen tank.
- 2. Rapid prechill of the hydrogen tank does not appear to be an important consideration. Figure 3-4 indicates that up to 64 hours of activity is required to support a single orbiter/POTV rendezvous and transfer operation, five percent of which may be required for propellant transfer. It seems evident that propellant transfer operations could be increased to 10 percent of the total timeline without significant impact. This is nearly two orders of magnitude more time than the approximate 200 second prechill time indicated by Figure 3-20.
- 3. Liquid hydrogen consumed for the tank prechill process will have an insignificant influence upon overall efficiency and cost of transporting propellants into space for POTV refuelling. As a result, propellant transfer efficiency should not be an important consideration in the prechill process selection.

It is concluded that the prechill process described and analyzed in Section 3.3.2.5 will satisfy the requirements of simplicity, reliability and safety.

3.3.3 RECEIVER TANK FILL. Tank fill will be initiated after the prechill requirements have been satisfied. The single requirement for tank fill is to maintain acceptably low pressure during the process. Tank pressures will be at a minimum if thermal equilibrium conditions are maintained during fill. Thermal equilibrium will be approached as heat and mass exchange between the phases is increased, which can be achieved by creating a highly agitated fluid condition. Given the assumption of a thermal equilibrium tank fill a simple relationship can be obtained between initial tank temperature at the start of no-vent chill and fill, incoming liquid vapor pressure, and final liquid vapor pressure. This relationship is derived below from the First Law of Thermodynamies for liquid flow into a closed container.

$$\frac{dE}{g} + \frac{dE}{L} + \frac{dE}{w} = \frac{h}{L} \frac{dm}{L}$$
(3-14)

$$dEg = (u_g m_g)_2 - (u_g m_g)_1 = change in ullage energy$$
(3-15)



Figure 3-23. Charge Mass Influence Upon Liquid Hydrogen Tank Peak Pressures

 Table 3-6. Baseline conditions selected for liquid hydrogen tank prechill procedure.

- 1. Initial Propellant Tank Temperature = 289 K (520R)
- 2. Initial Pressure = 6.9 kN/m^2 (1.0 psia)
- 3. Hydrogen vapor saturated at 103.4 kN/m² (15 psia) enters propellant tank
- 4. Entering flowrate = .91 kg/sec (2 lb/sec)
- 5. Entering velocity = 6.7 m/sec (22 ft/sec)
- 6. Hydrogen charge terminated after 9.1 kg (20 lb) enters tank
- Tank vent initiated when tank-to-ullage temperature difference becomes
 5.6 K (10R)
- 8. Propellant tank vented to 6.9 kN/m^2 (1.0 psia)
- 9. Vent area = 37.2 cm²(5.76 in.²)
- 10. Tank mass = 447.2 kg (986 lb)
- 11. Tank volume = 116 m³ (4100 ft³)

9

$$dE_{L} = (u_{L} in_{L})_{2} - (u_{L} m_{L})_{1} = change in liquid energy \qquad (3-1c)$$

$${}^{n}_{L} {}^{dm}_{L} = {}^{h}_{L} ({}^{m}_{L_{2}} - {}^{m}_{L_{1}}) = enthalpy change due to entering liquid (3-17)$$

ЧE ์พ'

$$= (u_{w}m_{w})_{2} - (u_{w}m_{w})_{1} = m_{w}(u_{2} - u_{1})_{w}$$

= (m_{w}\Delta u_{1}) = change is to 1 (3-18)

change in tank wall energy W w

For the assumption of an initially evacuated tank

$$m_{g_1} = m_{L_1} = 0$$

and

3

$$dE_{g} = u_{g_{2}} m_{g_{2}}$$

$$dE_{L} = u_{L_{2}} m_{L_{2}}$$

$$h_{dm} = 1$$

$$(3-19)$$

$${}^{L} {}^{Um}{}_{L} = {}^{h}{}_{L} {}^{m}{}_{L} {}^{(3-20)}$$

Combining Equations 3-14 through 3-21 we have

$$u_{g_2} m_{g_2} + u_{L_2} m_{L_2} m_{W_2} + m_{W_2} \Delta u_{W_2} = h_{L_2} m_{L_2}$$
 (3-22)

Solving equation 3-22 for h_L , results in $^{h}{}_{L}$

 $= u_{g_2} m_{g_2} / m_{L_2} + u_{L_2} + \Delta u_{w} m_{w} / m_{L_2}$ (3 - 23)

At thermal equilibrium h_L , u_{L_2} and u_{g_2} can be related to liquid vapor pressure and temperature. Consequently, u_{L_2} and u_{g_2} are known once final vapor pressure is specified and Δu_w is known since initial and final temperatures are given. Finally, h_{L} (which is a function of entering-liquid-vapor-pressure) can be determined for a

Equation 3-23 is summarized in Figures 3-24 and 3-25 which give entering liquid vapor pressure as a function of initial tank temperature and final tanked liquid vapor pressure for a 95 percent liquid fill condition. Note that final vapor pressure will be greater than entering liquid vapor pressure. This difference is due to the combination of initial tank wall energy and the heat of compression, which are released to the tank fluid during chill and fill. Figure 3-25 shows that final LO_2 vapor pressure will be about 6.9 kN/m² (1.0 psia) greater than entering vapor pressure. This small difference is due to the high propellant thermal mass. Final LH₂ vapor pressure will be

(3-21)



Figure 3-24. Final LH₂ Tank Pressures for Thermodynamic Equilibrium Fill Process (POTV and COTV Tanks).



about 20.7 kN/m² (3.0 psia) greater than entering vapor pressure, as indicated by Figure 3-24. All of the above is based upon the assumption that near-thermal equilabrium conditions will exist during tank fill, as will be discussed below.

3.3.3.1 <u>Tank refill (autogenous)</u>. Refill of a propellant tank containing liquid and its own vapor as a pressurant is a straightforward operation. Liquid must be introduced at the correct vapor pressure, and fluid inflow conditions must be sufficiently high to assure near-thermodynamic equilibrium conditions during fill. An evaluation will be made of the relationship between initial and final tank fluid conditions, and entering liquid vapor pressure for a thermal equilibrium process. The inlet vapor pressure is determined on the basis of the following First Law analysis

$$\frac{dE}{L} + \frac{dE}{g} = hdm \qquad (3-24)$$

$${}^{dE}_{L} = ({}^{u}_{L}{}^{m}_{L})_{2} - ({}^{u}_{L}{}^{m}_{L})_{1} = change in liquid energy \qquad (3-25)$$

$$dE_{g} = (u_{g}m_{g})_{2} - (u_{g}m_{g})_{1} = change in vapor energy$$
(3-26)

$$m_{T} = m_{L} + m_{g}$$

The following equalities result from the assumptions of phase equilibrium and initial liquid temperature equals final liquid temperature: $u_{L2} = u_{L1} = u_{L}$ and $u_{g2} = u_{g1} = u_{g}$. Combining these conditions with Equations 3-24 through 3-27 results in

$$u_{L} (m_{L2} - m_{L1}) + u_{g} (m_{g2} - m_{g1}) = h (m_{T2} - m_{T1})$$
 (3-28)

Now, total mass within the tank can be expressed as

$$m_{T} = V_{T} \rho_{L} - (\alpha - 1) m_{g}$$
 (3-29)

and

$$M_{T2} - m_{T1} = (\alpha - 1) (m_{g1} - m_{g2})$$
 (3-30)

1

where

$$m_T = m_L + m_g = \text{total mass of fluid in tank}$$

 $\rho_L = \text{liquid density}$
 $\rho_g = \text{vapor density}$
 $\alpha = \rho_L / \rho_g$
 $3-36$

Combining Equations 3-28 through 3-30 we find that

$$u_{L} (m_{L2} - m_{L1}) + u_{g} (m_{g2} - m_{h1}) = h (\alpha - 1) (m_{g1} - m_{g2})$$
 (3-31)

Also, by adding and subtracting $u_L m_{g1}$ and $u_L m_{g2}$ to the left side of Equation 3-31

$$u_L (m_{T2} - m_{T1}) + u_{ev} (m_{g2} - m_{g1}) = h (\alpha - 1) (m_{g1} - m_{g2})$$
 (3-32)

where

$$u_{ev} = (u_g - u_L) = internal energy of evaporation$$

Combining Equations 3-30 and 3-32

$$u_{L} (\alpha - 1) (m_{g1} - m_{g2}) + u_{ev} (m_{g2} - m_{g1}) = h (\gamma - 1) (m_{g2} - m_{g1})$$
 (3-33)

Solving for the entering liquid enthalpy,

h =
$$u_L - uev/(\alpha - 1)$$
 (3-34)

Results are plotted in Figure 3-26 showing liquid hydrogen tank final vapor pressure as a function of incoming liquid vapor pressure. It is interesting to note that this vapor pressure relationship is independent of initial propellant load at the start of refill. Furthermore, although Equation 3-34 was developed for an initial thermal equilibrium condition, results are also applicable to an initially superheated ullage.

The above results indicate that refill can be simplified because the same entering liquid vapor pressure will be required regardless of initial fill condition and ullage temperature. Figure 3-26 indicates that final tank pressure will be approximately 27.6 kN/m² (4 psia) higher than entering liquid vapor pressure.

The thermal equilibrium processes described above have demonstrated that tank fill (and refill) can be performed without having to vent. This is an important conclusion because selective vapor venting will not be possible once tank fill is initiated. Liquid venting must be avoided because the propellant loss and resulting disturbing torques could be unacceptably high. Consequently, it is mandatory that the refueling process be performed with a closed tank. The next step is to analytically describe the tank fill process in order to assess the influence of such key variables as entering flowrate and velocity upon thermal equilibrium. If thermal equilibrium conditions can be readily achieved for a reasonable range propellant flowrates and velocities, then orbital refueling, without venting, will be possible.

3.3.3.2 <u>Tank fill analysis</u>. The intent of the tank fill process will be to create turbulent conditions within the tank. These conditions will be achieved by introducing



Figure 3-26. Entering Liquid Hydrogen Vapor Pressure Required to Maintain a Constant Vapor Pressure in Tank During Fill

liquid into the tank at high velocities (and perhaps through a spray nozzle) to provide the high heat-transfer rates needed to attain near-thermal equilibrium. As tank-fill continues, the internal tank fluid environment changes from liquid droplets in the ullage volume to vapor bubbles entrained within a liquid bulk. It is expected that the transition from heat transfer dominated by liquid droplets to heat transfer dominated by vapor bubbles will occur in the range of 40% to 60% liquid fill. The mechanism of liquid spray in a vapor environment will change to that of vapor bubble entrainment and dispersal, within the liquid volume. This mechanism will be the dominant mode of heat and mass exchange throughout much of the tank fill process, and is the only mechanism that will influence tank pressures toward the completion of tank-fill.

The basis for any fill process is that sufficient fluid motion created within a propellant tank will maintain near thermal equilibrium. As equilibrium is approached, the pressure difference (ΔP) between tank pressure and liquid vapor pressure will approach zero, and absolute tank pressures during fill will approach a minimum. To aid in describing the phenomena that are expected to occur, tank fill experiment results are hypothesized and given in Figure 3-27. The figure illustrates two important points: First, tank ΔP becomes smaller as flowrate for a given tank configuration is increased. This is expected, because heat consfer coefficients will increase as flowrate



Figure 3-27. Flowrate influences tank pressure during fill.

increases. Second, there is a fill flowrate above which variations in ΔP become insignificant because fluid agitation has been increased to a level where near thermodynamic equilibrium already exists. The objective of this analysis will be to identify fluid inflow parameters that will provide near-equilibrium conditions during fill.

Fill Model for Liquid Spray Dominance.

During the early phases of tank fill, nearthermal equilibrium conditions will be achieved by introducing liquid into the

propellant tanks through spray nozzles. The resulting spray will create a large liquidvapor surface. The combination of large surface area and high droplet velocity will provide the high heat transfer rates needed to attain near-equilibrium conditions.

General Dynamics developed a computer program (HYPRES) on IRAD funds that deseribes the thermodynamic and fluid mechanic phenomena occurring during the early phase of a receiver tank fill process. This tank fill program was used to predict fluid management system fill pressures during a previous study. Filling of Orbital Fluid Management Systems, Reference 3–3. Subroutine DROP, of program HYPRFS, which describes the ullage-spray droplet interaction is given in Appendix C of that Reference. The equations contained in this subroutine indicate that droplet diameter, spray velocity and mass flowrate may have a major influence upon propellant tank pressures during the early stages of the refuelling operation. Consequently, a series of computer runs were made with HYPRES to evaluate these variables and to assess whether acceptably low liquid hydrogen tank pressures would be maintained. These results are summarized

Figure 3-28 shows the influence of droplet diameter upon tank pressure during the liquid spray dominance period. The range of droplet sizes considered include sizes that should be present during an actual refuelling operation. However, the figure indicates that droplet diameter will not have a major influence on tank pressure. Thus, it appears that droplet diameters outside the indicated range will have a minimal impact upon refuelling operations.

Figure 3-29 gives the influence of liquid spray volume upon hydrogen tank fill pressures. An input to HYPRES includes a variable (called PACK) which identifies the volume of liquid in droplet form that will exchange energy with the ullage. Specifically, PACK is the volume ratio of liquid to ullage, and this term is multiplied by ullage volume to obtained total liquid droplet spray volume. Figure 3-29 indicates that this


Ą

Figure 3-29. Influence of Liquid Spray Volume in Ullage Upon Liquid Hydrogen Tank Pressure During Fill

spray volume will have a greater influence upon pressures than droplet diameter. This influence, however, does not appear to be significant. Note that the two curves cross-over at about 4000 seconds (36% fill level. Beyond this time the case for PACK=0.2% predicts a lower pressure than the case for PACK=0.05%. This cross-over is probably due to the greater liquid-ullage heat exchange (for PACK=0.2%) which tends to maintain near-thermal equilibrium conditions.

Figure 3-30 shows the influence of propellant flowrate upon tank pressure. For this case pressures are plotted against propellant fill levels in order to normalize the influence of different tanking flowrates. It is seen that higher flowrates will result in slightly lower tank pressures during the early stages of fill, followed by slightly higher tank pressures later in the fill process. The lower initial pressures during the early part of tanking are caused by the quenching influence of the higher mass flow condition. The higher pressure rise rates that occur later in the tanking operation are due to an inability to transfer the higher the higher tank propellant.

Table 3-8 and Figure 3-31 serve to illustrate this point. The energy exchange required to achieve thermal equilibrium between ullage and propellant bulk is summarized in Table 3-8. This quantity for the POTV hydrogen tank is approximately 13.48 kW-hr (46010 Btn), and remains independent of refueling duration. The rate of heat exchange,



Figure 3-30. Mass Flow Rate Influence Upon Liquid Hydrogen Tank Pressure During Fill

 Table 3-7. Baseline conditions selected for liquid hydrogen propellant tank fill analysis.

- 1. Initial tank temperature = 200 K (360 R)
- 2. Tank mass = 447.2 kg (986 lb)
- 3. Tank volume = 116 m³ (4100 ft³)
- 4. Initial tank pressure = 0 kN/m^2 (0 psia)
- 5. Liquid hydrogen, saturated at 103.4 kN/m² (15 psia) enters propellant tank
- 6. Entering liquid flowrate = 0.68 kg/sec (1.5 lb/sec)
- 7. Entering liquid velocity = 3.05 m/sec (10 ft/sec)
- 8. Spray droplet diameter = 2540 μ (0.1 inch)
- 9. (Liquid spray volume in ullage/ullage volume) = .002

however, will be inversely proportional to tank fill duration, as illustrated by Figure 3-31. This figure indicates that the average heat exchange rate will be 13.48 kW (12.78 Btu/sec) and 6.74 kW (6.39 Btu/sec) respectively, for one hour and two hour tank fill durations.

Computer runs were also conducted to determine the influence of inlet velocity upon fill pressures. No plots are given because it was found that tank pressure variations will be insignificant over a velocity range of 3.05 m/sec (10 ft/sec) to 15.25 m/sec (50 ft/sec).

	Energy KW-hr(Btu)	Comments
Initial Ullage Energy	10.4 (35500)	Based upon tank wall temperature of 194.4 K (350R) at end of prechill
Heat of compression	3.33 (11380)	Compression heating due to fill process
Final ullage energy	0.25 (840)	Assumes 5% ullage volume at saturated conditions
Ullage energy removal requirements	13.48 (46010)	This energy must be transferred to liquid in order to attain thermal equilibrium by the end of tank fill.

Table 3-8.	Ullage cooling required to attain thermal equilibrium during POTV LH ₂
	tank fill.



Figure 3-31. Required Average Ullage-to-Liquid Heat Transfer Rate for Liquid Hydrogen Tank Refuelling Operation

Fill Model for Vapor Bubble Dominance. When the propellant tanks are approximately 50% full, the dominant heat exchange mode will be that of convection and condensation between the liquid bulk and entrained vapor bubbles.

In a previous study (Reference 3-3) equations were obtained which predict this heat exchange during tank fill. These equations (given below) include the influence of inlet fluid conditions, fluid properties and tank geometry.

$$\left(\frac{h}{Cp} \frac{\rho}{L}\right) \left(N_{pr}\right)^{2/3} = .163 \left[\frac{(\dot{mv}^2/V_L) \mu_L}{\rho_L^2}\right]^{1/4}$$
(3-35)

where:

h

'n

= heat transfer coefficient

C = constant pressure heat capacity

 $\rho_{L} =$ liquid density

$$N_{\Gamma}$$
 = Prandtl number

= entering mass flow rate

v	= entering liquid velocity	
$^{\mathrm{v}}{}_{\mathrm{L}}$	= liquid volume in tank	
$^{\mu}{}_{ m L}$	= liquid viscosity	
1.1	$134 \sigma^{0.6} \epsilon^{0.5}$	(0, 0, 0)
$d =\gamma$	$(10^{-2} (\text{mv}^2/\text{V}_{\text{L}})^{0.4} + .09$	(3-30)
where:		

d	=	bubble diameter, cm
σ	¥	surface tension, dyne/cm
E	=	entrained vapor volume to total vapor plus liquid volume
γ	×	liquid specific gravity

Note: These equations are from Reference 3-3 equations 3-53 and 3-33, respectively.

Equations 3-35 and 3-36 are modifications of empirically derived expressions from industrial applications for which vapor dispersal in liquid is essential to achieving a high rate of heat and mass transfer. This subject has been extensively studied and much of the work has been collected and summarized in Reference 3-4. The empirically derived expressions were applied to the refueling process by replacing the mechanical mixer power term, employed for industrial applications, with an equivalent fluid power expression, $\dot{m}v^2$. This conversion was accomplished with the following relationships:

Fluid Power = Mixer Power Output Mixer Power Output = Efficiency × Mixer Power Input \dot{mv}^2 = 40 Percent × Mixer Power Input

where 40 percent efficiency represents a conservative value.

The heat transfer rate between an individual **bubble** and the liquid bulk can be expressed as

$$\frac{Q_B}{V_B} = \frac{hA_B(T_B - T_L)}{V_B} = \frac{\pi d^2}{(\pi/6) d^3} h (T_B - T_L) = \frac{6h (T_B - T_L)}{d}$$
(3-37)

where

${}^{Q}_{B}$	=	heat transfer rate from each bubble
v _B	8	bubble volume
т _в	=	bubble vapor temperature
T _L	H	liquid bulk temperature

Total heat transfer rate from the total dispersed vapor volume can be determined from Equation 3-37 by introducing the total number of vapor bubbles, n,

$$\frac{Q_{T}}{V_{g}} = \frac{n\dot{Q}_{B}}{nV_{B}} = \frac{\dot{Q}_{B}}{V_{B}} = \frac{6h(T_{B} - T_{L})}{d}$$
(3-38)

where h and d are determined from Equations 3-35 and 3-36, respectively, and

$$Q_{T}$$
 = total heat transfer rate from the entrained vapor
 V_{g} = total vapor volume entrained in liquid

Equations 3-35, 3-36 and 3-38 were added to the HYPRES program to provide capability for evaluating tank fill conditions during the vapor-bubble dominance mode. A series of computer runs were conducted to evaluate this mechanism. Results are given in Figures 3-32 through 3-34 which are, respectively, a continuation of Figures 3-28 through 3-30. A transition from liquid spray heat exchange to vapor-bubble heat exchange was imposed at the 70 percent propellant fill level. A sudden pressure drop occurred coincident with this heat exchange transition. This pressure drop was obviously due to the increased heat transfer rate that created near-thermal equilibrium conditions.

The curves of Figures 3-32 through 3-34 require clarification, especially following the transition to vapor-bubble heat exchange mechanism. First, there is no indicated tank pressure change due to variations in spray droplet diameter or PACK factor. This occurs because heat exchange will be influenced only by entering flowrate, velocity, and entrained vapor volume, E, and these quantities are identical for each case plotted in Figures 3-32 and 3-33.

Figure 3-34 also shows a negligible difference in pressures, due to flowrate variations, following the transition in heat exchange mechanism. For these conditions, however, compensating factors may influence tank pressures: a) the high flowrate condition requires a high liquid-ullage heat exchange rate in order to maintain low pressures,







Figure 3-33. Influence of Vapor-Bubble Heat Exchange Mechanism Upon Hydrogen Tank Fill Pressures for Range of Liquid Spray Volume



Figure 3-34. Influence of Vapor Bubble Heat Exchange Mechanism Upon Hydrogen Tank Fill Pressures for Range of Tanking Flow Rates

and b) the high flowrate condition creates a high heat transfer coefficient, which serves to maintain low pressures. The negligible difference in tanking pressure indicates that the above factors were, in fact, compensating, or that the heat exchange mechanism is sufficient to assure near-thermal equilibrium conditions. The latter condition is a more likely possibility.

Results of this analysis supports the assertion, stated earlier, that heat and mass exchange due to vapor bubble entrainment and dispersal within the liquid volume will dominate. More important is the likelihood that this is the only mechanism that will influence tank pressure towards the completion of tank fill. Furthermore, it appears that heat and mass exchange rates will be of such a high magnitude that nearthermal equilibrium conditions will be maintained over a broad range of entering liquid flowrates and velocities. This is a significant conclusion because the thermal equilibrium parametric data of Figures 3-24 through 3-26 can be employed to predict tank pressure conditions at fill (or refill) completion.

3.3.3.3 <u>Supply tank influence</u>. All analyses to date have been based on the assumption that propellant enters the receiver tank at a constant temperature. This temperature corresponds to a vapor pressure of 103.4 kN/m² (15 psia). In fact, entering liquid vapor pressure (and temperature) will vary throughout the refueling process because of the supply tank pressurization method. Helium pressurization of the supply tank during propellant transfer was selected because it appears to have fewer complications than other approaches.

Helium will be bubbled into the propellant to effect boiloff, which aids in pressurization. This boiloff cools the liquid and reduced its vapor pressure during propellant transfer. Figures 3-35 and 3-36 show the resulting supply tank propellant temperatures and vapor pressure histories during outflow.

Figure 3-37 compares receiver tank pressure histories for the constant temperature and varying temperature conditions. Note that for the variable supply tank conditions the tank pressure rise-rates are not as great as for the constant temperature case. Furthermore tank pressure begins to decay after the receiver tank is 40 percent filled, whereas pressure continues rising for the other case. This behavior can be explained by referring to Figure 3-24, which shows that final vapor pressure is a function of entering liquid vapor pressure. Since the average entering vapor pressure will be lower for the variable supply tank conditions, final vapor pressure must also be lower.

There are two factors that control the increasing and decreasing pressures indicated by Figure 3-37; the heat of compression caused by propellant fill, and the steadily decreasing enthalpy of entering propellant. The heat of compression causes tank pressure to rise continuously, as indicated by the constant incoming temperature case. The second factor is responsible for pressure decay, as the lower temperature incoming liquid mixes with the propellant bulk. Evidently, the heat of compression is the greater influence during the early stages of fill, and the lower temperature liquid is dominant beyond the 40 percent fill condition.

It should also be emphasized that a single orbiter-tanker mission will not refuel the POTV propellant tanks. The Reference 3-1 study was based upon an orbiter payload capability of 45,360 kg (100,000 lb), which translates to a liquid hydrogen supply tank volume of 77 m³ (2720 ft³). This system is capable of refueling the POTV fuel tank to the 66.4 percent level, as indicated by Table 3-9. The liquid oxygen tank would also be filled to the same level. As Figure 3-2 indicated, a second tanker flight would refuel the second POTV stage, followed by a third flight which would top off both stages and deliver the crew module.

3.3.3.4 <u>Alternative refill concept</u>. An alternative concept of on-orbit refuelling was evaluated during the study. This concept, which requires introducing propellants through a diffuser to achieve extremely low entering velocities, is a major departure from the selected technique of a high velocity liquid spray. This alternative was based upon the premise that large quantities of liquid would enter the propellant tank before striking the opposite tank wall. The resulting boiloff would not create a high tank pressure because the propellant bulk would mix with vapor and maintain a 'ow pressure. In assessing the spray nozzle versus inlet diffuser configurations the following minimum set of requirements were considered:

1. Avoid excessive tank pressures throughout the prechill and fill processes



•

Figure 3-35. Supply Tank Liquid Temperature During POTV Refill



Figure 3-36. Supply Tank Liquid Vapor Pressure During POTV Refill



Figure 3-37. Supply tank LH_2 temperature influence upon POTV tank pressure during refill.

Table 3-9. Supply tank conditions during LH_2 tank POTV refill.

= 103.4 kN/m^2 (15 psia)
= 20.3 K (36.6°R)
= 77 m ³ (2720 ft ³)
= 3.5 percent
= 2 percent
$= 72.7 \text{ m}^3 (2570 \text{ ft}^3)$

* This is 66.4% of POTV LH2 tank volume.

2. Minimize vent mass requirements

Ķ

3. If venting is necessary, avoid two-phase venting

With regard to the LO_2 tank, any tank inlet configuration will be satisfactory. Venting will not be required during prechill; therefore, two phase venting is not a possibility. The only question that may arise is that of excessive tank pressure during the fill process. It is obvious from Figure 3-25 that moderately low pressure will exist for near-thermal equilibrium conditions. But these conditions will occur only if sufficient fluid turbulence is present to enhance heat and mass exchange between the phases. Fluid turbulence is more likely to occur with a spray nozzle than with an inlet diffuser, whose primary function is to introduce liquid into the tank at extremely low velocities. Thus, a spray nozzle inlet configuration would appear to best satisfy the LO_2 tank fill requirements.

A spray nozzle inlet configuration for the LH_2 tank will result in venting during the prechill process, as indicated by Figure 3-19. This is so because the spray velocity conditions will create a high heat and mass exchange environment between hydrogen vapor or liquid and the tank walls. However, two-phase venting will be avoided because liquid will not be present in the tank when venting is initiated. Once tank fill is initiated, the spray nozzles will create the turbulent fluid conditions that are beneficial to maintaining low pressures during fill.

The affect of an inlet diffuser upon hydrogen tank pressure during prechill is not quite so clear. Ideally, a low velocity diffuser will allow large quantities of propellant to be introduced into the tank before the leading edge of a larbe-diameter jet impinges at the opposite end of the tank. It has been hypothesized that a large propellant mass will serve to quench the pressure-rise that occurs due to liquid impingement upon the hot walls. However, the propellant mass can only be effective if sufficient heat exchange occurs with the vapor. Such conditions may not exist because of the deliberately low velocity of entering liquid. Thus, it is possible that an inlet diffuser configuration may not be able to satisfy the conflicting requirements of both a low velocity, (to assure a large propellant mass). Should verting be required because of insufficient heat exchange between liquid and vapor, liquid may be lost overboard in the process.

Once tank fill is initiated, the spray nozzle will be preferable to an inlet diffuser because of the *i* srbulent conditions that will serve to maintain low pressure

The choice is betweer a spray pozzle, which will result in venting during prechill, or an inlet diffuser, which may not require venting, but should it occur, could result in liquid lest overboard. The spray nozzle configuration was selected because prechill losses with this concept will be insignificant compared to other losses (Table 3-5). **3.** 3. 5 MECHANICAL MIXERS TO ASSIST PROPELLANT REFILL. The tank fill analysis of Section 3. 3. 3. 2 described how the key to a successful refill operation would be the turbulent environment created by the entering fluid. Work done on the ullage mass during propellant transfer will be convected to the

liquid via the heat exchange mechanism set up by fluid turbulence. If, however, the turbulent heat transfer rates should be inadequate, propellant transfer would have to be interrupted, in order to prevent the continuation of a high pressure rise rate. It is possible that a long time duration would then be required to effect a pressure reduction and to approch near-equilibrium conditions. This delay would occur because flow termination would also reduce the ullage-to-liquid heat transfer process. The consequence of an inadequate fluid turbulent environment could be an undesirably long tank fill process, caused by numerous flow interruptions.

Analyses to date indicate that near-thermal equilibrium conditions can be achieved for a wide range of tanking flowrates. Nevertheless, a backup position should be available in the event that of an excessive tank pressure increase. The solution would be to use a mechanical mixer to provide the additional fluid turbulence needed to achieve nearequilibrium conditions. Mixers will already exist since they are an integral part of the zero-g vent system required for each propellant tank.

3.3.5.1 <u>Mixer power relationship to liquid-ullage heat exchange</u>. The heat exchange mechanism that will exist during periods of mixer operation is that of vapor-bubble dominance described in Section 3.3.3.2. Equations 3-35 and 3-36 apply, except that the original expressions from Reference 3-6 were employed. That is, the equations included an input power term rather the fluid power term.

Figures 3-38 and 3-39 give ullage-to-liquid heating rates for hydrogen and oxygen as a function of mixer power, percent liquid fill, and vapor hold-up (i.e., the percent vapor entrained in liquid). Referring to Figure 3-31, it is seen that the required heating rate can be achieved for a power input of less than 4 watts. Note that input power requirements to achieve a given heating rate are a strong function of vapor hold-up and a very weak function of the percent liquid fill condition. According to Figures 3-40 and 3-41, vapor bubbles generated within the OTV propellant tanks by a mixer (or its equivalent in fluid power) are a function of fluid power for power levels less than about 12 watts.

The mixer power requirements identified by Figures 3-38 and 3-39 are within the range currently being considered for zero-g vent system mixers. Consequency, this propellant transfer assist will be available without expending additional resources.

3.3.5.2 <u>Mixer power/fluid power equivalence</u>. Analyses to date include an implied assumption that tank fill durations will be selected on the basis of heat and mass transfer considerations. That is, spray nozzle, flowrate, and velocity solutions will be made to assure thermal equilibrium throughout tank fill. It is possible, however, that flowrate constraints may be imposed by other factors. For example, fluid



oxygen heat transfer rate.





loads acting on the transfer lines and/or propellant tanks may preclude tank fill durations of less than five or six hours. It might not be possible to maintain low tanking pressures at these lower flowrate without contin. us mixer power assist. For this circumstance the required fluid turbulence would be provided with a combination of fluid power and mixer power. This combination could be selected from curves similar to those given in Figures 3-42 and 3-43. An example of how these figures can be applied is given below:

- 1. Assumptions
 - The fluid power equivalence of 15 watts mixer power is required during tanking to maintain near-equilibrium conditions.
 - Other considerations require that tank fill be performed in six hours using a 6.35 cm (2.5 inch) diameter equivalent nozzle inlet.
- 2. LH₂ Tank Solution
 - Fluid power = 2.75 watts
 - Mixer Power = 15.00 2.75 = 12.25 watts required
- 3. LO₂ Tank Solution
 - Fluid power = 1.75 watts
 - Mixer power = 15.00 1.75 = 13.25 watts required.

3.3.6 START BASKET REFILL. If propellant screen acquisition devices (start baskets) are included as POTV subsystems, an additional complication must be addressed in selecting an on-orbit refill concept, that of completely refilling the start baskets with-out trapping vapor. Vapor entrapment is unacceptable because pure liquid flow from the start basket is normally required, and cannot be guaranteed unless the screen device is free of vapor. Vapor will be present at some time within the start basket regardless of the method of propellant fill. Any trapped vapor bubbles must subsequently be condensed by using pressurant. The type of pressurant can either be helium (non-condensible) or vaporized propellant (condensible). The question of trapped helium bubbles will be discussed in Section 3.3.6.3.

Two methods of vapor bubble collapse were evaluated; passive and active. In each case the propellant tank will be pressurized to sub-cool the liquid surrounding the entrapped vapor, so that vapor condensation will occur. The passive method assumes that conduction heat transfer is the only mechanism that exists to cool and subsequently condense the vapor. The active method relies upon forced convection heat transfer to condense the entrapped vapor. An evaluation of each method follows.

3.3.6.1 Passive method of bubble collapse. Start basket designs, generated for POTV during the Reference 3-2 study, were analyzed to determine if vapor-free refill would be possible. A sketch and start-basket-dimensions are given in Figure 3-44 and Table 3-10. A very conservative assumption was made for this analysis. The start basket would be filled with a single large bubble at the end of the on-orbit







Figure 3-43. Fluid Power Input Equivalence to Mixer Power During OTV LO2 Tank Fill





	LH ₂ Tank	LO ₂ Tank
A. Radius, cm (in)	142.2(56.0)	85.1(33.5)
B. Standpipe height, cm (in.)	11.9 (4.7)	4.6 (1.8)
C. Cone height, cm (in.)	16.5 (6.5)	9.9 (3.9)
D. Ellipsoidal height, cm (in.)	42.2 (16.6)	17.5 (6.9)
E. Basket volume, m^3 (ft ²)	1,92 (67.8)	0.34(11.9)

Table 3-10. POTV start basket characteristics.

refill operation. The largest spherical bubble that can be trapped within each screen device was computed to be about 0.61 m (2 ft) and 0.3 m (1 ft). An analysis was conducted to determine the time for each bubble to collapse.

Bubble condensation times were determined using bubble collapse equations described in Reference 3-3. Computer output results are given in Figure 5-45 and 3-46 for hydrogen and oxygen, respectively. The oxygen bubble will collapse in about 5 minutes if it is subcooled by about 20.7 kN/m² (3 psid), which is an acceptable maximum helium partial pressure during refill. The hydrogen bubble will require about three to four hours to collapse at the same degree of subcooling, which may be an unacceptably long duration. These calculated collapse times should be greater than actual collapse times because of the conservatively large bubble sizes selected. It should also be mentioned that the bubble collapse predictions are subject to added uncertainty because the analytical model was developed for small diameter bubbles.

3.3.6.2 <u>Active method of bubble collapse</u>. If vapor-bubble collapse times which use the passive method are excessive, an alternative is to use an active method to greatly decrease bubble collapse times. This method requires that propellant be sprayed into the start basket during tank fill. The fluid agitation induced by entering propellant will create a high heat exchange mechanism equivalent to that occurring outside the start basket during propellant fill. Furthermore, this turbulent fluid condition will serve to create small vapor bubbles which will greatly enhance the condensation process.

A model has been developed to determine the conditions under which the condensation of all propellant vapor within a start basket will occur. The model is based upon the following assumptions:

- 1. Liquid inflow velocity and flowrate are known.
- 2. Only liquid will exit the start basket; vapor is removed only by condensation.
- 3. Liquid enters the basket at a constant temperature.
- 4. Vapor trapped within the basket is in spherical bubble form (Figure 3-47). Thus, this model is valid only for vapor volume conditions of about 40 percent or less by basket volume.
- 5. Bubble diameter is obtained from equation 3-36,

$$d = \frac{1.134 \sigma^{0.6} \epsilon^{0.5}}{\gamma^{0.2} (mv^2/V_L)^{0.4}} + .09$$



Figure 3-45. Collapse Times for Spherical Bubbles in Liquid Hydrogen



Figure 3-46. Collapse Times for Spherical Bubbles in Liquid Oxygen



Figure 3-47. Active Metnod for Start Basket Bubble Collapse where

d = bubble diameter, cm

 σ = surface tension, dyne/cm

 ϵ = entrained vapor volume to basket volume ratio

 γ = liquid specific gravity

 $V_{L} = (1 - \epsilon) V_{T} = \text{liquid volume within basket}$ $V_{T} = \text{basket volume}$

Substituting (1 - ϵ) V for V gives,

$$d = \frac{1.134 \sigma^{6} (1 - \epsilon) \cdot 4 \epsilon^{5}}{\gamma^{2} (mv^{2}/V_{T})^{4}} + .09$$
(3-39)

Figures 3-48 and 3-49 give plots of hydrogen and oxygen bubble diameter as as a function of $\dot{m}v^2/V_T$ and ϵ .

- 6. Bubble condensation rates are extremely high for the small bubble diameters anticipated (see Figures 3-50 and 3-51). Consequently, it is expected that liquid within the basket will rapidly saturate at tank pressure due to absorbing the heat of condensation.
- 7. All liquid leaving the basket will exit saturated at tank pressure.
- 8. Vapor bubble pressure = tank pressure throughout tank fill.

An analytical model that includes the above list of assumptions is given below:



Figure 3-48. Hydrogen Vapor Bubble Diameter During Start Basket Refill.



Figure 3-49. Oxygen Valor Bubble Diameter During Start Basket Refill.



Figure 3-50. Maximum Allowable Bubble Diameter for Condensing Hydrogen Vapor in Start Basket During POTV Propellant Tank Fill



Figure 3-51. Maximum Allowable Bubble Diameter for Condensing Oxygen Vapor in Start Basket During POTV Propellant Tank Fill

First Law

$$(mu)_{2L} + (mu)_{2V} - (mu)_{1L} - (mu)_{1V} = h_i \Delta m_i - h_o \Delta m_o$$
(3-40)

i.

where

m	=	mass
u	=	internal energy
h_i	=	enthalpy of entering liquid mass increment
h	=	enthalpy of exiting liquid mass increment
Δm_{i}	=	liquid mass increment entering tank in time,
Δm _o	=	liquid mass increment leaving tank in time,

subscript

2	=	conditions at end of interval
1	æ	conditions at start of interval
L	=	liquid
v	=	vapor

Now, $u_{2v} = u_{1v} = u_{v}$ (vapor is saturated at tank pressure, assumption 8) Also, u_{2L} $= u_{1L} = u_{L}$ (assumption 6)

Therefore,

$$(m_2 - m_1)_L u_L + (m_2 - m_1)_v u_v = h_i \Delta m_i - h_o \Delta m_o$$
(3-41)

From continuity,
$$m_1 L + m_1 v + \Delta m_i = m_{2L} + m_{2v} + \Delta m_0$$
 (3-42)

or,
$$m_{2T} - m_{1T} = \Delta m_{BASKET} = \Delta m_1 - \Delta m_0$$
 (3-43)

where
$$m_T = m_L + m_v$$
 (3-44)

 Δm_{BASKET} = propellant mass increment accumulated in start basket

Adding and subtracting $m_{2v}^{\ u}L$ and $m_{1v}^{\ u}L$ to equation 3-41 results in,

$$(m_{2T} - m_{1T}) u_{L} + (m_{2v} - m_{1v}) u_{ev} - h_i \Delta m_i - h_o \Delta m_o$$
 (3-45)

where

$$u_{ev} = (u_g - u_L)$$

But,

$$m_{2v} - m_{1v} = -\frac{(m_{2T} - m_{1T})}{(\rho_{L} / \rho_{g})^{-1}}$$
(3-46)

Therefore, equation (3-45) becomes

$$(m_{2T} - m_{1T}) (u_{L} - \frac{u_{ev}}{(\rho_{L} / \rho_{g}) - 1} = h_{i} \Delta m_{i} - h_{o} \Delta m_{o}$$
 (3-47)

Substituting equation (3-43) into (3-47) gives,

$$(\Delta m_{i} - \Delta m_{o}) (u_{L} - \frac{u_{ev}}{(\rho_{L}/\rho_{g}) - 1}) = h_{i} \Delta m_{i} - h_{o} \Delta m_{o}$$
 (3-48)

Rearranging terms and solving for $\Delta m_0/$ Δm_i we have,

$$\frac{\Delta m_{0}}{\Delta m_{i}} = \left[h_{i} - u_{L} + \frac{u_{ev}}{(\rho_{L}/\rho_{g}) - 1}\right] / \left[h_{0} - u_{L} + \frac{u_{ev}}{(\rho_{L}/\rho_{g}) - 1}\right]$$
(3-49)

However, from equation (3-43) we can show that

$$\frac{\Delta m_{\text{BASKET}}}{\Delta m_{\text{i}}} = 1 - \frac{\Delta m_{\text{o}}}{\Delta m_{\text{i}}} \quad \text{or,}$$

$$\frac{\Delta m_{\text{BASKET}}}{\Delta m_{\text{i}}} = 1 - \left[h_{\text{i}} - u_{\text{L}} + \frac{u_{\text{ev}}}{(\rho_{\text{L}}/\rho_{\text{g}}) - 1}\right] / \left[h_{\text{o}} - u_{\text{L}} + \frac{u_{\text{ev}}}{(\rho_{\text{L}}/\rho_{\text{g}}) - 1}\right] (3-50)$$

Equation (3-50) gives the maximum fraction of entering propellant that will accumulate within the start basket as a result of vapor condensation. This fraction is shown in Figure 3-52 for hydrogen and oxygen as a function of the differential pressure term, ΔP (tank pressure minus entering liquid vapor pressure). These curves indicate that an increasing fraction of entering liquid remains in the baske as ΔP increases. An increase in ΔP is equivalent to liquid entering the basket at cooler temperatures, which increases its heat absorbing capability. This in turn will increase the rate of vapor condensation, and subsequent liquid accumulation within the basket.

According to Figure 3-52, 100% of LO_2 entering the basket will remain within the basket, if the entering liquid is subcooled by about 6.8 kN/m² (1.0 psid), or greater. For this condition one could select a flowrate (m_o) that would just fill the basket at tank fill completion (Figure 3-53). A conservative, and preferred, approach would be to provide



$$\dot{\mathbf{m}} = \dot{\mathbf{m}}_{\mathbf{0}} / (\Delta \mathbf{m}_{\mathbf{B}} / \Delta \mathbf{m}_{\mathbf{i}})$$

$$\dot{m}_{c} = \underline{maximum basket liquid mass}$$

tank fill duration



Figure 3-52. Influence of liquid subcooling upon start basket refill flow parameters.

a greater basket flowrate (\dot{m}) to assure basket refill even at ΔP 's less than 6.9 kN/m² (1.0 psid). As a first approximation it is suggested that $\dot{m} = 3 \dot{m}$. It is seen from the curve that 'he basket will completely refill for this flow condition at ΔP 's greater than 1.7 kN/m² (0.25 psid).

Hydrogen properties are such that a considerably greater ΔP is required to accumulate the same fraction of LH₂ entering a basket than if LO₂ was entering a basket. This is illustrated by the hydrogen curve of Figure 3-52, which shows that about 40 kN/m² (5.7 psid) is needed for all entering liquid to remain within the basket. If, as above, we select $\dot{m} = 3 \dot{m}_{o}$, then basket refill will occur for a ΔP greater than 10 kN/m²

3-65

(1.4 psid). This condition should be satisfied since Figure 3-24 shows that final LH_2 equilibrium tank pressure for POTV will be about 21 kN/m² (3 psid) greater than entering LH_2 for initial tank wall temperatures less than 200K (360°R). The degree of liquid subcooling will be even greater if non-thermal equilibrium conditions occur during tank fill.

The above analysis has indicated that liquid inflow to a start basket is a feasible means of providing basket refill during the propellant tank fill process. A conservative thermodynamic mechanism has been analyzed which indicates that a reasonable range of propellant inflow rates will provide basket refill for POTV. In fact, a flowrate of $\dot{m} = 3 \dot{m}$ should result in a completely filled start basket at the end of propellant tank fill.

3.3.6.3 <u>Summary</u>. The liquid-oxygen tank start basket will be vapor free within onehalf hour after being pressurized by 20.7 kN/m² (3 psid). This same passive approach may require a considerably longer time for the liquid hydrogen tank. An alternative active-method was considered for the hydrogen tank in order to reduce the time required to condense all vapor. This condition will be satisfied at the end of propellant tank fill if hydrogen flows into the start basket at three times the rate indicated by Figure 3-53. This method requires the additional complication of a small diameter line plumbed between the main fill duct and the start basket. A preliminary design sketch of this set-up is given in Section 3.4.



Figure 3-53. Propellant Flowrate Required to Fill Start Basket During Tank Fill Operations

3.4 ORBITAL PROPELLANT TANKING OPERATIONS

The impact of the on-orbit refill upon vehicle design is described in this section. Conceptual design modifications are presented which depict propellant tank, transfer line and helium system modifications. A detailed discussion of an on-orbit refill procedures is given, including how the procedures are influenced by the presence of helium and; or start baskets.

5.4.1 CONCEPTUAL DESIGN MODIFICATIONS FOR ON-ORBIT REFILL. The analyses of Section 3.3 have established requirements for achieving POTV on-orbit refill from an orbiter-tanker. Requirements which affect the vehicle design include:

- a. A fluid spray circuit through which propellants are introduced into the propellant tanks.
- b. A bleed line to route LH_2 to the start basket during refill.
- c. A non-propulsive vent system through which propellant vapor is vented during tank blowdown.

These and other plumbing modifications are depicted in Section 3.4.1.1. Section 3.4.1.2 and 3.4.1.3 includes sketches of recommended transfer line designs and a modularized helium pressurization system.

3.4.1.1 Propellant tank modifications. Figures 3-54 and 3-55 show some tankage system arrangements for a two stage POTV. Propellant refill is accomplished from the shuttle, therefore, the reaching capability of the transfer lines is limited to the length of the Remote Manipulating System (RMS). We have the choice of 1) loading each stage separately, followed by interstage attachment maneuvers, 2) having the stages initially interconnected and equipped with external plumbing kits or 3) adding a motor driven hinge system at the POTV separation plane so that the vehicle can be "jack knifed" to within reach of the RMS. It is assumed that each stage will be tanked separately, thus avoiding the crossing of the separation plane with propellant ducts. The systems shown are common to each stage and include propellant fill circuits vent systems and acquisition devices for both the fuel and oxidizer tanks.

The fuel tank fill circuit starts at the skin line of the POTV body structure with disconnect assembly containing an internal poppet closure, a static seal interface, and a cone section which serves as an alignment tool and a structural attachment with the mating half. A flexible duct section is routed from the disconnect to a fuel tank inlet fitting. The duct material is Cres and incorporates three axially restrained flex joints which permits length and angular changes between the duct ends.

The tank inlet fitting is equipped with a flange (inside the tank) for receiving a distribution duet, and a boss for attaching a 3/4-inch bleed line. The distribution duet runs along the cylindrical section of tank and is equipped with two outlets containing spray



Figure 3-54. POTV Tankage Systems

۰,



Figure 3-55. POTV Tankage Systems

•

nozzles. Two nozzles are shown in Figure 3-54. Although it is not possible to analytically determine how many nozzles will be required to assure adecuate fluid mixing, it is believed that two or three nozzles will be satisfactory.

The 3/4-inch bleed line will direct LH₂ into the start basket to assure that all initially trapped vapor will be condensed by the completion of propellant fill.

The fill system for the oxidizer tank is basically the same as that for the fuel tank except the internal distribution duct has been replaced with a short tapered duct section equipped with a spray nozzle at the end. A bleed line to the acquisition device is not shown because calculations indicate that all vapor initially trapped within the basket will be condensed without resorting to liquid spray inflow.

Vent systems for both the fuel and oxidizer tanks consist of one vent valve located outside the tank wall, a non-propulsive vent duct, a thermodynamic vent system mounted off the forward bulkhead inside the tank, and a small vent tube for the thermodynamic vent system. The non-propulsive vent duct is routed from the vent valve to two points at the skin line and spaced 180° apart. The thermodynamic vent tube is supported from the vent duct and terminates near the duct ends.

3.4.1.2 <u>Transfer line design</u>. The transfer system is basically two separate circuits (one for LO_2 and one for LH_2) running from the POTV loading point back through to the orbiter supply tanks. Each circuit has a movable transfer line section connected to a fixed tank outlet line. The basic plumbing material is 304L cres.

Each movable transfer line consists of two sections interconnected with a tangential rotary joint. The forward section is equipped with a disconnect system for connecting to the POTV. The lines are deployed by the Shuttle RMS which engages with the disconnect valve. The transfer lines are supported at the base through tangential rotary joints attached to a support bridge forward of the fuel tank. In the stowed position the lines have additional support from the tank girth rings. Each line is also equipped with pneumatic tubes and electrical cables for actuating and monitoring the disconnect valve.

A disconnect arrangement for the POTV side only is included in Figure 3-56. The disconnect is designed so that the task of the RMS is to position the mating half approximately within the alignment cone. This mating half (attached to the RMS) has latch systems for gross capture followed by "draw down" and final alignment. The RMS, therefore, does not react disconnect loads. The arrangement shown is passive and consists of a cone and a disconnect assembly equipped with an internal spring loaded poppet and a flat external land for sealing with the mating half. The disconnect is attached to the cone using a ball-socket type fitting which permits angular misalignments. The mating disconnect assembly has an electromechanical drive which positions a pressure sensitive static seal against the flat land. The seal arrangement is shown in detail "C" of the figure. The electromechanically driven assembly is basically a short



Figure 3-56. Typical vehicle propellant disconnect arrangement.

telescopic tube sealed externally with a bellows and actuated externally by three screw jack actuators or by a single threaded collar with a worm drive. GD/C has designed a liquid flourine tandem butterfly valve for zero spillage at separation using such techniques. For this application, the telescopic section would have an internal poppet closure similar to that shown in Figure 3-56.

In summary, a typical connection starts with the RMS positioning the transfer line disconnect inside the cone. Angular, axial and lateral misalignments are permitted at this time. The second step is to actuate the locking pawls on the transfer line disconnect to an inboard position which captures the flange on the cone. This is a loose engagement only between cones. On the third step the pawls are actuated parallel to the disconnect center line which pulls the two cones together thus completing the structural connection. Fourth, the motorized disconnect carrying the external seal is actuated pressing the seal against the flat land. Upon further actuation, the poppet on the transfer line disconnect forces the poppet shown in the layout to an open position. Fluid transfer is now permitted. For disengagement, the above steps are reversed. After fluid transfer both poppets are closed before the external seal is disengaged from the flat land. Also the poppet interfaces are designed for near zero spillage.

A detailed discussion of the design and operation of a transfer line disconnect valve is given in Section 5.4.1.

3.4.1.3 <u>Helium system design.</u> Space-based vehicles will require gaseous helium resupply during refueling operations. One method of resupply is by flowing helium from a storage tank (located in the Shuttle payload bay) to the vehicle through a long transfer line. However, analyses indicate that helium bottle charge operations would

be lengthy if excessively high bottle temperatures are to be avoided.

The transfer line will have an electrical power cable attached to the line and permanently connected to the disconnect system located at the end of the line. The transfer line must have mobility which in turn requires flex joints either in the form of braided hoses, swivel type joints with dynamic seals, or loop bends in the tubing. Conventional bellows joints would not be applicable due to high pressure requirements. The use of hoses, swivel joints and loop bends presents packaging and weight penalties. Also, the safety of this transfer operation is of concern because an approximate 20 m (60 ft) line, pressurized to 34500 kN/m² (5000 psi), will be essentially unsupported except at the orbiter and at the vehicle. Consequently, an alternative method of resupply was selected.

An alternative would be to employ separate pre-loaded modular helium bottles that would be externally attached to the vehicle. A disconnect system capable of mating a structural attachment and a fluid connection with the vehicle is required with either approach. Several disconnects are required of the modular bottles, however, as compared to only one for a transfer line.

The orbiter remote manipulating system (RMS) will be employed to connect transfer line or modular bottles to the vehicle. The RMS will place the disconnects reasonably close to the mating target. Pressure-area loads will be reacted only by the disconnect structure, and not by the RMS.

The helium bottle modules shown in Figure 3-57 are intended for use on POTV's, COTV's and LTL's serviced from the shuttle. Only the module station locations would be changed. Basically, the pre-loaded helium modules are picked up with an RMS and plugged into external OTV interfaces which in turn have interconnecting plumbing to the OTV systems.

Detail "B" of Figure 3-57 shows a typical helium module which is a high pressure bottle equipped with a tangential support skirt, a docking cone, a latching system, a shutoff valve and a motor driven disconnect. The tangential skirt section has a fitting which interfaces with the RMS. The only requirement in the case of a Shuttle RMS would be the addition of an end fitting. This fitting would include a power cable from the Shuttle for actuating the latch and disconnect systems.

A typical transfer sequence consists of connecting an RMS to the helium module; placing the module loosely (wide tolerances) into a mating cone on the OTV; actuating the latches to an inboard position to insure a gross capture and finally moving the latches in an axial direction which completes the structural connection. With the structural connection completed, the disconnect system is energized making a seal between module and vehicle. The socket portion of the disconnect contains the seals and the probe section on the OTV is float mounted to compensate for misalignments. Since temperatures are basically ambient, the seal system consists of "O" rings equipped with backup rings to prevent "blow out." The seal design includes provisions for each replacement.





.

3-73

3.4.2 ORBITAL PROPELLANT TANKING OPERATIONS. On-orbit refueling of a dual stage POTV will require three Shuttle flights, each having a payload capability of 45, 360 kg(100,000 lb). The first Shuttle flight will transfer propellant to the second POTV stage. The second Shuttle flight will transfer propellant to the second stage. The third flight will carry the POTV payload and sufficient propellants for topping the two stages prior to mating and launch.

Refill operations will be influenced by type of POTV subsystems and by whether it is the first or subsequent shuttle flight for the mission. These influences will first be evaluated before a detailed refueling procedure is presented.

3.4.2.1 Subsystem influence upon refill procedures. Two subsystems that will have a direct effect upon refill operations are the pressurization and propellant acquisition (start basket) systems. If helium is required for vehicle tanks pressurization, the refill process will become complicated because of the need to first expel the pressurant. The presence of propellant tank start baskets require procedural changes to accommodate the requirement that the screened volumes be vapor-free at the end of tanking. Table 3-11 indicates what elements of an orbital refill process will be affected by these variables. It is evident from this table that a vehicle having no helium pressurant and no start basket will be the simplest to refuel in space. The most difficult combination would be the inclusion of the two subsystems. This latter combination presents a potentially serious concern that is addressed below.

Start basket/helium combination. This subsystems combination can result in helium entrapment within the start basket at refill completion. Helium entrapment is a serious concern because 1) the start basket must be vapor-free in order function properly during the OTV mission and, 2) the entrapped helium, unlike propellant vapor, cannot be removed by condensation. The best solution is to perform a series of propellant tank blowdown modes, similar that described by Figure 3-13, until a negligible helium quantity remains. This would be a time-consuming process and is best performed prior to orbiter-rendezvous.

3.4.2.2 Shuttle flight influence upon refill procedure. Three shuttle flights will be used to refill the two POTV stages. Refill procedures for the arst two shuttle flights will be identical because each stage will be empty. Thus the decision on how to handle liquid residuals, tank vent, and tank prechill will be the same. The third shuttle flight will provide the remaining propellants to complete tank fill. Obviously, the above tasks will not be performed. It will only be necessary to chill the transfer line prior to resuming the tank fill process. Transfer line chilldown may be performed differently for the third shuttle flight than for the first two flights. With the first two shuttle flights it is likely that vapor generated during chilldown will flow into the empty propellant tanks. Vapor generated during transfer line chilldown of the third shuttle flight will likely be vented overboard rather than into the propellant tanks.

		No Start F	Basket	Start Basket	
	ltem	No Helium	Helium	No Heijum	Helium
Lic	uid Residual				
•	Dump liquid prior to rendezvous	No	Yes	No	Yes•
Та	nk Vent				
•	Tank Temperature >200K (360R)	Yes	Yes	Yes	Yes*
•	Tank Temperature < $200K$ (36 ∂R)	No	Yes	No	Yes*
Tank Prechill					
•	Tank Temperature > 200K (360R)	Yes	Yes	Yes	Yes
•	Tank Temperature < 200K (360R)	No	No	No	No
Tank Fill					
•	Provide start basket bleed line	N.A.	N. A.	Yes	Yes
•	Pressurize tanks for start basket vapor collapse	N. A.	N. A.	Yes	Yes

 Table 3-11.
 Subsystem Influence Upon Refill Procedures

*A potential concern is that helium may be trapped within start basket. Solution is discussed in Section 3.4.2.1.

3.4.2.3 <u>Tank fill procedures</u>. On-orbit tank fill/refill procedures will be influenced by many variables including subsystem selection (see Table 3-11) and propellant (the oxygen tank requires no refill). It is not practical to prepare a table itemizing the individual procedural steps for each type of operations. Rather, a fill procedure has been developed for the most complicated case, which includes:

- a) Propellant tanks, helium pressurization
- b) Start baskets
- c) Initial tank temperatures at 289 K (520R).

A total propellant transfer operation of three hours was selected. Table 3-12 shows that this transfer duration can be accomplished by over-lapping LH_2 and LO_2 transfer operations.

Table 3-12. Propellant Transfer Can be Accomplished Within Three Hours


POTV propellant transfer timelines for orbiter flights 1 and 2 are given in Table 3-13. In addition to the prechill and tank fill events, these timelines include the transfer line operations of attaching, venting, purging, disconnecting and returning to orbiter. LO_2 transfer operations are initiated 49 minutes into the LH_2 transfer operations. This overlapping is required because a single RMS is employed for both propellant tanks. LH_2 transfer timelines are determined for the tank fill conditions of Table 3-14, using the analysis of section 3.3. LO_2 transfer timelines are estimates extrapolated from the LH_2 analysis. A total of 176 minutes is required to complete propellant transfer.

POTV propellant transfer timelines for orbiter flight 3 are given in Table 3-15. The primary difference between this flight and the earlier flights is that 1) the propellant tanks are about 75 percent full, and 2) each propellant tank of each stage will be filled, which requires increased RMS activity. A total of 122 minutes is required to complete transfer operations.

The procedures listed in Tables 3-13 and 3-15 do not include all fluid transfer operations required of on-orbit refill. It is likely that helium and earth storable propellants 1 also be transferred. Transfer operations for these fluids were identified during the "Orbital Propellant Handling and Storage Systems Definition Study", Reference 3-1. An evaluation of these operations indicated that transfer times of 162 minutes and 222 minutes respectively, will be required for the modular helium bottles and N₂H₄ storage system refill. Additional details and assumptions related to the transfer of helium and N₂H₄ are given in Reference 3-1.

3.4.3 ZERO-G MASS GAUGING. One of the more serious problems with any propellant transfer process is to determine when to terminate tank fill. For the geosynchronous mission, each receiver tank must be filled to about the 97% level. Tank overfill must be avoided because of the potential for tank overpressure, but tank underfill could jeopardize mission success. Estimating propellant mass loaded with flow-meter measurements will not be sufficiently accurate. It is expected that a zero-g mass gauging device will be required. Liquid-level sensors will be useless in a low-g environment, because liquid and vapor will be distributed throughout the tank. Yet some technique for accurately measuring propellant mass quantities in this environment is essential to on-orbit propellant transfer.

3.4.3.1 <u>Current Mass Gauging Devices</u>. Technology studies and development engineering has previously been performed on the following mass gauging devices:

- 1. The TRW developed system based on absorption of gamma radiation.
- 2. The General Nucleonics radiation absorption system using Krypton 80 as the source.
- 3. The Bendix system, which uses a radio frequency (RF) technique.

-	Elapsed Time	Cumulative Time
Event	(min)	(min)
TRANSFER LH2 ⁽¹⁾		
Vent OTV LH ₂ tank	15	()
Attach LH ₂ transfer line	15	15
Vent LH ₂ transfer line	1	23
Close LH ₂ transfer line	_	24
Close LH_2 tank	_	24
LH ₂ transfer and	- (2)	24
thermal hold	0	29
Vent LH ₂ tank	_c (2)	
Close tank	5	34
LH ₂ transfer and	c (2)	34
thermal hold	O T	39
Vent LH ₂ tank		
Close vent	_	44
Transfer LH ₂	0.0	44
Switch to topping rate	90 10	134
Sense vapor flow in supply tank	-	144
Close valves	_	144
Vent LH ₂ transfer line	-	144
Purge LH ₂ transfer line with He	3 5	147
Close LH ₂ transfer line vent	- -	152
Disconnect LH ₂ transfer line	1	152
Purge LH ₂ transfer line with He	2	153
Return LH ₂ transfer line to Orbiter	2	155
TRANSFER LO (2)	0	163
Attach LO ₂ Attach LO ₂	_	49
Vent LOo transfer line	7	56
Close LO, transfer line work	1	57
Hold		57
Vent OTV LOs tank	2	59
Clove LOo tank	20	79
Transfer LOo		79
Switch to topping mto	60	139
Sense vapor flow in supplie (10	149
Close valves	-	149
Hold	-	149
Vent LOo transfer line	8	157
Purge LOs transfer line with Ha	3	160
Close LO ₂ transfer line vent	5	165
Disconnect LOo transfer line		165
Purge LOo transfer line with the	1	166
Return LO2 transfer line to Orbiter	2	168
nelines are based upon f	0	

Table 3-13. Propellant transfer timeline (Flights 1 & 2).

(1) Timelines are based upon flow conditions given in Table 3-14.
 (2) Timelines are estimates.

Prechill ⁽¹⁾	
LH ₂ flowrate LH ₂ velocity Tank Fill ⁽²⁾	<pre>= .45 kg/sec (1.0 lb/sec) = 3.05 m/sec (10 ft/sec)</pre>
LH ₂ flowrate LH ₂ velocity Topping ⁽³⁾	<pre>= .91 kg/sec (2.0 lb/sec) = 6.10 m/sec (20 ft/sec)</pre>
LH ₂ flowrate LH ₂ velocity	<pre>= .45 kg/sec (1.0 lb/sec) = 3.05 m/sec (10 ft/sec)</pre>
(1) Prechill duratic (2) Tank fill duratic 3-34.	ons of Tables 3-13 and 3-14 are acceptable per Figure 3-21.
3-34.	ns of Tables 3-13 and 3-14 are acceptable per Figures 3-30 and

Table 3-14. LH₂ Propellant Transfer Flow Conditions Selected for POTV Refill Operations

The RF type, whose development was also undertaken by the National Bureau of Standards, has an accuracy of about ± 3 percent but tends to be geometry and configuration sensitive. The nucleonics type, developed by TRW and General Nucleonics, uses a radiation/ detector device and has an accuracy of about $\pm 2\%$. Resistance to the operational use of this system has been reported because of the potential radiation hazard. If the radiation were not a problem the system could be a very viable approach.

Although much progress has been made with these devices, system verification remains to be demonstrated on large scale systems in a zero-gravity environment. Until such time that one of the above devices has been proven for orbital propellant transfer, a search for alternative methods should continue.

	Elapsed Time	Cumulative Time
Event	(min)	(min)
TRANSFER LH2 ⁽¹⁾		0
Attach LH ₂ transfer line to 2nd stage	8	
Vent transfer line	1	0
Close transfer line vent	_	9
Slow flow into transfer line	1	10
Transfer LH_2	25	10
Sense tank LH_2 mass	_	30
Switch to topping rate	10	30
Sense tank LH ₂ mass	_	40 40
Close LH ₂ valves	-	45
Attach LH_2 transfer line to 1st stage	8	40
Transfer LH_2	25	53
Sense tank LH ₂ mass		78
Switch to topping rate	10	78
Sense tank LH ₂ mass	_	88
Close LH ₂ valves		88
Vent LH ₂ transfer line	3	88
Purge LH ₂ transfer line with He	5	91
Close LH ₂ transfer line vent	_	96
Disconnect LH ₂ transfer line	1	96
Purge LH ₂ transfer line with He	2	97
Return LH ₂ transfer line to Orbiter	8	99
TRANSFER LO ₂ ⁽²⁾	Ū	107
	-	10
Attach LO_2 transfer line to 2nd stage	7	17
Vent transfer line	1	18
Close transfer line vent		18
Slow flow into transfer line	1	19
Fransfer LO_2	15	34
Sense tank LO_2 mass		34
Switch to topping rate	10	44
Sense tank LO_2 mass	-	44
Close LO ₂ valves	-	44
Hold	11	55
Attach LO ₂ transfer line to 1st stage	8	63
Transfer LO_2	15	78
Sense tank LO_2 mass	-	78
Switch to topping rate	10	88
Sense tank LO_2 mass	-	88
Close LO ₂ valves		88
Hold	13	101
Vent LO ₂ transfer line	3	104
Purge LO ₂ transfer line with He	5	109
Close LO_2 transfer line vent		109
Disconnect LO_2 transfer line	3	112
Purge LO ₂ transfer line	2	114
Return LO_2 transfer line to Orbiter	8	122

Table 3-15. Propellant transfer timeline (Flight 3).

(1) Timelines are based upon flow conditions of Table 3-14.

(2) Timelines are estimates.

3.4.3.1 <u>Thermodynamic Mass Gauging</u>. Thermodynamic mass gauging is an alternative approach that may be useful for orbital refuelling operations. This approach relies upon measuring the tank pressure increase resulting from a known helium mass introduced into a nearly full propellant tank. Ullage volume can be determined from these measurements in combination with a knowledge of propellant temperature. The tanked propellant mass can then be determined from the product of calculated liquid volume and liquid density. The following procedure, and its attendant assumptions, is considered reasonable for zero-g mass gauging operations (liquid hydrogen tanking is assumed, but the approach is also applicable to liquid oxygen):

- 1. Conduct tank prechill and fill as described in section 3.2 and 3.3.
 - Rely upon a flowmeter (assume $\pm 8\%$ accuracy) to provide gross indications of propellant loading.
 - Terminate propellant transfer at an 88% fill indication. The actual tanked quantity will reside between 80% and 96%. This will prevent the possibility of tank overfill.
- 2. Turn on thermodynamic vent system (TVS) mixer to achieve and maintain thermodynamic equilibrium.
 - •. At equilibrium, vapor and liquid reside at the same temperature. Temperature can be accurately determined from tank pressure.
 - Assuming a 2.1 kN/m² (0.3 psi) error on absolute pressure (measurement), the temperature error will be about 0.06 K (0.1 R), which is equivalent to a 0.3% error for tank conditions of 138 kN/m² (20 psia).
- 3. Initiate helium pressurization after tank pressure has stabilized (i.e. pressure rise rate ≈ 0). Figure 3-58 depicts the anticipated propellant tank pressure and volume excursions during this period.
 - Measure P_{He,} after tank pressure has stabilized
- 4. Ullage volume determined from Equation of State calculations

•
$$V_{u_1} = (\Delta m \times R \times T_L) \neq P_{He_2}$$
 (3-51)



TANKING ERROR = ROOT-SUM-SQUARE (RSS) OF $\delta V_{u_1}/V_{u_1}$ and $\delta \Delta V_L/\Delta V_L$

- PRESSURE RISE RATE DURING TANK FILL
- LIQUID HYDROGEN VAPOR PRESSURE DURING FILL PLH2 ∆P
- PRESSURE DECAY MAY OCCUR IF NON-EQUILIBRIUM CONDITIOUS EXIST WHEN MIXER IS TURNED ON
 - PRESSURE RISE DURING HELIUM PRESSURIZATION П PHE1
 - HELIUM PRESSURE AT EQUILIBRIUM н
 - HELIUM PRESSURE AT END PROPELLANT FILL H PHE2 PHEf
- ULLAGE VOLUME MEASURED FROM THERODYNAMIC MASS GAUGING ii In.1
 - = DESIRED ULLAGE VOLUME AT END PROFELLANT FILL Juf ∆VL
- PROPELLANT VOLUME ADDITION TO ATTAIN FINAL LIQUID FILL

Figure 3-55. Propellant Tank Conditions for Thermodynamic Mass Gauging Operations

3-81

where:

 T_{L} = liquid temperature

• The inaccuracy of the ullage volume calculation can be determined with the following error analysis on (3-51):

$$\frac{\delta V_{u}}{V_{u}} = \frac{\delta \Delta m}{\Delta m} + \frac{\delta T_{L}}{T_{L}} + \frac{\delta P_{HE}}{P_{HE}}$$
(3-52)

ı

By taking a root-sum-square (RSS) of the above variables (which is the accepted approach), we have

$$\frac{\delta V_{u_1}}{V_{u_1}} = \pm \sqrt{\left(\frac{\delta \Delta m}{\Delta m}\right)^2} + \left(\frac{\delta T_L}{T_L}\right)^2 + \left(\frac{\delta P_{HE_2}}{P_{HE_2}}\right)^2$$
(3-53)

where:

$$\Delta m = \pm 7\%$$
 (assumed helium mass flow accuracy)

$$\frac{\delta T_{L}}{T_{L}} = \pm 0.3\% \text{ (from item 2 above)}$$

 $\delta P_{HE}^{=} \pm 0.1$ psia (estimate based upon Centaur experience with 2 high accuracy transducers)

Calculate the required liquid volume to achieve a final fill of $95^{\rm e}_{\rm d0}$

•
$$\Delta V_{L} = (V_{u_1} - V_{u_f})$$

where:

$$V_{u_{f}}$$
 = ullage volume at 95% liquid fill
 ΔV_{L} = LH₂ volume addition to achieve tank fill

• The inaccuracy in providing ΔV_{L} is,

$$\frac{\delta \Delta V_{L}}{\Delta V_{L}} = \pm 8\% \text{ (flowmeter accuracy)}$$

- 6. Re-initiate tank fill and introduce ΔV_{L} propellants to achieve a final propellant load.
 - Propellant loading error can be defined as

$$\frac{\delta m_{L}}{m_{L}} = \frac{\rho_{L} \delta V_{L}}{\rho_{L} V_{L}} = \frac{\delta V_{L}}{V_{L}}$$
(3-54)

where:

$$\frac{\delta V_{L}}{V_{L}} = \pm \left(\frac{\delta V_{u_{1}}}{V_{L}}\right)^{2} + \left(\frac{\delta \Delta V_{L}}{V_{L}}\right)$$
(3-55)

<u>Propellant Loading Error</u>. Equations 3-51, 3-53 and 3-55 were solved for liquid fill conditions of 80% and 88% at the start of mass gauging operations. Additionally, a $P_{\text{He}2}$ range of 3.4 kN/m² (0.5 psi) to 7.6 kN/m² (1.1 psi) was assumed during mass gauging. Results are summarized in Figure 3-59. Note that tanking error can vary between 1.5% and 4.4% with this mass gauging method.

It is evident from Figure 3-59 that tanking error can be reduced by increasing P_{He_2} . The only limit to this increase is that the resulting P_{He_3} at tank fill completion cannot be excessive. Figures 3-60 and 3-61 display P_{He_3} as a function of final propellant fill and P_{He_2} . It is seen that P_{He_3} can become as great as 62 kN/m² (9 psi).

3-83



Figure 3-59. Thermodynamic Mass Gauging Tanking Error

A more acceptable pressure level would appear to be about 28 to 34 kN/m^2 (4 to 5 psi). If a propellant fill goal of 95% to 97% is considered acceptable, the Figure 3-60 indicates that the maximum acceptable P He is about 5.2 kN/m^2 (0.75 psi). 3 According to Figure 3-59, this will result in a minimum tanking error of 1.9% to 3.2%.

<u>Mass Gauging Concerns.</u> The success of the thermodynamic mass gauge rests upon two key factors; confidence in the liquid mass flowmeter, and the ability to maintain thermodynamic equilibrium with the TVS mixer during the mass gauging operations.

This technique depends upon a mass flowmeter to provide a gross indication of tank fill. Thus, there must be sufficient confidence in this instrument that propellant transfer will not be terminated until a high fill condition is indicated. If such confidence does not exist, the alternative would be to terminate propellant transfer at the 70% or 60% fill indication. An early flow termination could increase tanking error to the point of rendering this technique useless.

The second important factor is that of maintaining near-thermodynamic equilibrium conditions. Without this guarantee, ullage temperature uncertainty would increase the uncertainty in δV_{u_1} . Such as increase would result in a corresponding propellant loading uncertainty. It appears, however, from the analysis of section 3.3 that near-thermal equilibrium conditions will not be difficult to maintain.

In spite of the above concerns, it appears that thermodynamic mass gauging is a viable technique that warrants further consideration.









4

COTV ORBITAL RESUPPLY

In this section a mission scenario will be developed for the Cargo Orbital Transfer Venicle (COTV) concept selected in Section 2. A mission will be defined as an aid in identifying the key issues of orbital refill operations. These operations will include all major activities from post-mission return to LEO through resupply. Vehicle and depot subsystem orbital refill requirements will also be identified. Operational procedures and techniques for propellant transfer will then be developed. Attention will focus only on COTV differences with POTV refill operations, because many aspects of orbital propellant transfer operations will be similar or identical.

4.1 MISSION SCENARIO

Large space industrialization programs have been conceived for the 1990's and beyond. One of the most ambitious is the Solar Powered Satellite (SPS) program that will require propellant quantities several orders of magnitude greater than those identified for the near term. In this SPS project Heavy Lift Launch Vehicles (HLLV) will be developed to transport personnel, material and propellants into low earth orbit (LEO). Liquid hydrogen and liquid oxygen will be transported in large supply tanks containing approximately 436 metric tons (960,000 lb) at a 6 to 1 ratio. These supply tanks will be transferred directly to depots in LEO, and empty tanks will be returned to earth for refuelling. Depot ancillary equipment will include reliquefiers for reclaiming stored propellant boiloff, and large solar arrays for converting solar power to the electrical energy needed to operate the orbital facility.

Personnel and material will be transported from depots to higher orbital altitudes to support industrialization programs. Personnel will be transported in POTV's; material (or cargo) will be dispatched to work sites (perhaps GEO) in a common stage COTV. Conceptual studies have been performed to identify COTV configurations, some of which were discussed in Section 2.

4.1.1 SELECTED COTV MISSIONS. A two stage COTV will fly a round trip mission to dispatch its payload to GEO. Total mission duration will be approximately five days; the first stage will return to LEO a short time after the vehicle is launched. The second stage will return to LEO following the five day round trip mission to GEO. Upon return, each vehicle stage will enter a phasing orbit preparatory to depot rendezvous. Several revolutions of this orbit (~ 3.1 hours per revolution) may be required before rendezvous is attempted. Rendezvous and docking maneuvers will consume approximately two hours. Post-docking operations will include cost effective procedures to reclaim all propellant liquid and vapor practicable. This is the primary area where substantial differences between POTV and COTV operations will occur because the depot ancillary equipment can be employed to minimize propellant loss. The impetus for such procedures is derived from the relatively high cost of transporting propellants into space.

4.1.1.1 Timelines. Timelines have not yet been developed for COTV operations. However, a previous study "Orbital Propellant Handling and Storage Systems for Large Space Programs," Reference 4-1, indicated that a launch schedule of ten or more flights per year is conceivable. Therefore, a rapid turnaround time between missions is expected. Subsystems will be inspected and tested to verify flight worthiness once post-mission operations are complete. A safed condition will exist once liquid propellants have been transferred and tank pressures stabilized. Pressure stabilization will be relatively easy to manage once liquids have been transferred.

4.1.2 ORBITAL DEPOT CONFIGURATION. Orbital depot conceptual designs were developed under Contract NAS9-15640 (Reference 4-2) for Johnson Space Center (JSC). That study concluded that large scale space activities would benefit from orbital propellant depots such as shown in Figure 4-1, which illustrates a five-storage-module depot, with optional crew quarters and maintenance hangars, refueling a manned OTV. Basic features include capillary propellant acquisition systems so that no rotation or thrusting is necessary to position propellants. Reliquefaction systems eliminate long term boiloff losses.

Benefits of such a depot are:

- a. Reliquefaction eliminates boiloff losses.
- b. Operations are more flexible, accommodating launch delays and OTV waiting.
- c. Operating costs are reduced for probable 1990's OTV missions.

Investment considerations are:

- a. Reliquefaction with power and radiators is a major item.
- b. Station keeping uses significant amounts of propellants.
- c. Propellant delivery tanks may also be used for storage.
- d. Docking provisions, solar power array, and radiators are common to any space base.

4.1.2.1 Ancillary equipment. Reliquefiers will be permanently plumbed to the supply tank farm for continuous reliquation of propellant boiloff. Studies conducted in Reference 4-2 indicate that reliquefiers will be cost-effective in contrast to the alternative of transporting additional propellants from earth. Since they require electrical



Figure 4-1. Orbital Propellant Depot

power for operation and radiators for heat rejection, electrical power will be provided using solar arrays that will directly convert solar energy. Figure 4-2 provides size and cost data on reliquefiers, solar arrays and radiators obtained from the previously mentioned study. It is apparent that the depot physical configuration will be dominated by the solar array and radiators. These data are for reliquefaction rates of 3.5 kg/hr (7.7 lb/hr) LH₂ and 7 kg/hr (15.6 lb/hr) LO₂.



Figure 4-2. Schematic: ET Propellant Depot

Two phase pumps. The ability to readily transfer propellants from supply to receiver tank is a necessary requirement for orbital depot operations. It has generally been assumed that pressure-fed systems will be employed to effect on-orbit transfer of propellants. There are instances, however, when this approach appears impractical. An excellent example of this occurs when transferring residual propellants from COTV to the depot supply tanks. The mass required to pressurize the vehicle tanks will be of the same magnitude as the residual liquid to be transferred. A preferred, and cost effective approach will be to use pumps for transferring residual propellants, rather than a pressure-fed system. Such pumps require two phase pumping capability because bulk boiling occurs within the propellant tanks and saturated propellants are expelled. The absence of buoyancy during on-orbit propeliant transfer may allow considerable vapor flow, and pumps will have to be designed for this flow condition.

4.1.3 COTV CONFIGURATION. The basic COTV tank parameters are given in Figure 4-3. In this basic structure are included those subsystems needed to achieve COTV mission and orbital refill requirements. These subsystems are pressurization, insulation, thermodynamic vent system, non-propulsive vent system, propellant

1	F i	·	Stage 2		Stage 1
Payload		LH ₂ Tank	LO2 Tank	LH ₂ tank	Lo tank

Parameter	Fach Store			
Physical	rach Stage			
Length, m (ft) Diameter, m (ft) Weight, kg (lb)	48 8.4 251,750	(157.5) (27.6) (555.000) wet		
Tankage	15,880	(35,000) we		
LH Capacity, kg (lb) LO ₂ Capacity, kg (lb) Material Insulation	37, 050 198, 820 2219 M	(81,690) (438,300) 9 - T87 A1 1LI		

Figure 4-3. COTV Characteristics

management (start basket), an advanced attitude control system (AACS), and an advanced main engine system. Only the advanced attitude control and main engine systems will be substantially different from the POTV systems described in Section 3. Neither system will have a major influence upon orbital refill operations.

4.1.3.1 Advanced attitude control system. The AACS will be designed to extract liquid hydrogen and liquid oxygen propellants from the main tanks. This system will be employed for all precision maneuvers required during the COTV mission and the post-mission docking operations. However, the AACS will not be a factor once dock-ing is complete.

4.1.3.2 Advanced main engine. The advanced main engine system will be able to operate at zero net positive suction pressure (NPSP) and have "boot-strap" capability. That is, the vehicle will be able to achieve main engine start without benefit of pre-pressurization or propellant settling. When the engine valves are opened, propellants will flow from the acquisition devices to their respective main engine inlets and generate a low thrust. This low thrust will result in a higher flow rate demand, which in turn will generate a higher thrust level until the engine has "boot-s.rapped" to steady state operating levels.

Because this advanced engine will preclude pre-pressurization for main engine start (MES), helium will not be present in either liquid hydrogen or liquid oxygen propellant tank. This condition can greatly simplify refill operations, as was discussed for POTV. However, in order to provide a more thorough discussion of orbital depot refill operations, a helium pressurization system will also be assumed for COTV. The following variations to the basic vehicle configuration will be analyzed in this section.

- 1. No helium pressurization and no start basket.
- 2. No helium pressurization with start basket.
- 3. Helium pressurization and no start basket.
- 4. Helium pressurization with start basket.

4.1.3.3 COTV/POTV procedural differences. Refill operations for all four configurations will be identical to their respective POTV configurations analyzed in Section 3. No further analysis is required for COTV, although curves will be provided depicting the influence of prechill and tank fill flow parameters upon each process. Only the post-mission operations will differ from those previously discussed in Section 3. These operations will focus upon reclaiming residual propellant vapor and liquid using the depot ancillary equipment; there will be no provisions for reclaiming propellant residuals on POTV missions.

4.2 POST MISSION DE- TANKING OPERATIONS

Once vehicle docking is complete and the payload/crew module is secure, post mission de-tanking operations will be initiated. One or π ore of the following reasons may be justification for initiating propellant de-tanking in a timely manner.

If it is feasible to reclaim propellants -

- Return liquid to the better insulated supply tanks in order to minimize boiloff. Reliquefy propellant vapor and return to supply tank, thereby reclaiming vapor for

If a helium pressurization system is used (i.e., helium is in propellant tanks) -

Helium must be expelled before refill can be initiated. This requires that tank propellants be expelled.

- If an autogenous pressurization system is used (i.e., no helium) -
- Propellants will be expelled if vehicle component replacement requires that • propellant tanks be "opened-up". As an example, vent valve replacement can be performed conveniently only if tank pressure resides at or near zero.

It is seen from the above that propellants should almost always be transferred from the COTV shortly after docking. An exception to this would be if a) there was no advantage to reclaiming residual propellants, b) the vehicle stage had autogenous pressurization, and c) component replacement would not be performed between vehicle missions. Under these conditions, propellants would remain aboard, gradually evaporate and be vented. There would be no compelling reason for transferring liquid propellart because all maintenance, except for major component replacement, could be performed with liquid on-board.

The remaining discussion and analysis will deal with the question of when it is cost effective to reclaim residual propellants, and when these propellants should be dumped overboard. Because helium can complicate procedures, operations for an autogenous pressurization system will be evaluated separately from helium system operations.

4.2.1 OPERATIONS FOR AUTOGENOUS PRESSURANT. A totally autogenous pressurization system for COTV will only be realized for an advanced engine system with zero NPSP capability. Propellant vapor will be tapped off from the engine system for tank pressurization during "transient" and steady state operation. Consequently, only liquid and its vapor will reside within the propellant tanks following a mission. This condition reduces the complexity of propellant transfer operations.

4.2.1.1 The cost of propellant dumping - The cost of dumping residual propellants overboard rather than reclaiming them is determined by the cost of transporting the equivalent propellant mass from earth. Residual propellant quantities for each COTV stage are itemized in Table 4-1, which totals for two stages are 5,896 kg (13,000 lb). Propellant transportation costs during the 1990's are expected to be in the range of 22\$/kg (10\$/lb) to 44\$kg (20\$/lb). The total estimated costs of dumping propellant is given in Figure 4-4, and can be as great as 30 million dollars for a ten year period. The cost of producing propellants was not included because these costs will represent less than one percent of propellant transportation costs.

-		ł			LILOO ICBIC	luais.
	Tank	Final MECO	$Vapor^{(1)}$	Vapor	$Liquid^{(2)}$	Total
	Volume,	Pressure,	Density	Residual	Residual,	Residual
	$m^3(ft^3)$	_kN/m ² (psia)	$kg/m^3(lb/ft^3)$	kg(lb)	kg(lb)	kg(lb)
H ₂ Tank LO ₂ Tank	548 (19360) 183 (6460)	103.4 (15) 103.4 (15)	6.62 (.085) 22.18 (.285)	747 (1648) 835 (1840)	195 (430) 1171 (2581)	942 (2078) 2006 (4421)

Table 4-1. COTV propellant tanks final MECO residuals.

(1) saturated vapor

(2) assumes 0.5% of total propellant load.



Figure 4-4. Cost of Replacing Dumped COTV Residuals

4.2.1.2 Propellant reclamation — Ideally, it will be desirable to first transfer all liquid from COTV, and then deal with the question of reliquefying the propellant vapor. However, pure liquid transfer from COTV to the depot supply tanks will be extremely difficult to achieve because liquid-vapor distribution, created by the near zero-g environment, may be unknown. Consequently, it is necessary to consider two-phase fluid transfer rather than the separate operations of liquid phase transfer followed by gas phase transfer. There are three alternatives available for transferring residual propellants to depot supply tanks. These are

- Case 1 transfer a two phase fluid directly to the supply tank.
- Case 2 transfer the fluid to a reliquefier; the resultant pure liquid can then be transferred to a supply tank.
- Case 3 Connect propellant tanks vent plumbing to the supply tank and reliquefy boiloff on a continuous basis.

A schematic of each alternative, with a brief discussion of advantages and disadvantages, is given in Figure 4-5. The most costly approach will be Case 2 because substantially larger reliquefiers are required if the residual vapor is to be recondensed during the transfer process. This higher flow rate requirement translates into larger reliquefiers, radiators and solar array. Case 3 is unacceptable because propellant tank pressures will remain high for the entire period between launches, thereby violating the groundrule to reduce tank pressures to a low level in a timely manner. Case 1 will have the shortest duration for propellant transfer since the only limitation on transfer flow rate is pump size. This case will be analyzed in greater detail.

<u>Case 1 Evaluation</u>. A compressor will be required for propellant transfer to the supply tank because receiver tank pressure will decay to a low level during this process. Since compressors may not operate satisfactorily under mixed phase conditions, a vent tube will be needed to probe the ullage such that a high percentage of vapor by volume is expelled at all times. Once fluid transfer is complete, the vapor is processed through the supply tank reliquefiers over a long time period. The time required to complete reliquefaction will depend upon fluid residual mass and reliquefaction capacity.

It must be recognized that reliquefaction is not cost-free. The supply tank reliquefaction systems will be sized to handle a design boiloff rate. Reliquefactor of propellant residuals represents a load over and above the design conditions. Consequently, separate units or added capability for existing units is required. In either case, the additional cost must be borne as the cost to reclaim residual propellants. These costs are explored in the following paragraphs.

<u>Reclamation Costs.</u> The rate at which propellant quantities are reliquefied will depend, in part, upon the available residual mass and the time available for reliquefaction.



Estimated residual propellant quantities have been given in Table 4-1. These propellants should probably be reliquefied in a time comparable to the interval between COTV launches; which is about 30-36 days for the selected scenario of 10-12 launches per year. Figure 4-6 gives reliquefaction rates as a function of time for the COTV hydrogen residuals. (Note: This discussion will focus on hydrogen because its reliquefaction costs are considerably greater than for oxygen.)



Figure 4-6. Reliquefier Capacity-Time Requirements for Reliquefying COTV Hydrogen Residuals

The cost of hydrogen reliquefaction units and their associated capacities are given in Table 4-2 for two unit sizes. These costs were determined from Reference 4-1 and do not include development costs; only the cost of an additional unit is assessed. The smallest unit has a hydrogen reliquefaction capacity of 9.5 kg/hr (21 lb/hr) which is capable of reliquefying COTV hydrogen residuals in about 14 days. This unit is an adequate size for propellant reclamation, and its 16.6 M\$ cost represents the cost of reclaiming propellants. Figure 4-7 indicates that about five to ten years of operation may be required to recover reliquefier costs.

4.2.1.3 Residuals for RCS propellants — The orbital depot will consume substantial quantities of hydrogen and oxygen for drag makeup and attitude control requirements. The annual usage rates were estimated in Reference 4-1 at 22608 kg (49, 872 lb)



Figure 4-7. Cost of dumping COTV hydrogen residuals compared to cost of reclaiming vapor residual.

hydrogen and 48144 kg (106,104 lb) oxygen. Propellant residuals (from 12 COTV launches/year) could provide up to 40 percent of the annual RCS propellant needs if the residuals can be transferred to suitable storage tanks. That is, the fluids must be transferred and stored in vapor form in order to avoid the high cost of reliquefaction. Performance of an oxygen/hydrogen vapor feed system, is shown in Figure 4-8. A vapor feed system appears to be feasible because performance degradation is somewhat insensitive to chamber pressure (which reflects storage pressure conditions). It **is also** estimated that storage temperature variations will have minimal effect on RCS engine performance.

۲,



		5 M lb Ca	Syster to Rel of CO lant Re	System Dedicated to Reliquefaction of COTV Propel- lant Residuals		
Cost Element	Size	Dev.	Proc	5 M lb Capacity	10 M lt y Capacity	o v
Structure	15 K lb	13.23	0,06	1 _		4
Avionics/Software	500 lb	24.48	2.61	_	-	
Solar Array	33.3 m^2	-	0.01	0.01	0.02	
Electrical Power System	1000 lb 1500 lb 2200 lb 300 lb	6.26	0.95	0.95	1.90	
Reliquifiers*		5.23	3.30	3.30	6.60	
Radiators		16.32	6.69	6.69	13.38	
RCS System	400 lb	0.54	0.20	0.20	0.40	
Subtotal	400 10	<u>6.56</u>	2.62			
Floating Items		72.71	16,44	11.15	22.30	
Initial Spares		27.04	5.59	3.79	7.58	
Initial Transportation			2.47	1.67	3.34	
		100.35	24.71	<u> </u>		
				10.01	33.22	I.

Table 4-2. Propellant Depot Facility Cost Estimate (Table 9-2, Reference 4-1) (Millions of 1979\$)

* 9.5 kg/hr (21 lb/hr) GH_2 reliquefaction and 29.5 kg/hr (65 lb/hr) GO_2 reliquefaction.

- (1) Hydrogen reliquefaction requirements represent major portion of costs.
- (2) No development costs for solar array. It is assumed that development costs have been borne by the Solar Powered Satellite (SPS).

Two disadvantages have been identified for an oxygen/hydrogen vapor feed system. These are: a) the need for compressors to transfer residual propellants to the RCS storage tanks, and b) the increased storage tank mass and volume required for vapor storage over that of liquid storage.

These disadvantages appear to be minimal as explained below:

- 1. Compressors and/or considerably more complex equipment will be required for the reliquefier approach.
- 2. The increased RCS storage tank weight should be small in comparison to the residual mass reclaimed annually.

4.2.2 OPERATIONS FOR HELIUM PRESSURANT. Reclaiming COTV residual propellants can be a more complicated process, than described in Section 4.2.1, if helium is present in the propellant tanks. The helium will have been introduced during the vehicle mission to provide NPSP for each main engine start. As discussed in Section 3, helium must be removed prior to a refill operation because excessive partial pressure may result for a fully loaded tank condition. Steps must also be taken to prevent helium from entering the start baskets during refill because, once present, that helium will remain until expelled during a vehicle mission. Pure liquid flow from the start baskets is a requirement; consequently, helium trapped within a start basket is unacceptable.

These potential helium problems will be avoided with POTV by expelling virtually all propellant residuals and helium from the propellant tanks through a blowdown process. Several blowdown periods are contemplated to provide assurance that helium is diluted to an acceptably low level. This approach is acceptable for POTV because equipment and facilities will not be available in the orbiter to reclaim propellant residuals. But, since it can be assumed that the appropriate facilities and equipment will be an integral part of the orbital depot, and evaluation is required to determine what options are available, and which option is acceptable. Table 4-3 identifies the problems (caused by helium) associated with the COTV post-mission operations; possible solutions are summarized for each problem. The following three solutions (which are identical to those of Section 4.2.1) will be discussed:

- 1. Propellant dump to space (with no attempt to reclaim residuals).
- 2. Reliquefy propellant vapor and return to the depot supply tanks.
- 3. Transfer residuals to depot RCS propellant tanks for subsequent reuse for drag makeup and attitude control requirements.

4.2.2.1 The cost of propellant dumping — The cost of dumping propellants from a COTV will be the same whether the vehicle includes a helium pressurization system or an autogenous pressurization. This is because propellant tank total residuals will

Pressurization System)	
(Helium	
On-Orbit Refueling of a COTV	
Table 4-3.	

Requirement No. 1: Refull partially full LM2 and LO2 propellant tanks containing belium pressurant in the ullage.	VEHICLE HARDWARE REQUIREMENTS	ADVANTAGES	DISADVAN'I AGES
Problem No. 1. Refill attempted without venting will result in an excessive belium partial pressure.			
Solutions: 1-A Expel belium by venting the tank to a low pressure prior to initiating refill.	None . But must have capability of maintairing vent valves open down to low pressures .	Simple approach.	Will lese approximately 1353 kg GH2 and 160° kg GO ₂ plus all propellant residual during process.
1-B Dilute helium concentration by introducing a clowed cycle purge procedure	Plumbing mods for closed cycle purge.	No propellant vapor or liquid will be lost.	Complex and costly space 4k pot equipment required, Proc- ess takes weaks to complete.
1-C Pump tank contents to depot RCS propeliant tanks. Use for depot drag makeup and attitude control requirements.	None	Propellants will provide about 40% of depot RCS requirements. Minimum cost.	Helium may degrade H ₂ ⁽ O ₂ engine ISP.
Problem No. 2: COTV tank venting on-orbit will include provellant vapor and possibly liquid proverlant. How can liquid be recovered brefore a venting or purge procedure is inditated?			
Solutions: 2-A Orient the vehicle such that acrodynamic drag will 2-A Orient the vehicle such the start basket. Slowly drain liquid from the tanks until all available liquid is reclaimed.	Plumbing mods to drain liquid from propellant tanke.	Simple modification	It is unlikely that acrodynamic Urag at 250 n.mi, is sufficient to collect propellants.
2-B Establish a procedure whereby vehicle propellants are transferred to a small, well-insulated receiver tank while under RCS thrust.	Flumbing mode to support receiver tank and to drain liquid from propellant tanks.	Propellant drain can be conducted in timely manner.	Additional RCS propellants required. Procedures and operations may be complex and time consuming.
2-C Introduce a screen channel capillary device that can scavinge liquid residuals off the tank walls during an extended time period.	Design a screen channel device for each propellant tank.	Simple propellant dratn procedure.	Major modification to projedhan tank designs. Propellant drain may he time consuming.
 2-L No problem exists if Solution 1-C for Requirement No. 1 is selected. 			

Recommended solution.

•

Table 4-3. On-Orbit Refueling of a COTV (Helium Pressurization System) (Contd.)

ļ

ADV ANT AGES DISADV ANT AGES		lant vapor or liquid will Complex and costly spare dryed cquipment required. Process take weeks to complete.	lant vapor or liquid Same as above but less re-liquefac it.				ch guarantees that Substantial propellant lost for the not penetrate start overboard dump process. No disadvantage for the recovery proce	Cannot guarantee that start basket will remain filled with lineid distant
VEHICLE HARDWARE REQUREMENTS		Plumbing mode for closed cycle No prope he iout.	Plumbing mods for open cycle purge. No propel will be to				The approvent the temperature of temperature	
	Problem No. 3: How can liquid and vapor the recovered during a purge or vent process?	volutions 3-A The closed cycle purge process will reclaim liquid, proyeellant vapor and helium while maintaining a constant propellant tank pressure.	3-B The open cycle vent process described by Figure will reclaim liquid, propellant vapor and helium during the tank pressure decay.	• 3-C No problem exists if solution 1-C for Requirement No. 1 is selected.	<u>Requirement No. 2</u> : Guarantee that helium will not penetrate and remain trapped within the start baskets.	Freddem No. 1: How can helium entry to the start baskets be prevented?	Sulutions: • 1-A If the propellant tanks are vented down for purposes of dumping fluid overboard or recovering fluids, it will be possible to ultimately evacuate the tank and therety eliminate all helium.	1-E Start baskets will remain filled with liquid during the closed cycle purge process as long as communications is maintained between tank

Recommended solution.

4-15

ł

be the same for either vehicle system. Vapor residuals will be about the same because they are only minimally affected by pressurization system. Liquid residuals will be the same because they will be controlled by the same type of propellant utilization system. The cost data of Figure 4-4 is applicable.

4.2.2.2 Propellant reclamation — The possibility of excessive propellant losses was the impetus for considering reliquefiers to reclaim all residual propellants following a COTV mission. A schematic of this method is shown in Figure 4-9 for hydrogen; procedures and assumptions are given below.

- 1. A hydrogen-helium vapor mixture will be extracted from the propellant tank. Little or no liquid is vented because it remains in contact with the tank surfaces and start basket.
- 2. Tank pressure will decay to approximately 6.9 kN/m² (1.0 psia) during the vapor expulsion process. Much of the liquid will boil during this period.
- 3. A compressor in the vent line will increase vapor pressure as it flows to the reliquefier. This compressor must be capable of increasing pressures to greater than 103 kN/m^2 (15 psia), the storage cank pressure.

Note: Compressor costs are not included in this evaluation.

- 4. The hydrogen-helium mixture enters the reliquefier where propellant vapor is recondensed. Reliquefier electrical power is provided by an existing solar array, and heat rejection is provided by space radiators.
 - Note: Only the cost of additional reliquefiers and radiators will be considered in this evaluation. These costs are given in Table 4-2.
- 5. Liquid hydrogen is separated from helium using a mechanical separator, with the helium being returned to its storage tank, and LH_2 returned to its storage tank.

The cost of reliquefying residual vapor containing helium will be about the same as the costs for no helium. Consequently, the same cost curve (Figure 4-7) is applicable. The same conclusion is also drawn, which is, the cost of reliquefaction is sufficiently high that propellant residual reclamation is not an obvious choice.

4.2.2.3 Residual for RCS propellants – Employing propellant residuals for RCS propellants is the same task whether or not helium is involved. The discussion of Section 4.2.1.3 is, therefore, applicable. The single exception to Section 4.2.1.3 is that RCS engine performance will be degraded by the presence of the inert gas. However, degradation will be small because helium represents a fraction of the total propellant residuals.



Figure 4-9. Schematic of a Propellant Residual Reclamation Process

4.3 COTV ON-ORBIT RESUPPLY

The technical obstacles associated with COTV refill operations from an orbital depot are similar to those identified for POTV refill from an orbiter-tanker. That is, the hostile space environment (zero-g and vacuum) combined with the limited resources available for space-based operations will complicate refill operations. A major difference between refill from an orbital depot or from an orbiter tanker is that resources available to the former should be substantially greater in terms of personnel and facilities. The additional resources should make it easier to perform COTV refill.

The thermodynamic and fluid mechanic considerations of COTV on-orbit refill are identical to those for the POTV. Consequently the same elements of an acceptable propellant transfer procedure apply: initial vent, prechill and fill. The POTV initial vent is equivalent to the COTV de-tanking operations of Section 4.2 Also, prechill and tank fill criteria for COTV is identical to that for POTV because the physical mechanism will be the same. Consequently, this discussion will focus on the propellant transfer flow conditions required to satisfactorily prechill and fill the propellant tanks. 4.3.1 COTV PRECHILL. The requirement for prechill of the propellant tank is to reduce tank temperatures sufficiently that the fill process will be accomplished without having to vent. The theoretical maximum tank pressures during prechill, as described by equations 3-7 and 3-8 are plotted in Figure 4-10 for the COTV and POTV. Note that peak pressures appear to be virtually independent of tank size. This is because the volume-to-mass ratios of each vehicle propellant tank is nearly the same. It will be shown in Section 6, on scaling, that this ratio is an important prechill scaling parameter. As was determined in Section 3.3.2, the LH₂ tank will be prechilled to a temperature not exceeding 200K (360R). This step will guarantee that the absolute maximum tank pressure will not exceed 138 kN/m² (20 psia). The liquid oxygen tank will not require a prechill process (as was previously determined for POTV).

4.3.1.1 Prechill procedures — The following charge and vent procedure was selected for LH_2 tank prechill:

- 1. Meter LH_2 into the tank at a high velocity to provide good heat exchange with the walls.
- 2. Allow time for tank pressure to peak out at about 69 kN/m² (10 psia). Vapor temperature equals tank temperature at this time.
- 3. Vent the tank to near zero pressure and repeat steps 1 and 2 as required to reduce tank temperature below 200 K (360 R).



Figure 4-10. COTV and POTV Peak Prechill Pressures Will be the Same

OR.CI.VAL PAGE IS

Figures 4-11 and 4-12 give tank ullage pressure and wall temperature histories during this prechill period. These curves are based upon introducing 42.6 kg (94 lb) hydrogen at 0.91 kg/sec (2 lb/sec) during the charge period. Peak pressures for this procedure will not exceed 79.3 kN/m² (11.5 psia). Flow conditions selected for this simulation are given in Table 4-4 and appear to be readily attainable conditions. It is concluded that LH₂ tank prechill will be a rather straightforward process, and of a sufficiently short duration that it can readily be integrated into a vehicle mission prelaunch sequence.

A comparison of the aforementioned ullage pressure and wall temperature histories with those for POTV (Figures 3-19 and 3-20) reveals an obvious similarity between both sets of conditions. The sin. 'larity is even more striking if the data is plotted with respect to a normalized time (Figure 4-13). Time is normalized by dividing the actual time by the total time of each charge duration. The excellent data correlation supports the thesis that prechill data on one vehicle configuration and size can be extrapolated to other sizes. This subject will be discussed in Section 6.







Figure 4-12. COTV Liquid Hydrogen Tank Temperature History During Prechill



Figure 4-13. Excellent Correlation Exists Between COTV and POTV LH₂ Tank Prechili Predictions

Table 4-4. Conditions Selected for COTV Liquid Hydrogen Tank Prechill Procedure

I		
	1.	Initial propellant tank temperature = 289°K (520R)
	2.	Initial pressure = 6.9 kN/m^2 (1.0 psia)
	3.	Hydrogen vapor saturated at 103.4 kN/m ² (15 psia) enter propellant tank
	4.	Entering flowrate = .91 kg/sec (2 lb/sec)
	5,	Entering velocity = 6.7 m/sec (22 ft/sec)
	6.	Hydrogen charge terminated after 42.6 kg (94 lb) enters tank
	7.	Tank vent initiated when tank-to ullage temperature difference becomes 5.6 K (10R)
	8.	Propellant tank vented to 6.9 kN/m^2 (1.0 psia)
	9.	Vent area = 148.6 cm ² (23 in. ²)
	10.	Tank mass = 2228 (4911 lb)
	11.	Tank volume = 548 (19363 ft^3)

4.3.2 COTV TANK FILL. Tank fill will be initiated after prechill is completed. The single requirement for tank fill is to maintain acceptably low pressures during the process. The ideal condition will be to maintain thermal equilibrium during fill, which can be approached as heat and mass exchange between the phases is increased. It was proposed in Section 3.3.3 that near-thermal equilibrium conditions can be attained by introducing propellant to the tanks through spray nozzles. A high energy-exchange rate will be provided during the early part of the fill process (to about a 40 percent fill level) as a result of liquid spray interaction with the ullage. An even greater energy exchange rate will occur during the latter stage of tank fill (about 60 percent to 100 percent liquid fill) due to the interaction of vapor entrained in the liquid bulk.

Figures 3-24 and 3-25 give the relationship between entering hydrogen and oxygen liquid vapor pressure and final tank pressure for thermal equilibrium. This data is applicable both to POTV and COTV propellant tanks. These curves show that tank fill pressures will be maintained within acceptable levels if near-equilibrium conditions are achieved. The HYPRES computer program (which was used for POTV analysis) was employed to determine propellant tank pressure histories for a representative COTV fill condition. Results are plotted in Figure 4-14 for the set of flow conditions identified in Table 4-4. The only difference between the two computer runs is in the entering liquid temperature. The higher receiver tank pressure history is based upon a fixed inlet temperature condition (an idealized assumption). The lower tank pressure



Figure 4-14 Supply Tank Temperature Influence Upon COTV Pressure During Refill.

history reflects the decreasing temperature conditions that occur within the supply tank during propellant expulsion. Figures 4-15 and 4-16 compare inlet hydrogen temperature and pressure histories for the two tank fill conditions.

SUMMARY

Orbital refill operations for a COTV will be very similar to POTV refill. Prechill procedures will be virtually identical, including the requirement to prechill the liquid hydrogen tank below 200K (360R) before tank fill is initiated. The same number of charge and vent cycles can be employed, and the resulting maximum pressures will be the same. Tank fill procedures will also be similar between the two vehicle stages. Propellants will be sprayed into the tanks to enhance mixing between the fluid phases needed to approach near-thermal equilibrium fill. The primary difference between POTV and COTV fill will not be procedural, but thermodynamic. The different receiver-tank-supply tank volume relationships will result in different propellant inlet temperature conditions during fill. The subsequent receiver tank pressure histories



Figure 4-15. Predicted Supply Tank Liquid Temperature During COTV Refill



Figure 4-16 Predicted Supply Tank Liquid Vapor Pressure During COTV Refill

will also differ (compare Figure 3-37 to 4-14). This difference, however, is not sufficient to alter the conclusion that POTV and COTV orbital refill operations can be considered identical. Any experimental program devised for one vehicle configuration will also be applicable to the other.

The only new requirements identified by the COTV analysis dealt with alternatives of reclaiming, or employing, the vehicle propellant residuals. It was concluded that reclaiming residuals through a reliquefaction process was marginal at best because of high reliquefier costs. A suprising result was that dumping propellants overboard could be preferrable to reliquefaction. The best alternative, however, could be to use the residuals to supplement the orbital depot RCS system propellant requirements.

5

LTL ORBITAL RESUPPLY

A mission scenario will be developed in this section for the low thrust liquid (LTL) earth storable propellant vehicle concept. A mission will be defined which encompasses the important issues of orbital-refueling operations using earth storable propellants. These operations will include the major activities from post-mission storage in the LEO parking orbit through resupply. Particular emphasis will be placed upon vehicle "safing" operations that can be performed prior to orbiter-tanker rendezvous. Vehicle and orbiter-tanker subsystem requirements needed for orbital refueling will be identified. Operational procedures and techniques for orbital propellant transfer will then be developed.

A groundrule imposed upon this vehicle concept was to utilize hardware from existing/ on-going programs, if possible. Because the data base for earth storable vehicles and missions was considerably smaller than for cryogenic OTVs, no attempt was made to optimize the vehicle configuration. Rather, the intent was to select a configuration that would be representative of its vehicle class.

5.1 MISSION SCENARIO

In the 1980's-1990's time frame a requirement may exist for a versatile, low cost vehicle capable of accomplishing small orbital transfers, service, inspection, and retrieval tasks. It will require frequent refill from a dedicated orbiter-tanker. The orbiter-tanker will be equipped to resupply the main propellant tanks and the RCS supply tanks with N_2O_4 and MMH, and the pressurization system bottles with helium.

5.1.1 SELECTED LTL MISSION. As previously stated, there is not the data base available describing potential earth storable vehicle missions that exists for cryogenic POTV's. Fortunately, it appears that a wide range of missions could be selected without impacting orbital refill procedures. For this study, the LTL primary task will be to transfer large space structures (65317 kg (144,000 lb) from LEO ($\sim 200 \text{ n. mi.}$) to a slightly higher orbit ($\sim 600 \text{ n. mi.}$) under low accelerations. A vehicle thrust level of approximately 26.7 kN (6000 lbs) will be required.

5.1.2 LTL CONFIGURATION. In keeping with the intent of a low cost vehicle design, the LTL configuration is comprised of existing hardware. Figure 5-1 provides a description of the selected systems, all of which are currently available.

5.1.2.1 <u>Main propulsion tankage</u> — This propulsion tankage system consists of four identical propellant tanks, each containing a screen acquisition device (four galleries)





.
and a bulkhead screen, Figure 5-2. These propellant tanks were designed for the Shuttle orbital maneuvering system (OMS) and use N_2O_4 and MMH as propellants. The tanks are designed to operate at approximately 1760 kN/m^2 (255 psia), which is required to satisfy engine inlet conditions. Helium will maintain the operating tank pressures since propellant vapor pressures are relatively low; about 131 kN/m^2 (19 psia) and 6.9 kN/m² (1.0 psia) respectively for N_2O_4 and MMH.

The engine system will have a thrust level of 26.7 kN (6000 lbs), which was selected for the mission.

5.1.2.2 <u>Reaction control system</u> — Very little data was obtained on the Reaction Control System (RCS), other than the system (thrusters, supply tank, plumbing) was developed for orbiter and includes a screen acquisition device in each tank. This system is designed for the same operating pressure and uses the same propellants as does the main propulsion system.

It was assumed for this study that problems associated with orbital refill of the OMS and RCS propulsion systems would be identical. Analysis effort was concentrated on the main propulsion system, and the results would also be applicable to the RCS system.

5.1.3 LTL VEHICLE CONCEPTS. Two vehicle concepts were evaluated to determine which would most closely satisfy the guidelines established for a low cost vehicle. Each concept employs existing hardware.

5.1.3.1 <u>Vehicle concept one</u> – Each of the two Shuttle pod assemblies contains two cylindrical 2.55 m³(90 ft³) propellant tanks for the OMS engines and two spherical tanks for the reaction control systems. Figure 5-3 shows a MMH/N_2O_4 LTL vehicle concept which uses these shuttle tanks. A cylindrical body structure (equipped with a thrust cone), a main engine, four OMS tanks, four RCS tanks, two modular helium bottles, four RCS clusters, an avionics section, and a docking system are the basic vehicle components. The OMS and RCS tanks are supported from the body structure with a system of struts and yoke fittings.

The body structure is a cylinder equipped with kick rings for reacting the loads from the main engine, the tanks, and the payload. The structure also includes provisions at the forward end for mounting astrionics packages. The type of construction is not indicated; however, several are readily adaptable such as the open truss, composite cylinder, skin stringer frame, semi monocque, and the open or closed isogrid.

The main engine is a 26.7 kN (6000 lb) thrust unit with an overall length of 196 cm (77 in.) coupled with an exit diameter of 117 cm (46 in.). The Isp = 310 sec and the expansion ratio = 55. The engine is gimbal mounted for a 7° excursion in any direction using two actuators located 90° apart.





•





,

5-5

Some minor additions to the OMS tanks will be required such as an aft conical skirt equipped with a disconnect for receiving the helium bottle modules; a forward cylindrical skirt for mounting the ACS clusters and several fittings for structural and plumbing supports. Alterations for plumbing interfaces may also be required to accommodate the vent and fill circuits. The aft conical skirt section previously mentioned will be required for two tanks only; one for each helium bottle module required.

The four RCS tanks are spheres equipped with acquisition devices. Two of the spheres are located inside the body structure and two are located externally near the main engine thrust cone. A strut support system is indicated; however, other methods may be used such as girth flanges or skirts.

The helium supply is two 102 cm (49 in.) spherical modules equipped with a latch system and a motor driven disconnect. Details of this system will be discussed in Section 5.4.2. Additional conceptual information is shown in detail "A" of Figure 5-3.

A typical reaction control system (RCS) cluster consists of four thrusters mounted inside the OMS tank skirt structure ∞ pointing outboard (see layout). Four clusters are used and are 98 N (22 lb) for the unit with an Isp = 280 sec and a mixture ratio of 1.65. The approximate overall length of each thruster is 25.4 cm (10 in.) Flange type mounting is provided.

The astrionics section is located at the forward end of the body structure. The four OMS tanks shadow this section therefore heat dissipating systems may be required such as heat pipes and radiators. Possible locations for the radiators would be between the OMS tanks.

The docking and payload attachment system is located forward of the astrionics and interfaces with the shuttle or the payload. Features such as gross capture under wide misalignments, shock absorbing, pull down and final alignment, followed by structural attachment would be included in this system.

5.1.3.2 <u>Vehicle Concept 2</u> — In Concept 1, two types of propellant storage tanks are shown. For the main propulsion system, four OMS tanks are used and for the RCS system, four spherical tanks are used. An alternate approach is shown in Figure 5-4 which uses six OMS tanks. Two of these tanks serve the RCS system and the remaining four supply the main propulsion. Similar to Concept 1, the tanks are attached to a cylindrical body structure equipped with a thrust cone. In this case the thrust cone is reversed to minimize the vehicle length.

Two of the RCS clusters are mounted on the forward ends of two OMS tanks and the remaining two clusters are supported from an open truss structure located between the OMS tanks. The purpose for the truss structure is to permit a 90° spacing between clusters. The 'russ is shown attached to the main body section, however,





•

other methods could be used such as a bridge structure between two OMS tanks or a cantilever support fitting attached to one OMS tank.

Similar to Concept 1, modular type helium bottles are used. For this case, three bottles are required.

The primary advantage of concept 2 over concept 1 is that it could be simpler to construct. There is a single type of propellant storage tank (the OMS tank) rather than both the OMS tank and RCS tank. Furthermore, structural attachments would be simpler for the six common tanks. It is possible that concept 2 would be less costly than concept 1. The primary disadvantage of concept 2 is that it is a heavier vehicle. The two OMS tanks weigh more than the four RCS spheres they replace. Also, the increased volume of the OMS tanks will require an additional helium bottle for pressurization.

Problems and/or solutions to on-orbit resupply should be similar between concepts 1 and 2. These are refill procedural differences that may favor concept 1, which is why this concept was selected for evaluating orbital refill. These differences will be discussed in Section 5.3.2.

5.1.3.3 Fluid systems for concept 1 – Figure 5-5 shows the basic plumbing for the propellant fill, vent and cross-over circuits. A system schematic is given in Figure 5-6. Plumbing for the main propulsion, RCS, and pneumatic systems are shown. Referring to the schematic, the overall system for each propellant consists of a fill circuit and a vent circuit. On the fill side, each pair of OMS tanks and also each pair of RCS tanks are tied together and fed with a common line which starts at the d'sconnect. The OMS tanks only are equipped with internal fill tubes incorporating spray nozzles. All tanks are equipped with shut-off valves for controlling, filling and transferring.

For the vent circuits, each pair of tanks are tied together and routed through a non-propulsive overboard vent. Each tank is equipped with a vent valve.

Tube routings can vary considerably depending upon vehicle configuration, supports, etc., therefore the purpose of Figure 5-6 is to show only a general approach as to how a system may be laid out. Referring to the N_2O_4 side, the fill circuit starts at the disconnect which is supported from the aft conical skirt on one of the OMS tanks. The fill line is routed to the nearest OMS tank and across the vehicle to the opposite OMS tank. The loop shown in the cone line is due to support attachments on the thrust cone and on the conical skirts. The same arrangement is used for the MMH. For the N_2O_4 tanks located inside the body structure, the fill tubes are routed aft to the main cross over line. For the MMH side, these tubes are routed forward to the cross-over line.





,

5-9





.



- Valves & B. Brungh & Lunahik Hupud Hum Frau Prog. Huma (and a capitary at reculary spectra to conjust a bruncia si, or furths adjacont programma (and).
- 2. Vahrea ficialityang ki ara caujaliyada darrang propa bland Hili operations.
- Valeta Bathrough 12 are culphojed turing propelland facts venting operations.

A non-propulsive vent system for each propellant is located at the forward end. The exits for each of these circuits are positioned outboard to prevent impingement on the OMS tanks, payload and adjacent structures. Four open truss arms attached to the body structure provide support for the tubing. Basically the non-propulsive vent is a loop of tubing with the ends of 180° apart. This arrangement is frequently referred to as the "steer horn". The vent lines from each tank are connected to this loop of tubing such that there is equal distance from each point of connection to the overboard ends. A typical tube assembly would feature welded joints and the use of bends instead of fittings where possible. Large sections would be bench fabricated to ensure maximum quality control.

The fill and vent circuits were designed for the purpose of enhancing on-orbit propellant refill. Refill must be conducted with caution because of the corrosive nature of the earth storable propellants. Refill can be difficult because the OMS and RCS propellant tanks must first be vented from about 1760 kN/m² (255 psia) to 207 kN/m² (30 psia) in order to expel sufficient helium to enable refill.

5.2 ORBITAL PROPELLANT RESUPPLY TECHNIQUES

The obstacles of on-orbit refill are the same for the LTL vehicle as for a cryogenic POTV. These obstacles are: 1) the hostile space environment and (2) the limited resources available for space-based operations. The primary variable complicating refill is the zero-g environment.

Propellant tank fill with N_2O_4 or MMH in an orbital environment poses fewer problems than liquid hydrogen because of temperature environment and operating pressure level differences between a earth storable and cryogen system. System temperatures will not vary significantly for these propellants at any time during a fill or refill operatio. Consequently, prechill will not be required to precede tank fill.

Propellant tank operating pressure of 1760 kN/m^2 (255 psia) is considerably higher than for liquid hydrogen. This higher pressure level will simplify tank fill because fill pressures will be maintained below the normal operating levels without difficulty.

One problem that cryogenics and earth storables have in common is that an undefined (or poorly defined) liquid-vapor distribution will exist during low-gravity tank fill. This inability to determine vapor location will make it difficult both to assure that no liquid is vented during tank fill, and to assure that vapor is not trapped within the screen channel device.

For the analyses that follow attention will focus upon the problems associated with N_2O_4 and with main propellant tank refill. N_2O_4 is selected because it represents a more difficult propellant to handle than MMH due to its higher vapor pressure. The main propellant tanks were used in these analyses.

5.2.1 PROPELLANT TANK REFILL REQUIREMENTS. Several requirements were identified as being necessary to assure that LTL vehicle refill will be safely and effectively performed. These are general requirements that should apply to a variety of vehicle configurations. There will undoubtedly be configuration-related requirements for any vehicle. However, such requirements cannot be identified without first having considerably more vehicle design detail than was available for this study.

5.2.1.1 Vent propellant tanks prior to orbiter rendezvous – $N_2 Q_4$ is a particularly corrosive fluid, in vapor or liquid form. Fluid impingement upon the LTL or orbiter surfaces could have a long-term adverse affect upon vehicle components. The orbiter will be spared this potential hazard if LTL vent procedures are performed well in advance of orbiter rendezvous.

5.2.1.2 Minimize liquid vent potential — This requirement addresses two major concorns; propellant corrosiveness and vehicle control. Liquid venting must be minimized because it is potentially more damaging than vapor venting. First, a liquid vent plume cannot be readily confirmed, i.e., liquid exposed to a vacuum will boil and expand in all directions. Also, the impingement mass flux from a liquid vent is likely to be greater than from a vapor vent plume, which increases the corrosive potential.

The second aspect to the potential problems of liquid venting is vehicle control. Because procedures will be identified for remote venting of the LTL propellant tanks, it is mandatory that vehicle control be maintained. Liquid venting could jeopardize vehicle control because it is unlikely that a net zero thrust would result, even if venting through a non-propulsive (designed for vapor) vent system.

5.2.1.3 <u>Helium must not enter screen galleries</u> — The purpose of the screen acquisition device in each propellant tank is to provide the **capability** for pure liquid flow. Any helium entering a screen device will be trapped and remain trapped until removed through special procedures, or until it flows from the tank to the engine system. Special procedures for removing trapped helium are undesirable because they are time consuming and may be complicated. Helium expulsion with propellant is unacceptable because it violates the requirement of 100 percent liquid availability to the engine system.

5.2.2 INITIAL FILL. It is unlikely that the vehicle will undergo an initial propellant fill in space because of the ease with which it can be filled on the ground prior to flight. There is a possibility, however, that the propellant tanks could be completely evacuated at a future time for maintenance. Following such an occurrence propellant tank fill would be performed on evacuated tanks residing at ambient temperature.

5. 2. 2.1 <u>Non-equilibrium fill</u> — The high operating pressure levels will simplify OMS propellant tank fill because fill pressures will be maintained below the normal operating levels without difficulty. This is illustrated in Figure 5-7 which shows maximum tank pressure as a function of percent liquid fill. Isentropic compression of the ullage is assumed, which means that heat exchange with the liquid and tank walls is

zero. Note that 90 percent fill can be effected without exceeding 1378 kN/m^2 (200 psia) pressure. Unlike cryogenics this extremely conservative approach can be used because results are acceptably low.

5.2.2.2 <u>Thermal equilibrium tank fill</u> – Figure 5–7 provides evidence that initial tank fill will be readily achieved. It is more reasonable, however, to expect near-thermal equilibrium conditions to exist during fill. This process is described below.

Thermal equilibrium fill represents the minimum propellant tank pressure condition that can exist during fill. The thermal equilibrium tank fill relationship derived for liquid hydrogen (Equation 3-23) applies as well to N_2O_4 . This equation is given below.



$$h_{L} = u_{g_{2}} m_{g_{2}} / m_{L_{2}} + u_{L_{2}} + \Delta u_{W} m_{W} / m_{L_{2}}$$
(5-1)

Figure 5-7. Maximum OMS Propellant Tank Pressure Ducing Initial Fill With N2O4

5-13

where

h L	= enthalpy of entering liquid final vapor
$\begin{array}{cc} {\tt u} {\tt m} \\ {\tt g}_2 {\tt g}_2 \end{array}$	= total internal energy
^m L ₂	= final liquid mass
∆u m w w	= storage tank wall energy change during fill
^u L _o	= final liquid internal energy

It is possible to relate h_L , u_{L_2} and u_{g_2} to liquid vapor pressure and temperature under thermal equilibrium conditions. Thus u_{L_2} and u_{g_2} are known once final vapor pressure is specified. Δu_w can also be determined, when initial and final temperatures are given. Finally, h_L (which is a function of entering liquid vapor pressure) can be determined for a desired liquid fill condition.

Equation 5-1 is summarized for N_2O_4 in Figure 5-8 which gives entering liquid vapor pressure as a function of initial tank temperature and final tanked liquid vapor pressure. It is interesting to note that the effect of initial wall temperature upon final tank pressure is negligible. This data illustrates that the N_2O_4 thermal mass will so dominate tank fill that the tendency will be for final tank pressures to approach that of the incoming liquid vapor pressure.

The same fill technique, that of liquid spray into the tank, will be employed for N_2O_4 as for liquid hydrogen. This approach will assure that N_2O_4 will be at least as close to thermal equilibrium as liquid hydrogen, at the same fluid power inflow conditions. This assessment is made on the basis that the liquid-to-vapor thermal mass ratio is six times greater for N_2O_4 than for hydrogen. Thermal equilibrium is more readily achieved for fluids having high thermal mass ratios.

5.2.3 ON-ORBIT REFILL. Most, if not all, problems associated with LTL vehicle resupply will be associated with the need to vent helium before propellant refill can be initiated. Helium venting must be conducted with care under orbital conditions because liquid may also be vented. In addition, helium may enter the screen channel device (or galleries) during this period, unless precautions are taken.

Helium venting is necessary prior to attempting refill in order to avoid excessively high tank pressures during the refill process. To illustrate this point, helium partial pressure in a N₂O₄ tank will increase from 1758 kN/m² (255 psia) to 15820 kN/m² (2295 psia) as the tank is filled from 10 percent liquid to 90 percent liquid.

The refill process will not represent a concern once sufficient helium has been vented to avoid high partial pressures at small ullage volume conditions. Thus the key to a

CONDITIONS

- 1. TANK VOLUME = 2.55 m³ (90 ft³)
- 2. TANK MASS = 253 kg (558 lb_m), TITANIUM
- 3. TANK PRESSURE IS 0.0 kN/m² (0.0 psia) PRIOR TO FILL
- 4. NO VENTING DURING FULL





successful refill is helium venting which precedes this process.

5.2.3.1 <u>Thermodynamics of propellant tank vent</u> — Tank venting must guarantee that the screened volume will not be contaminated by helium entry. Helium cannot penetrate the device while it remains filled with propellant. However, once vapor resides within the device, helium entry can occur as a result of molecular diffusion or forced convection flow. Thus a vent procedure must be selected that will maintain the screen device filled with liquid.

There are two phenomenon of concern during tank venting: liquid boiling and surface evaporation. Liquid boiling within the galleries must be avoided. Surface evaporation cannot be avoided, but propellant must be available for liquid make-up. The following steps describe a vent process selected to satisfy the requirement of keeping the screen galleries liquid-filled (N_2O_4 properties are assumed because a procedure acceptable to N₂O₄ is also acceptable to MMH):

1. The partial pressure of helium and N_2O_4 vapor will decrease as tank pressure is decreased. N_2O_4 evaporation will begin at the screen surface once liquid

vapor pressure exceeds the partial pressure of the vapor.

- 2. Liquid will be drawn into the capillary device to replace liquid lost due to surface evaporation at the screen (Figure 5-9). Thus, sufficient liquid volume surrounding the galleries, and in communication with the contained liquid, must be available for liquid replacement. This quantity must be determined.
- 3. The liquid bulk cannot boil while pressure exceeds liquid vapor pressure. Consequently, tank vent will be terminated at a pressure greater than 131 kN/m^2 (19 psia) to assure that the capillary device remains filled.

Note: The only evaporation (or boiling) that can occur is at a liquid-vapor interface, and only if vapor pressure at the surface exceeds partial pressure above the surface.

It is estimated that properlant tank pressure must be reduced to about 210 kN/m^2 (30 psia) prior to starting propellant refill. Approximately 12.7 kg (28 lb) of N₂O₄ vapor will be vented in the process. Less than 2 kg (4.4 lb) of MMH vapor will be vented.

EVAPORATION WILL OCCUR DURING TANK VENT

SCREEN DEVICE TANK SKIN VAPOR BUBBLE

Vapor may form within galleries if propellant is not available for liquid make-up.

EVAPORATION LIQUID MAKE-UP LIQUID RESIDUAL

Figure 5-9. Propellant Tank Venting Can Occur Without Losing Liquid From Screen Galleries

5.2.3.2 Propellant tank refill - Propellant tank refill can be performed in a manner similar to that of initial tank fill. As with initial fill, the high operating pressure levels will tend to simplify tank refill. Unlike the initial fill condition, helium in the ullage precludes using the most conservative approach of isentropic compression to verify that refill can be achieved. There is a less conservative method, however, to illustrate that tank refill can be readily accomplished. Figure 5-10 indicates that refill to the 87 percent level will be possible even for isentropic compression. The goal, however, is to achieve a refill of 95 percent, which can readily be attained if nearthermal equilibrium conditions are maintained during refill. According to the discussion of Section 5.2.2.2, the liquid-to-vapor thermal mass of $\mathrm{N_2O_4}$ and helium is such that near thermal equilibrium conditions will be maintained. Figure 5-10 indicates that refill can be achieved even if the ullage is superheated by 55.5 K (100R) above the liquid. Such a temperature differential cannot possibly be sustained within the OMS tanks as propellant enters through spray nozzles. The RCS tanks do not include spray nozzles but, even so, a 55.5°K (100°R) temperature differential will be virtually impossible to sustain. It is concluded that refill will be a straightforward operation.

5.3 HELIUM VENTING

There are two categories of potential problems associated with venting the LTL vehicle OMS tanks; one is liquid venting, and the other is helium entry to the capillary device.



Figure 5-10. OMS Propellant Tank Pressures During Refill

5 - 17

The loss of liquid overboard during the venting process is not only an inefficient operation, it also represents a corrosion hazard if N_2O_4 is vented. Because of its corrosive properties even N_2O_4 vapor venting must be carefully performed to avoid vapor impingement upon LTL vehicle and orbiter surfaces. Liquid venting will represent a more serious concern because the vapor cloud formed when liquid is exposed to the vacuum environment will expand in all directions and be difficult to control.

Regarding the second category of problems, tank venting must guarantee that the screened volume will not be contaminated by helium entry. Helium cannot penetrate the device while it remains filled with propellant. However, once vapor resides within the device, helium entry can occur as a result of molecular diffusion or focused convection flow. Thus a vent procedure must be selected that will prevent the loss of liquid from the screen device.

5.3.1 ALTERNATIVE VENT PROCEDURES. Various alternatives were considered for the tank venting procedure. An overview of each alternative, is presented below.

- 1. Venting while the propellants are settled by the OMS main engines or the RCS. This concept will eliminate the possibility of venting liquid overboard. A major disadvantage is that the tank cannot be completely vented down. This is because both the main engines and the RCS require a substantial operating pressure.
- 2. Venting while docked to the orbiter with the propellants settled by the orbiter RCS. This concept also eliminates the possibility of venting liquid overboard. It will also be possible to completely vent the OMS tank down since the orbiter RCS is used to settle propellants. A possible disadvantage is that firing of the orbiter RCS engines after docking has been completed may be an operational complexity. The major disadvantage is that this approach violates the requirement for venting prior to orbiter rendezvous.
- 3. Venting while docked to the orbiter with the propellant settled by atmospheric drag. This concept eliminates the possibility of venting liquid overboard. One disadvantage is that an orbital altitude <130 n. mi. is required for aero-dyr amic drag to overcome propellant surface tension forces, which is substantially below the desired altitude of 200-250 n. mi. Another disadvantage would be the necessity of attaching a vent duct extension to keep the corrosive vapor of the earth storables from contacting the vehicle surface. The requirement for a vent duct extension is based on calculations, plotted in Figure 5-11 of the maximum Prandtl-Meyer expansion angle of an earth storable-helium mixture. The same major disadvantage exists as for Item 2, above.</p>
- 4. Venting with the propellant unsettled prior to docking with the orbiter. This concept-approach could result in the venting of liquid overboard, which violates a major requirement. It would also require the use of a vent tube to prevent vehicle contact with the vented propellant.



Figure 5-11. Expansion Angle of Ideal N2O4 - Helium and MMH-Helium Mixtures as a Function of Helium Partial Pressure

5. Venting through a helium recovery system prior to docking with the orbiter. The objectives of a helium recovery system are a) to eliminate the loss of helium pressurant as a result of venting overboard, and b) to eliminate the hazards of venting corrosive fluids. The recovery system, illustrated in Figure 5-12, is a closed system. Referring to Figure 5-11, a description of the system components is as follows: The molecular sieve is used to absorb any N_2O_4 or MMH vapor contained in the vapor mixture. The multi-stage compressor pumps helium from the OMS tank pressure up to a storage bottle maximum pressure of 33,100 κ N/m² (4800 psia). However, the helium must be passed through a radiator between each of the compressor stages to reduce vapor temperature to the allowable temperature range of the helium storage supply tank.



Figure 5-12. Helium Recovery System

This concept requires that the helium recovery system be contained as a package within the LTL vehicle.

6. Venting with the helium recovery system following docking with the orbiter. The helium recovery system used would be identical to that described in method (5). However, the system now would be a unit contained on the orbiter as opposed to a unit contained on the LTL vehicle. Consequently, there would be no need to provide special attention to the attitude control system.

The helium recovery system is a desirable solution to the venting problem since it provides for the continued reuse of the helium pressurant and eliminates any vent hazard. It does represent a major and, perhaps, costly vehicle modification because of the development effort required to integrate a compressor, space radiators and molecular sieves. Furthermore this hardware addition represents a reduction in payload capability and an increased vehicle complexity. 5.3.2 SELECTED VENT PROCEDURE. The preferred procedure for venting helium while minimizing or eliminating the problems of liquid venting is given in Table 5-1. The advantages of this approach are that it is simple and requires no development. Basically the approach is to utilize the two sets of main propellant tanks and RCS propellant tanks for transferring propellants from one tank to another. Table 5-1 describes how any given propellant tank can be drained by transferring propellant to an adjacent tank. Once drained, the tank can be vented with a minimum concern for liquid venting. It is visualized that the procedure can be initiated on a signal from the ground or from the orbiter. Also, portions of the procedure will be automated so that valves can be commanded open and closed on the basis of continuous monitoring of propellant tank pressures and propellant mass gaging output signals. Additional assurance for liquid free venting can be provided, if necessary, by commanding the RCS settling thrusters to fire prior to and during all venting operations.

The Table 5-1 procedure is applicable to N_2O_4 and MMH as well as to the main and RCS propellant tanks. At the completion of this venting operation one set of propellant tanks will be empty (except for a minimum liquid residual volume) and the other set of tanks will contain the bulk of propellant residuals. Tank pressures will be approximately 207 kN/m² (30 psia) in all tanks except for a set of RCS propellant tanks. These tanks cannot be vented until after docking with the orbiter is complete because LTL attitude control capability must be maintained until that time, and this capability will be lost once the RCS tanks have been vented. The two tanks will be vented in exactly the same manner as described in Table 5-1.

The following LTL vehicle conditions will exist prior to initiating refueling operations:

- 1. One each of the OMS and RCS propellant tanks will contain minimum liquid residuals. The other set of tanks will contain propellants in excess of the minimum liquid residuals.
- 2. Propellant lines are primed with liquid.
- 3. All screen devices are primed with liquid.
- 4. All tank pressures are about 207 kN/m² (30 psia).

The vent procedure of Table 5-1 is better suited to vehicle concept 1, which has a second set of RCS tanks, than vehicle concept 2. This second tank set provides the capability of first draining propellants from the RCS tanks to be vented while docked to the orbiter. This step greatly minimizes the possibility of losing liquid overboard during the RCS tank vent process. With concept 2, the RCS (OMS) tanks would be vented while containing all propellant residual.

Table 5-1. LTL vehicle tanks venting procedure prior to orbiter rendezvous.

- Close the shutoff valve between helium supply and OMS MMH tanks. This will 1. enable tank venting without helium resupply through the pressurization system pressure.
- 2. Monitor the zero-g mass gauging devices of tanks #1 and #2 to determine which contains fewer propellants. Note: The tank with less propellant will be drained first. (Assume for this discussion, that tank #2 has less propellant.)
- Close valves #3, #4. Valves #7 and #8 are also closed (Refer to Figure 5-6 3. schematic).
- Open valve #11. This enables venting through the non-propulsive vent system. 4.
- Close value #11 when pressure in tank #1 decays from the initial 1720 kN/m^2 5. (250 psia) to TBD kN/m^2 . This pressure will automatically be selected by computer which continually monitors mass gauging device output.

The procedures will not be performed simultaneously for both propellants because of the desire to avoid simultaneous venting of MMH and N_2O_4 vapors.

- Open valves #3 and #4. The pressure difference will enable propellant transfer 6. from tank #2 to tank #1. 7.
- A signal to close valves #3 and #4 will be sent when the mass gauging device indicates that minimum liquid residuals remain in tank #2. 8.
- Open value #12 and vent tank #2 from 1720 kN/m² (250 psia) to about 207 kN/m² (30 psia). The possibility of liquid loss during venting has been minimized because the bulk of propellants was previously transferred to tank #2. Helium penetiation of the screen device will not occur during venting because the device is filled with liquid. Furthermore bulk liquid boiling will not occur until tank pressure decays to 131 kN/m² (19 psia) for N₂O₄ and 6.9 kN/m² (1 psia) for MMH. Tank #2 is now ready to be refilled.
- Close valve #12 and open valves #3 and #4. The pressure difference will allow 9. propellant transfer from tank #1 to tank #2.
- A signal to close valves #3 and #4 will be sent when the mass gauging device 10. indicates that minimum liquid residuals remain in Tank #1.
- 11. Open valve #11 and vent tank #1.
- Close value #11 when tank pressure has decayed to about 207 kN/m² (30 psia). 12. Tank #1 is now ready to be refilled.

5.4 PROPELLANT REFILL PROCEDURE

The procedure of Table 5-2 has been selected as being applicable to the transfer of MMH and N_2O_4 into the OMS tanks and RCS tank from the orbiter supply tanks. These procedures include the steps needed to assure that propellant will not leak overboard during refill. Two items require particular attention; leakage through an improperly sealed disconnect valve, and residual propellant spilling from the transfer line after disengaging the transfer line disconnect for return to the orbiter cargo bay.

Any propellant spillage must be avoided, or minimized, because of its corrosive nature. The disconnect valve was of sufficient concern that design requirements were established, and a conceptual design was developed. Valve design and operation are further discussed in Section 5.4.1. Residual propellant spill can be readily avoided through a purge procedure which permits helium entry at the transfer valve disconnect and flow toward the supply tank. This low flowrate purge will force the liquid bulk back to the orbiter supply tank.

5.4.1 EARTH STORABLE PROPELLANT DISCONNECT VALVE. A preliminary design of an earth storable disconnect value is shown in Figure 5-13. The value is installed on the end of the Shuttle transfer line and makes both structural and fluid seal connections with the LTL prior to the transfer of propellants from the Orbiter-tanker. The Shuttle RMS attaches to the valve housing and deploys the valve and transfer line to the LTL mating interface. The transfer line is equipped with a power cable for operating and monitoring the valve.

As previously stated, Figure 5-13 is a preliminary design effort, and prior to a final selection, it would be required to generate several design options. For example, several actuating methods involving pneumatics, hydraulics, and electro mechanical devices would be included in the tradeoffs. Sealing is also a critical item and will require numerous investigations. The structural attachment system is another area which will require tradeoffs.

It was assumed for this design effort that the valve shall (1) be capable of attaching to the LTL within the positioning tolerance band of the Shuttle RMS, (2) incorporate final alignment provisions before fluid sealing, (3) feature flat sealing surfaces (no probes in holes), and (4) have zero spillage when disconnected. The valve shall also include systems for monitoring the seals before and after transfer. An electro-mechanical method of actuation was assumed.

Referring to Figure 5-13, the main sections of the valve are a flex duct assembly, two actuators, an outer housing, three latch systems, a mating passive assembly which is installed on the LTL, and seal monitoring systems. The overall dimensions shown are larger than initially expected and can be reduced by additional design

Table 5-2. LTL vehicle propellant tanks refill procedure (applicable to N_2O_4 and MMH).

- 1. Attach orbiter propellant transfer line to the LTL vehicle. This process includes the following steps as a minimum:
 - a. Structurally engage transfer line disconnect to vehicle disconnect.
 - **b.** Perform leak test of disconnect seal cavities to verify that the system is leak-free. This test is performed with GH_o.
 - c. Vent GH_e overboard. The transfer line is now engaged and fully evacuated, but the vehicle disconnect valve is still closed.
- 2. Pressurize the N_2O_4 supply tank to TBD kN/m².
- 3. Open value at supply tank outlet. N_2O_4 will fill the transfer line.
- 4. Open vehicle disconnect valve. A fluid path now exists from the supply tank to the vehicle.
- 5. Open valves #7 and #8. Propellant transfer to Tanks #1 and #2 is in progress. (Refer to Figure 5-6 schematic.)
- 6. Close values #7 and #8 when mass gauging devices indicate the tanks are full. Note: The initial tank pressures of 207 kN/m^2 (30 psia) will guarantee the tanks can be filled without exceeding the 1720 kN/m^2 (250 psia) operating pressures.
- 7. Open valves #5 and #6 to commence RCS tanks refueling.
- 8. Close valves #5 and #6 when mass gauging devices indicate the tanks are full.
- 9. Vent supply tank to reduce pressure to TBD kN/m^2 .
- 10. Close vehicle disconnect valve and purge transfer line with GH_e employing the following procedure:
 - a. Close supply tank valve.
 - b. Open supply tank acquisition device by-pass valve.
 - c. Initiate low flowrate GH_e purge through transfer line. GH_e enters at transfer valve disconnect and flows toward supply tank, forcing N_2O_4 into the tank.
- 11. Disengage transfer line disconnect and return to orbiter cargo bay. The LTL vehicle N_2O_4 refueling operation is now complete.





,

5-25

The flex duct assembly runs the full length of the valve and consists of two short tube sections, three bellows, one poppet with an actuating rod assembly, two guide vane fittings, one mounting flange, one interface fitting with seals, one fitting for actuating the poppet and one inlet fitting. Except for the seals and the poppet assembly, all parts are 304L CRES and are welded into one assembly. The seals would probably be a rubber type compound molded or bonded into the interface fitting and the poppet. The poppet assembly will probably be 304L CRES machined parts assembled with threaded fittings. The poppet spring material is 178 Inconel.

The flex duct assembly is the heart of the valve system. Opening or closing is accomplished by compressing or extending the bellows with actuators located outside the flow stream. No dynamic seals are required. The spherical connection between the actuator and the interface fitting plus the ball socket type connection between poppet and actuating rod permits angular misalignments between sealing surfaces.

The actuator system is two harmonic type drives. Each drive is integrated with a hollow shaft, electric motor. The drives and motors are configured to allow the flex duct assembly to pass through at the center line. To permit assembly, each drive has a split collar for attaching to the flex duct assembly. This type of actuator system permits a compact assembly but does require nume rous parts. The main parts for each drive are one wave generator, one flex spline, a circular spline and a dynamic spline. Approximately seven bearings will be required (plus retainers), including two fer the motor. Two additional screw jack parts are also required on the output end of each drive.

The outer housing consists of a cylindrical section which is equipped with a conical fitting at each end. One conical fitting has two external machined lands plus a flange for mating with the conical section on the LTL. The purpose for the conical mating surfaces is to permit loose engagement even with large misalignments plus accurate alignment when the cones are pulled together.

The second conical fitting has two flanges; one for attaching to the cylindrical section and one for the flex duct attachment. This conical fitting provides a rigid interface for connecting the transfer duct and reacts the pressure area loads from the flex duct bellows. To permit assembly, a split collar is used at the flex duct attachment.

The outer housing cylindrical section has three external longerons for supporting the latching systems and an internal spider beam for supporting the actuators. Each end is also equipped with a flange for attaching the cones. A fitting for attaching to the RMS is located between two of the longerons. This fitting is not shown on the layout.

All outer housing parts are 2219 aluminum alloy. The cones can be either single spun formed parts or weldments consisting of rolled cones and machined ring

flanges. The cylindrical section is a weldment. The latching systems are basically an over center device powered by an electromechanical screw jack actuator. The locking pawls are capable of rotating outboard combined with axial displacement away from the housing cone. The pawls can also rotate inboard followed by an axial movement toward the housing cone. This type of kinematics permits gross initial capture with the mating cone on the LTL followed by draw down and final alignment.

The passive section of the valve which attaches to the LTL is a single piece cone and valve body equipped with a spring loaded poppet. The valve body features a flat faced sealing surface equipped with flow passages for interface purging and seal monitoring. Except for the poppet spring, the entire assembly is 2219 aluminum alloy.

Flow passages are provided on both sides of the sealing plane. These passages provide a means for monitoring leakage and for purging the interface cavity. Small tubes with flex loops are routed from these passages to solenoid control valves. The circuits are activated with GH_e . Other additional means for monitoring leaks is the use of sniffers. GH_e supply for the system indicated can be from tubes routed along the transfer duct or from a small bottle (with controls) attached to the valve housing. A second small bottle (attached to the valve housing) can also be used to act as a catch reservoir when purging the interface cavity. The use of bottles attached to the valve simplifies the transfer line assembly since only one electrical cable is required.

An operation sequence is shown on Figure 5-14. Referring to Step 1, the two valve sections are in a gross capture mode. At this stage the latches are rotated inboard which captures the cone flange on the LTL side.

In Step 2 the latches are actuated parallel to the valve axis pulling the two cones together. The structural connection is now completed.

In Step 3 one of the harmonic drives is actuated which engages the primary seals with the mating flat face. Stop lands machined on the flange containing the seals controls the amount of squeeze on the seals. A leak check is performed by pressurizing the cavity between the two primary seals and monitored for pressure decay.

The second harmonic drive is actuated in Step 4 which opens the value for transfer. During transfer, the main seal is monitored with a sniffer located in the GH_e supply tube.

The valve is closed at the completion of propellant transfer, as indicated in Step 5, and the small interface cavity purged. The sealing arrangement for this purge circuit is shown in Detail "A". Before disengaging, the transfer line is purged and the poppet seal on the LTL side is checked for leaks by activating the sniffer circuit in the tube leading to the interface cavity.

In Step 6, the main seal is disengaged, the latches opened, and the valve separated by the RMS.

5.4.2 HELIUM BOTTLE RESUPPLY. Space-based vehicles will require gaseous helium resupply during refueling operations. One method of resupply is by flowing helium from a storage tank (located in the Shuttle payload bay) to the vehicle through a long transfer line. An alternative would be to employ separate pre-loaded modular helium bottles that would be externally attached to the vehicle. A disconnect system capable of mating a structural attachment and a fluid connection with the vehicle is required with either approach. Several disconnects are required of the modular bottles, however, as compared to only one for a transfer line.

The orbiter remote manipulating system (RMS) will be employed to connect transfer line or modular bottles to the vehicle. The RMS will place the disconnects reasonably close to the mating target. Pressure-area loads will be reacted only by the disconnect structure, and not by the RMS.

5.4.2.1 Helium transfer from orbiter — The transfer line approach will have an electrical power cable attached to the line and permanently connected to the disconnect system located at the end of the line. The transfer line must have mobility which in turn requires flex joints either in the form of braided hoses, swivel type joints with dynamic seals, or loop bends in the tubing. Conventional bellows joints would not be applicable due to high pressure requirements. The use of hoses, swivel joints and loop bends presents packaging and weight penalties. Also, the safety of this transfer operation is of concern because a 21m(70 ft.) line, pressurized to $34500 \text{ kN/M}^2(5000 \text{ psi})$, will be essentially unsupported except at the orbiter and at the vehicle.

Aside from the question of safety is the problem of excessive helium temperatures occurring the bottle charging period. This problem exists because the heat of cempression generated during the charge period cannot be readily dissipated; and may require the aid of space radiators. Potential solutions are summarized in Table 5-3, only one of which was considered acceptable.

5.4.2.2 Helium Modules — The preferred method for LTL vehicle helium resupply is to use helium bottle modules.

Basically, the pre-loaded helium modules are picked up with an RMS and plugged into external LTL interfaces which in turn have interconnecting plumbing to the LTL systems.

Detail "A" of Figure 5-3 shows a typical helium module which is a high pressure bottle equipped with a tangential support skirt, a docking cone, a latching system, a shutoff valve and a motor driven disconnect. The tangential skirt section has a fitting which interfaces with the RMS. The only requirement in the case of a Shuttle RMS

Re-supply Options
Tehicle Helium
LTL V
Table 5-3.

	- EBIC LE HAPDWARE REQUREMENTS	ADTAXTAGES	S FOX LAV AT A STORE
<u>Requirements</u> Pefill LTL belium bottles while in orbit			
Eroblene_Archient temperature belium transferred from origher to OMS and RCS belium bottles will create excessive belium temperatures during bottle charging period.			
<u>Solutions:</u> A. Secre billum in orbitier at reduced temperatures	ee	Bothe charging can be conducted In a timely manner.	Destred storage temperature and readily attained
 Cool belian: by flowing through radiator before it cours belian bottles. 	Nore	Bottle charging can be conducted in a timely manner.	Desired radiator outlet tomp rature may not be attained
 Charge slowly with helium and allow bottles to radiate energy to space. 	None	No modifications.	Charge process may be time consumined
D. Combine B and C	None	Bottle charging can be conducted in a threly manner.	Desired radiator outlet temperature may not be arened
 E. Replace expended helium bottles with full helium bottle. 	LTL pressurization systems must be re-designed for a modular concept.	Helium bottles replacement can be conducted in a timely manner.	Modification to LTL.

* Recommended solution.

5-29





,

5-30

Þ

would be the addition of an end fitting. This fitting would include a power cable from the Shuttle for actuating the latch and disconnect systems.

A typical transfer sequence consists of connecting an RMS to the helium module; placing the module loosely (wide tolerances) into a mating cone on the LTL; actuating the latches to an inboard position to insure a gross capture and finally moving the latches in an axial direction which completes the structural connection. With the structural connection completed, the disconnect system is energized making a seal between module and vehicle. The socket portion of the disconnect contains the seals and the probe section on the LTL is float mounted to compensate for misalignments. Since temperatures are basically ambient, the seal system consists of "O" rings equipped with backup rings to prevent "blow out". The seal design includes provisions for easy replacement.

5.4.3 ZERO-G MASS GAUGING. A zero-g mass gauging system will be as important to refill of earth storable vehicles as to POTV and COTV refill. The measurement of propellant mass quantities is critical at two intervals during on-orbit refill; when minimum liquid residuals and when final tanked mass must be measured. Total propellant load must be known to verify that sufficient propellant has been tanked to perform a mission. Propellant quantities in excess of requirements could unnecessarily penalize the vehicle mission. A continuous mass measurement during the latter stages of tank fill would also prevent the protential of tank over-pressure caused by over-fill.

The second occasion where mass measurement will be necessary is during the propellant tank venting procedure described by Table 5-1. Step 5 of this procedure states that a signal will be sent to terminate propellant transfer from one tank to another when the mass gauging device indicates that a minimum liquid residual remains. Screen gallery liquid replenishment (to make-up for surface evaporation) will be provided from this residual liquid volume. Less propellant than this minimum may not be sufficient to maintain communication with the screen device; resulting in vapor penetration due to the loss of liquid replenishment. Substantially more residual than the minimum will increase the probability that propellant can be vented overboard.



EXPERIMENTAL MODELING

On-orbit fill and refill of a propulsion tankage system is examined in this section to determine what experimentation is required to demonstrate this capability in a zero or low g environment. Emphasis is placed on identifying the major scaling parameters that must be satisfied in order to model the thermodynamic and fluid mechanic conditions of a refill operation. The influence of fluid properties and model tank scale on the validity of test results was also evaluated. Per the study guide-lines, the analysis effort was directed at conducting these experiments in low earth orbit inside the Spacelab.

Modeling or scaling analysis will be used to determine the feasibility, and subsequently the conditions and configurations, of the shuttle experiments for providing data useful in determining procedures for refueling a space-based vehicle. This discussion on scaling will focus on our understanding of the fluid phenomena as well as the complexities involved in experimental modeling.

Based upon the analyses performed in sections 3, 4 and 5, the important areas of cryogenic and earth storable propellant tank on-orbit refill have been identified. These areas, for a cryogenic stage, are tank vent, prechill and fill. The only area of concern for an earth storable stage is tank vent (experimentation is not recommended, as discussed in section 6.). The influence of such factors as helium present within the propellant tanks, and partial acquisition devices was also determined.

Not all of the important elements of a refill process should be subjected to a rigorous experimental program. Rather, only those elements that require verification, or processes which phenomenon is not well understood, should be considered for experimentation. As an example, propellant tanks vent will be an important part of any cryogenic refill operation. There is not a compelling reason, however, for performing such experiments in space. Means can readily be devised for safely venting the propellant tanks without just performing orbital tests. Those areas requiring experimentation are listed below. Specific outputs are identified that will provide design criteria and procedures for refueling operations of space-based OTVs:

1. Tank Prechill - Obtain empirical data to establish relationships for scaling peak tank pressures, prechill times, and vent mass requirements.

- 2. Tank Fill Obtain empirical data to establish the influence of propellant tank inflow parameters on thermodynamic equilibrium conditions.
- 3. Support Experiments Criteria Identify an empirical relationship between start basket in-flow parameters and successful refill.

Each phase of the propellant-transfer process will require a separate scaling analysis. Three basic scaling methods were considered in this study: (1) dimensional analysis, (2) order of magnitude analysis on the equations of motion, and (3) ratio of phenomenological equations.

Dimensional analysis is the classic method, and it provides a simple and direct procedure for obtaining all parameters that may affect a process. Since exact scaling is not usually possible, however, we must determine which parameters are most important. Dimensional analysis alone does not provide a solution to the problem, and a significant amount of experimentation is required to determine exact relationships between the resulting dimensionless groups. The complex heat and mass exchange mechanism associated with orbital refill did not readily lend itself to dimensional analysis.

The order-of-magnitude and ratio methods require writing the equations that describe the process. With order-of-magnitude analysis, the boundary value problem that describes the process is written and transformed into dimensionless variables. The important variables are determined from an order-of-magnitude analysis and the lower-order terms are neglected in the scaling process. The ratio method is the same approach that is used when an analytical solution is described.

If the differential equation can be solved accurately, the scaling problem is redundant and is replaced by an exact solution. However, it is usually impossible to solve the differential equation with boundary and initial conditions without making some gross simplifying assumptions that may considerably reduce the credibility of the mathematical model.

The ratio method was employed as the scaling technique for the orbital refill process. Empirical equations are identified in Sections 6-2 and 6-3 which describe the prechill and tank fill processes of a cryogenic vehicle refueling operation. The key variables of these processes were readily identified. Model test flow parameters and time scaling relationships were subsequently identified as a function of tank scale.

6.1 RECEIVER TANK SCALE

Normally, when an experimental test program is defined, scaling equations derived from the modeling analyses will serve to identify the tank scale and fluid selection. For this study however, a groundrule to perform tests within the Spacelab facility limited the test tank size to a 1/10th scale maximum. The following relates the specifics of tank size, in addition to tank shape.

6.1.1 RECEIVER TANK SHAPE. The OTV includes liquid hydrogen and liquid oxygen propellant tanks, both of which will be refilled in space. A single receiver tank will be selected for conducting orbital refueling experiments, and the question is, which propellant tank configuration shall be tested?

It was shown in Section 3.3.2.1 (Figure 3-14) that the liquid oxygen propellant tank can be refilled more easily than the liquid hydrogen tank becuase it will not experience excessive pressure during refueling. The conclusion was made, correctly, that liquid hydrogen tank refueling should be subjected to experimentation in order to verify the selected refill technique. However, it will always be more difficult to refill any tank with liquid hydrogen than with liquid oxygen; fluid property differences are responsible for this condition. Consequently, the difficulty in LH₂ tank refill is due to the propellant, not the tank shape. If liquid hydrogen is eliminated as a test fluid, what justification exists for using the LH₂ tank configuration?

It appears that adequate mixing of the liquid and vapor phases will be easier to attain in the LO_2 tank than in the LH_2 tank. This conclusion seems valid if we compare two tanks (having the same volume), where one is cylindrical and has a large lengthto-diameter ratio, and the other is a sphere. Intuitively, uniform mixing in a cylinder should be more difficult to achieve than in a sphere. Selection of the LH_2 tank configuration will guarantee that the most difficult configuration for orbital refill will be tested.

6.1.2 TEST SCALE. In general, when an experiment plan is developed it is necessary to compromise between the desire for a full scale test program and the limitations imposed by resources and facilities. Large scale tests are always desirable because the uncertainty of extrapolating test data to a prototype condition can be minimized. Thus, for the Spacelab experiment, the desire is to design the largest tank scale that can be accommodated. In this case, test tank size will be limited to a package that fits within a doublerack structure, Figure 6-1. Design details are provided in the following discussion.

6.1.2.1 <u>Preliminary test tank design</u>. In Figure 6-1, an experimental apparatus is shown positioned in a Spacelab doublerack structure, which will demonstrate onorbit propellant loading of OTV's. This apparatus consists of a scale model OTV LH₂ tank suspended within a vacuum-jacketed shell. The intent of Figure 6-1 is to determine a basic tank size for the apparatus within the nominal envelopes described by the Spacelab payload accommodation handbook, Reference 6-1. Details for plumbing, wiring and supports are emitted. For this case the areas controlling the tank size are Zones "B" and "C" shown cross hatched on the layout. Zone "B" is reserved for payload cabling and Zone "C" is reserved for subsystem access.

The apparatus is positioned near the right wall of the rack (as viewed by the operator) so that the tank bulkhead clears the corner of Zone "C". Tank length is then limited by Zone "B". Adjustments between diameter and length were made within these confines to arrive at an acceptable L/D ratio. The tank shown has a 457 mm (18 in.)



-

٤

Figure 6-1. Orbital sfill Experiment Installation in Spacelab Standard Double Rack

6-4

. .

٨

diameter and a 1092 mm (43 in.) overall length. The tank is equipped with an outer shell having a 559 mm (22 in.) diameter and a 1194 mm (47 in.) overall length. The selected location offers a volume directly below the tank bottom bulkhead which can be used for plumbing and wiring protrusions as shown on the layout. Additional adjustments in tank diameter and length can be made by minor infringements into Zones "B" and "C". For example, it appears that the bottom bulkhead could cut across the corner of Zone "C" without seriously compromising accessibility. For this effort, however, the tank size shown is the maximum within the nominal envelopes.

In Figure 6-2, details are shown for the experiment apparatus using the diameters and lengths established in Figure 6-1. The purpose of Figure 6-2 is to generate a detailed weight breakdown for the tank portion only of the apparatus so that the main drivers can be identified. This weight breakdown in turn was used to determine any revisions to the initial design in an effort to further reduce weight. The outer jacket is included to show the general relationship within the tank and the plumbing.

The tank is a 457.2 mm (18 in.) dia x 762 mm (30 in.) length cylinder equipped with ellipsoidal bulkheads (a/b = 1.38) at each end. The material is 2219-T87 aluminum alloy and the minimum gage is 0.51 mm (0.020 in.). The tank is supported from the outer jacket with three pairs of struts at one end and three drag links at the opposite end. Three fill manifolds are installed inside the tank. Provisions for ground fill and drain, vent and electrical, are also included. To permit hardware changes during ground tests, one 132.4 mm (6.0 in.) I. D. access opening is provided at each end. The external surface of the tank is equipped with strip heaters and a multilayer insulation (MLI) blanket.

Referring to the figures, the bulkhead labelled No. 1 has two 1.02 mm (0.040 in.) gage weld zones. The first zone is for the access opening ring and the second zone located at the girth is for the butt weld between the cylinder and bulkhead. This second zone also contains three tangential fittings for the drag link supports (see detail "G"). The bulkhead labelled No. 2 contains a wide weld zone at the girth which contains three pairs of tangential support fittings (see Detail "C"), and the tank wall penetration fittings. This zone also provides the gage increase for the butt weld at the cylinder. A second weld zone near the top of the bulkhead is for the access opening ring. The cylindrical portion of the tank has one weld zone at each end and one running longitudinally. Support lugs for the fill manifolds are welded to the inside surface of one of the end zones.

A typical fill manifold is a length of aluminum alloy tubing equipped with a tee fitting at each end. The side branch of each tee fitting is threaded for attaching spray heads. One of the tee fittings has a protruding rod section (see Detail "J") which engages with the support lug attached to the tank near Bulkhead No. 1. This is a sliding support which provides restraint in any direction normal to the tube while permitting

6-5



Figure 6-2. Test Tank Design Details

.




axial movement. The opposite end of the tube (near bulkhead No. 2) is butt welded to the tank wall penetration fitting as shown in Detail "B". The manifold is welded to this penetration fitting prior to installation.

One internal tube extending from bulkhead No. 2 to bulkhead No. 1 is provided for ground venting. Similar to the fill manifolds, this vent tube is welded to a tank wall penetration fitting at bulkhead No. 2. The opposite end of the tube (at bulkhead No. 1) is supported in a manner similar to that described for the fill manifolds.

Ground fill and drain is provided by a penetration fitting located at bulkhead No. 2. This is not shown on the drawing but is similar to that shown for the fill manifolds.

Internal instrumentation is routed through a boss (located through bulkhead No. 2) equipped with a flanged multi-pin connector. A radial seating seal is used for sealing the connector to the boss. An arrangement is shown in detail "B" of Figure 6-3 and an alternate approach is shown in Detail "L". All instrumentation wires inside the tank are bundled into a single cable and supported from the vent tube with clips.

The outside tank surface is equipped with heaters. Several types are available such as coatings directly deposited on the tank wall, blanket types which are fitted with a glove over the tank, and strip or ribbor types which are bonded to the tank wall. Due to versatility and simplicity, the strip type was selected. The heaters are approximately 1.5 in. wide and are a ranged in a circumferential pattern along the length of the cylinder. This circumferential pattern may also be extended to the bulkheads, cr a longitudinal type pattern can be used similar to gore lines on a bulkhead. All wiring is supported from the tank wall with tape strips which are lapped over the wire and bonded to the tank. The wires are bundled into a single cable at the No. 2 bulkhead and routed through the MLI blanket at a single point.

The entire tank sorface, including strip heaters, is covered with a multilayer insulation (MLI) blanket. The blanket is applied in gore sections which interface with two circular cap sections located on the ends of the bulkheads. The sections are attached to the tank wall using "Velcro" tape patches. The pile portions of the tapes are bonded to the tank wall and the hook portions are bonded to the blankets. Cutouts in the blankets provide clearances for the plumbing, wiring and support struts.

The outer jacket is a cylinder equipped with two removable ellipsoidal bulkheads. The material is 2219-T87 aluminum alloy and the wall gage is sized to 207 kN/m^2 (30 psi) external pressure. The cylinder is equipped with two rings (one at each end) which are attached to the tank support struts. These rings also incorporate flanges which interface with the bulkheads and external fittings for attaching to the rack structure. The flanged connections between bulkheads and cylinder use metal "O" rings or radial seating "Cono" seals. All plumbing and wining circuits penetrate one bulkhead only through cup/sleeve fittings. The cup fittings are welded to the plumbing and the sleeve is welded to the bulkhead. The ends of the cup fittings are sealed to the sleeves with peanut type welds (see detail "K"). When removing the bulkhead, the peanut welds are ground off. When replacing the bulkhead, the cups are rescaled with peanut welds. The length of the cup and sleeves are determined for severa bulkhead removables. A vacuum environment is maintained in the annulus volume between tank and jacket during ground test and Spacelab operation. Therefore, the bulkhead is equipped with a flanged penetration fitting for attaching a duct.

Referring to the parts list in Figure 6-2, the total weight for the complete tank assembly is 5.8 kg (13.0 lb). The basic tank shell weighs 2.76 kg (6.08 lb) including the weld lands. The balance is for accessories which accounts for 53 percent of the total weight. Ring3 and covers for the access openings represent a major portion of the accessory total weight, and these can be eliminated if we are willing to compromise accessibility to the tank interior. For example, access to the tank interior can be accomplished by simply cutting bulkhead No. 1 off at the weld line and rewelding. The width of weld lands would be greater initially than the 25.4 mm (1 in.) shown to permit several cuts, trims and re-welds. This method has risks however, because the procedure amounts to reworking a thin walled aluminum tank with a minimum gage of only 0.55 mm (0.020 in.). Another possible area for weight reduction is the electrical penetration fitting. 0.16 kg (0.34 lb) can be trimmed from this fitting by using the design shown in Detail "L". This alternate design eliminates the separate clamping flange, and integrates the collar (which welds to the tank wall) with the receptacle. The seal is also eliminated. Cooling provisions for the receptacle would be required during welding to prevent damage to the core material which contains the conductor pins. In summary, for both the access holes and the electrical penetration fitting, a 1.93 kg (4.24 lb) weight reduction can be realized which results in a total tank weight of 4 kg (8.84 lb), 31 percent of which is for accessories.

A weight and configuration summary of this tank design is given in Table 6-1. A comparison of test tank to prototype is given in Table 6-2. Note that this tank scale, which is the largest that can be designed into the Spacelab doublerack structure, is 0.108 of the POTV.

			Conventiona	Design		
Layout No.	Tank Volume m ³ (ft ³)	Tank Surface Area w ² (ft ²)	Total Tank Weight Inc'l Accessories, kg (b)	Accessories % of Total	Non-conven Total Tank Weight Inc'l AC:essories	Accessories 7 of Totai
#2	.16 (5.7)	1.64 (17.6)	5.8 (13.0)	Tank Weight 53	kg (lb)	Weight, kb (lb)
#3	. 93 (32. 76)	5.1 (55.0)	10.8 (23.8)	32	(*8.84) 9.0	17
π4	3.13 (110.7)	11.5 (123.9)	25.4 (60.2)	15	(19.9) 25.4 (55.9)	9

Table 6-1.	OTV LH2 Model	Test Tont	TT 7 • • • • •
		- COU I ANK	weight Summary

6-9

		Conventiona	d Design	Non-conventional Design		
Tank Scalo L*	Tank Diameter em (in)	$\frac{V \cdot M}{m^3 \cdot kg(ft^3 \cdot lb)}$	V.* (1)	V/M m ³ /kg(ft ³ /lb)	V + M + (1) m ³ kg (ft ³ 1b)	
.108	45.7(18)	. 0278 (. 438)	. 105	. 0410 (. 645)	. 155	
. 20	85.9(33.8)	. 0874 (1. 376)	. 331	.1045 (1.646)	. 396	
- 30	128.8(50.7)	. 1168 (1. 839)	. 442	. 1257 (1. 980)	. 476	

Table 6-2.Model Test Tank Volume-to-Mass Ratio ComparisonWith OTV LH2 Prototype Tank

Prototype Tank (PDTV) Measurements

Volume (V)= 116 m^3 (4100 ft^3) Mass (M)=447 kg(986 lb)Diameter=4.29 m(169 in)Cylinder length=59.5 m(234.2 in)

(1) (*) refers to ratio of model-to-prototype. An exact (and ideal) scaling of tank properties would result in $V^*/M^* = 1$.

6.1.2.2 <u>Larger test tank designs</u>. The inexact sealing of the one-tenth scale model will necessarily create a variance between model and prototype test results. This variance is related to the volume-to-mass ratio differences between tank scales, which is quantitatively evaluated in section 6.2. There was an interest in determining how the volume-to-mass ratio would vary with tank scale. Consequently, a preliminary design was also performed on a two-tenths and a three-tenths scale tank. The two-tenths scale model is basically the same as that described in Figures 6-1 and 6-2. The only differences are minor items such as weld land areas, plumbing lengths, support fitting sizes and quantity of support lugs for the internal plumbing. No outer jacket is shown since this size tank would be transported outside the Space-lab. The tank would probably be suspended from a truss cylinder which in turn interfaces with the shuttle payload support journals.

Referring to the parts list in Figure 6-3, the total tank weight is 10.8 kg (23.8 lb) of which 32 percent is accessories. If the access openings are eliminated and the electrical penetration fitting simplified as described in Layout No. 2, this total tank weight is reduced to 9.0 kg (19.9 kg), of which 17 percent is accessories.

Figure 6-4 is the same as Figure 6-3, except the tank is a three-tenths model. For this case, the basic wall gage was increased from 0.51 mm (0.020 in.) to 0.635 mm (0.025 in.). Minor items such as weld land areas, plumbing lengths, plumbing sizes, and quantity of support fittings have been increased. Results from these two designs are summarized in Tables 6-1 and 6-2.



Figure 6-3. Test Tank Design Details (0.2 scale)

6-11

のであって



6-12

6.2 PRECHILL MODELING

The prechill process that has received much attention during this study is illustrated by Figure 6-5. The process will be accomplished by 1) metering liquid at a high velocity into the receiver tank for a fixed duration, 2) allowing heat exchange between fluid and tank walls for an unspecified duration until tank pressure has increased to the vent level, and 3) venting the tanks back down to near zero pressure. The carly prechill period is characterized by a complex thermodynamic and fluid mechanic process due to liquid impingement on the hot tank walls. The resulting forced convection nucleate and film boiling phenomena are extremely difficult to analytically model or scale.

Fortunately, it can be shown that this initial period of tank chill may not be significant to the overall process. First, the heat exchange during the limited boiling period represents only about ten percent of the total energy removed during prechill. Second, we should be more concerned with the tank conditions prior to tank vent rather than with the initial transient. The initial transient pressures will be well below the nearsteady-state pressures if propellant inflow is adequately controlled. Figure 3-15 indicates that precision metering of LH₂ is not needed to avoid overpressure during prechill. Thus, effort can be concentrated on scaling steady-state conditions of the prechill process.



Figure 6-5. A Prechill Procedure Can be Identified to Eliminate Excessive Tank Pressures Due to Wall Boiling

6.2.1 SCALING PEAK PRESSURES. This is an important requirement because excessive pressures must be avoided during POTV on-orbit prechill. Peak pressure will occur as tank and fluid temperatures reach equilibrium. Figure 6-6 shows the theoretical maximum pressure for a POTV liquid hydrogen tank as a function of tank thermo-physical properties and geometry. This figure also shows that tank volume-to-mass ratio (V/M) is an important variable.



Figure 6-6. Maximum Pressure During OTV Propellant Tank Prechill

Figure 6-7 suggests that the pressure scaling parameter is PV/M. These curves have been generated for a V/M range of .125 m³/kg (2 ft³/lb) to .259 m³/kg (4.16 ft³/lb) (V/M for OTV is .259 (4.16)). This figure shows that PV/M is a function of initial tank temperature, tank material, and propellant. By selecting these variables to be the same for a model test as for the prorotype (full scale) vehicle, the resulting PV/M will also be the same. Expressed mathematically, $(PV/M)_m = (PV/M)_p$, or

$$P*V*/M* = 1$$
 (6-1)

where:

	P V M	=	peak tank pressure during prechill tank volume tank mass
subscript,	m p	=	model prototype
superscript,	(*)	=	the ratio of model to prototype.

When model test variables are selected such that P*V*/M* = 1,

then

$$P^* = P_m/P_p = 1$$
 (6-2)

Equation 6-2 means that prototype tank peak pressures will be equal to model test tank peak pressures. There are other boundary ∞ nditions (discussed later) to be satisfied for the above statement to hold true.

If test constraints are such that $V^*/M^* \neq 1$, then

$$P^* = M^*/V^*$$
, or $P_p = P_m (V^*/M^*)$ (6-3)

In this instance, P_p is determined by multiplying the observed P_m by (V*/M*).



Figur: 6-7. PV/M is a Parameter for Scaling Peak Pressures During Prechill

6.2.1.1 <u>Model Tank Size Influence</u>. Test tank scale will have a major affect upon the peak pressures experienced during experimentation, because of the V^*/M^* influence. Figure 6-8 shows the relationship V^*/M^* and tank scale (L*), where L is a characteristic tank dimension. Figure 6-9 shows the influence of scale upon model tank pressures when modeling of POTV peak pressures is attempted. Referring to section 3.3.2.5 (Figure 3-19), a POTV peak pressure of about 69 kN/m^2 (10 psia) was selected as being acceptable for a prechill procedure. The .108 scale model (for Spacelab) would experience a pressure of nearly 690 kN/m^2 (100 psia) under similar conditions. Thus, it is seen that prechill experiments conducted on Spacelab would produce results substantially different from what would be predicted for a prototype vehicle.



Tank Scale, L* Figure 6-8. Test Tank Scale Influence on V^*/M^*



Model Tank Pressure, N/m^2 (psia)

Figure 6-9. Peak Prechill Pressures May Be Excessive for Small Scale Experiments

6.2.2 <u>TIME SCALING</u>. A second major requirement of the prechill experiment will be to develop a time scaling parameter applicable to the tank charging process. This is an important objective because we need to establish propellant transfer timelines. A potentially useful parameter may be developed by assuming the receiver-tank charging process to be equivalent to the transient heat conduction process of a lumped-mass system. Consider the prechill condition described by Figure 6-10:

- 1. LH_2 flow to OTV is initiated at zero time.
- 2. A convective heat transfer coefficient exists at the end of the flow period as a result of inflow conditions.



Figure 6-10. (hA/m Cp)t is Applicable as Time Scale Parameter for Prechill Process

3. The temperature-time history as indicated by Figure 6-10 can be characterized by

$$\frac{T - T}{T_{o} - T_{\infty}} = e^{-(hA/mCp)\tau}$$
(6-4)

where

Т	=	Fluid temperature at time, t
T	=	Initial fluid temperature
T 🗙	=	Equilibrium temperature of tank and fluid
h	=	Fluid/tank-wall heat transfer coefficient
Α	=	Tank wall surface area
m	=	Fluid mass
Ср	=	Fluid heat capacity
au	=	Time
θ	=	(hA/mCp) τ , dimensionless time

It is obvious from equation 6-4 that the dimensionless temperature parameters will be identical if $\theta^* = 1$. Furthermore, the actual temperature changes will be identical if we impose the additional requirement that

$$m */M * = 1$$
 (6-5)

where

m = fluid mass M = propellant tank mass

This requirement is derived from the expression for tank-vapor energy exchange that occurs during prechill. The following expression describes energy exchange between tank and vapor.

 $\mathbf{m} \mathbf{C}_{\mathbf{v}} \Delta \mathbf{T}_{\mathbf{v}} = \mathbf{M} \mathbf{C}_{\mathbf{w}} \Delta \mathbf{T}_{\mathbf{w}}$ (6-6)

where

 ΔT_{v} = fluid temperature change ΔT_{w} = tank wall temperature change

C = heat capacity

6-18

subscript

w = tank wall v = fluid vapor

Taking the ratio of model to prototype gives

$$\mathbf{m}^* \Delta \mathbf{T}^* = \mathbf{M}^* \Delta \mathbf{T}^*_{\mathbf{W}}$$
(6-7)

or

$$m^{*}/M^{*} = \Delta T_{w}^{*}/\Delta T_{v}^{*} = 1$$
 (6-8)

By imposing a) the requirement of 6-8 and b) the same initial tank temperature for model and prototype, we are assured that tank temperature changes will be identical.

All variables of the dimensionless time parameter, with the exception of h and A, can be described. However, Equation 3-11, shown in Filling of Orbital Fluid Management Systems, NASA CR-159404, relates h to fluid properties and tank inflow conditions. This relationship is given as:

$$\frac{h}{\rho_{\rm C}} \left(N_{\rm PR}\right)^{2/3} = C_1 \left[\frac{(mv^2/V)\mu}{\rho^2}\right]^{1/4}$$
(6-9)

where:

ρ = Fluid density C_p = Constant-pressure heat capacity $^{\mathrm{N}}\mathrm{_{PR}}$ = Prandtl number m = Entering mass flow rate v = Entering fluid velocity V = Tank volume μ = Fluid viscosity с₁ = Empirical coefficients <u>mv</u> = Fluid inflow parameter

It will be now possible to identify the relationship required to satisfy $\theta^* = 1$. This development is given below for the case where tank material are the same for both model and prototype conditions:

Normalizing the dimensionless time parameter, we have

$$\theta^* = h^* A^* \tau^* / m^*$$
 (Cp* = 1) (6-10)

From (6-9) we have

h* =
$$(\dot{m} * v *^2 / V *)^{1/4} (\rho^*)^{1/2}$$
 (6-11)

But $\rho^* = m^*/V^* = \left(\frac{m^*}{M^*}\right) \left(\frac{M^*}{V^*}\right) = \frac{M^*}{V^*}$ since $\frac{m^*}{M^*} = 1$, from (6-5)

Thus, (6-11) becomes

h* =
$$(\dot{m} * v *^2 / V *)^{1/4} (M * / V *)^{1/2}$$
 (6-12)

Combining 6-10 and 6-12 results in

$$\theta^* = (\dot{m}^* v^* / V^*)^{1/4} (M^* / V^*)^{1/2} (A^* \tau^* / m^*)$$
(6-13)

$$\dot{m}^* = m^* / \tau^*$$
 (6~14)

Combining 6-13 and 6-14 gives

$$\theta^{*} = (\mathbf{m}^{*} \mathbf{v}^{*2} / \mathbf{V}^{*})^{1/4} (\mathbf{M}^{*} / \mathbf{V}^{*})^{1/2} (\mathbf{A}^{*} / \mathbf{m}^{*})$$
(6-15)

(6-16)

now, A*

Also the fluid inflow parameter can be modified as follows,

$$\dot{\mathbf{m}}^* \mathbf{v}^* / \mathbf{V}^* = \dot{\mathbf{m}}^* / (\mathbf{A}_n^* \mathbf{V}^*) - \dot{\mathbf{m}}^* / \mathbf{L}^*$$
(6-17)

where:

 $A_n = nozzle flow area$

= L^{*^2}

Combining (6-15), (6-16), and (6-17) gives

$$\theta^{*} = (\dot{m}^{*} / L^{*})^{1/4} (M^{*}/V^{*})^{1/2} (L^{*} / \dot{m}^{*})$$

$$= \left(\frac{L^{*}}{\dot{m}^{*}}\right)^{1/4} (M^{*}/V^{*})^{1/2}$$
(6-18)

Finally, in order to have similarity between model and prototype dimensionless temperature parameters we require that $\theta^* = 1$, which will be satisfied if

$$(L^{*}/m^{*})^{1/4} (M^{*}/V^{*})^{1/2} = 1$$
 (6-19)

An experiment model test package design will be heavily influenced by such variables as tank size, test duration, flowrate and velocity requirements, and the system pressures needed to provide these flowrates. Flowrate, \dot{m}^* , can be obtained from (6-19).

$$\left(\frac{\dot{m}^{*}}{L^{*}}\right)^{1/4} = \left(M^{*}/V^{*}\right)^{1/2} \text{ or, } \dot{m}^{*} = L^{*} \left(M^{*}/V^{*}\right)^{2}$$
 (6-20)

Velocity, v^* , can be determined from (6-20),

$$\dot{\mathbf{m}}^{*} = \rho^{*} \mathbf{v}^{*} \mathbf{A}_{n}^{*} = \mathbf{v}^{*} \mathbf{A}_{n}^{*} (\rho^{*} = 1). \text{ Therefore,}$$

$$\mathbf{v}^{*} = \frac{\mathbf{L}^{*}}{\mathbf{A}_{n}^{*}} (\mathbf{M}^{*} / \mathbf{V}^{*})^{2} = \frac{(\mathbf{M}^{*} / \mathbf{V}^{*})^{2}}{\mathbf{L}^{*}}$$
(6-21)

Time, τ^* , is determined by combining (6-14) and (6-20),

$$\tau^{*} = \frac{m^{*}}{\dot{m}^{*}} = \frac{m^{*}}{L^{*} (M^{*}/V^{*})^{2}} = \left(\frac{m^{*}}{M^{*}}\right) \left(\frac{V^{*}}{L^{*}}\right) = \frac{1}{(M^{*}/V^{*})} = \frac{L^{*2}}{(M^{*}/V^{*})}$$
(6-22)

The three flow test variables of equation.³ (6-20) through (6-22) have been determined for the actual model tank confluarations of Table 6-2, and are given in Table 6-3. Note that a considerable variation exists in the flow parameters selected for the actual and "ideal" Spacelab experiment test tank. Unquestionably, the flowrates and velocities indicated by Table 6-3 cannot be attained for the actual tank model. Consequently POTV prechill cannot be exactly simulated with 0.109 scale model Spacelab tests, even if liquid hydrogen is used.

Table 6-3. Model tank scale influence upon test variables.

	N	lodel Tan	<u>scale, L*</u>	
Model Test Variables	0,108 ⁽¹⁾	0.2	0.3	0.108 ⁽²⁾
m* v* τ *	9.80 839.8 1.23×10^{-4}	1.825 45.6 .013	1.536 17.1 .040	.108 9.26 .0117

(1) Tank model for Spacelab experiment.

(2) Exactly scaled model for Spacelab experiment $(V^*/M^* = 1)$.

6.2.3 FLUID SUBSTITUTE. Liquid hydrogen cannot be used within the Spacelab under any condition. Liquid nitrogen is the only cryogenic alternative that may be acceptable. The liquid nitrogen quantities allowed within the Spacelab will be determined by a payload safety review group which convenes to evaluate experiments planned for the Spacelab. Any experiment which requires large quantities of liquid nitrogen would be carefilly reviewed to assure that inadvertent spillage would not create a hazardous environment. It was necessary to assume for this study that an experiment test package could be designed to circumvent potential problems, since a detailed design effort was beyond the scope of this effort.

Before pursuing the influence of LN_2 upon experiment modelling, a point will be made about why non-cryogens may be unsuitable for this experiment. It is believed that tank pressure during much of the fill process will be heavily influenced by heat and mass exchange between the liquid and entrained vapor. Heat and mass exchange is directly proportional to liquid-vapor surface area which, in turn, is dependent upon whether fluid agitation has created individual bubbles or a froth. Now, it has been observed that a frothy condition can readily be created when a non-cryogen, such as Fréon, is mixed with a vapor. Conversely, it has also been observed that LH_2 and LN_2 do not create a froth under similar conditions of agitation. This difference in behavior which may be difficult to quantify, coupled with the complex nature of the tank fill process has led to the conclusion that LN_2 is the only viable fluid substitute for the orbital experiment.

<u>Prechill.</u> The scaling effects of LN_2 upon the prechill process can be determined by employing the following relationships

$$\Theta^{*} = (h^{*}A^{*}\tau^{*})/(m^{*}C_{p}^{*}) \text{ (from Equation 6-10)}$$

$$h^{*} = (\dot{m}^{*}v^{*}/V^{*})^{0.25} \rho^{*} \rho^{0.5} C_{p}^{*}\mu^{*} \rho^{0.25}/Pr^{*2/3} \text{ (from Equation 6-9) (6-24)}$$

where

 $C_p = fluid heat capacity$ $\rho = vapor density$ $\mu = vapor viscosity$ Pr = Prandtl number

superscript, (*) = the ratio of model to prototype.

In this discussion, model refers to tests conducted with LN_2 and prototype refers to LH_2 tests. Also, for convenience, it is assumed that the model and prototype scale are the same. That is, $L^* = A^* = V^* = 1.0$ where L = characteristic tank dimension

6-22

(···)

and A = tank surface area.

Now, the following fluid property ratios apply for nitrogen and hydrogen:

$$\mu^* = 4.0$$

$$C_{\rm p}^* = .088$$

$$Pr^* = 1$$

$$\rho_{\rm L}^* = 11.4$$

Since we are at liberty to select any value for ρ^* , it was decided to select that value which results when model test and prototype peak prechill pressures are the same. This constraint was found to result in $\rho^* = 6.2$, from computer simulations. This is also the same value for m* since its relationship to density is

$$\rho^* = m^* / V^* = m^* \text{ (since } V^* = 1) \tag{6-25}$$

By working with Equations (6-23) and (6-24) and applying the hydrogen-nitrogen property ratios, it now is possible to compute the influence of nitrogen upon the model test parameters. First, Equation (6-24) combined with Equation (6-25) can be simplified to the following expression when substitutions are made for μ^* , Pr* and V*,

h* = 1.41 (
$$\dot{m}$$
 * v*²)^{0.25} m*^{0.5} C_p* (6-26)

Combining Equations (6-23) and (6-26) will result in

$$\theta^* = 1.41 (\dot{m}^* v^*)^{0.25} \tau^*/m^{0.5}$$
 (6-27)

and substituting $m^* = 6.2$ gives

$$\theta^* = 0.57 (\dot{\mathbf{m}}^* \mathbf{v}^*)^{0.25} \tau_*$$
(6-28)

From the continuity equation we have

$$\dot{m}^* = \rho_L^* A^* v^* = \rho_L^* v^* = 11.4 v^*$$
 (6-29)

or

$$v^* = 0.088 \,\dot{m}^*$$
 (6-30)

Substituting (6-30) into (6-28) gives

$$\theta^* = 0.17 \tau^* \dot{m}^{*0.75}$$
(6-31)

Also

$$m^* = m^* / \tau^* = 6.2 / \tau^*$$
(6-32)

which when substituted into (6-31) gives

$$\theta^* = 0.67 \tau^{*0.25} \tag{6-33}$$

Finally, setting $\theta^* = 1$ we have

$$\tau * = 4.96$$
 (6-34)

Substituting (6-34) into (6-32), and (6-32) into (6-30) gives

$$\dot{m}^* = 1.25$$
 (6-35)

$$v^* = 0.11$$
 (6-36)

Equations (6-34), (6-35), and (6-36) give the time, flowrate, and velocity ratios that must be applied if nitrogen rather than hydrogen is employed as a test fluid for prechill tests. These ratios can be used as multipliers for the respective ratios of Table 6-3 to obtain the combined influence of fluid and tank scale (for the stated tank scale assumptions). It is concluded that substituting LN_2 for LH_2 should not compromise prechill test results. In fact, there appears to be an advantage in using LN_2 because model test velocities will be substantially lower with this fluid, which may make it possible to achieve some of the conditions for a small scale test tank.

The primary disadvantage with using LN_2 as a test fluid is that there are fewer potential problems with prechilling a propellant tank with LN_2 than with liquid hydrogen. LN_2 's thermo-physical properties are very similar to those of liquid oxygen, which was rejected as the preferred test fluid. The arguments against LO_2 as a test fluid also apply to LN_2 .

6.2.4 PREDICTED PRECHILL TEST VARIATIONS FROM THE IDEAL. An exact prechill simulation cannot be achieved using liquid hydrogen and the 0.108 scale model hydrogen tank. It would be useful, however, to quantify the deviation from the ideal model test condition. To this end, the HYPRES computer program was employed to predict propellant tank temperature and pressure histories for several model test conditions. Program results are shown in Figures 6-11 and 6-12 for the assumed flow condition of saturated hydrogen vapor entering the propellant tank.

6.2.4.1 Zero-g test environment limitations. Figure 6-11 gives predicted propellant tank temperature versus time from prechill initiation for several test tank configurations and flow conditions. The abscissa represents the product of model test time and the time ratio obtained from Equation (6-22). Case 1 data represents an exact simulation

NOTES:

]

- 1. Curves are "HYPRES" computer runs simulating prechill tests conducted with a 0.108 scale LH_2 tank.
- 2. Prechill tests simulate the full scale conditions given in Table 6-4.
- 3. Predicted full scale tank prechill duration is "model test" time divided by the time scale factor of Table 6-4.
- 4. Case (1), (2) and (3) conditions are identified in Table 6-4.



Figure 6-11. Predicted LH₂ OTV Tank Temperature Histories From "HY PRES" Computer Simulation of 0.108 Scale Model Prechill Tests

where $V^*/M^* = 1$. These results are identical to those obtained for the full scale tank conditions of Table 3. Case 2 shows how tank conditions would be altered if substantially lower model test flowrate and velocity conditions were imposed. This deviation is small compared to deviations which results when V^*/M^* is not matched (although velocity and flowrate are matched), Case 3.

Figure 6-12 gives predicted pressure histories for the same test tank and flow conditions identified in Figure 6-11. Again, only the first charge and vent cycle is shown for the non-ideal model test case. Also, as before, Case 1 results were identical to those of the full scale tank. Similarly, Case 2 and Case 3 results were patterned after those of Figure 1. That is, an order of magnitude change in incoming flowrate had a minor influence on results, whereas an order of magnitude change in V/M had a major

6.2.4.2 One-g test environment limitations. Ground-based tests for determining the validity of the prechill process will be applicable only if a) the normal gravity free convection heat transfer process is dominated by the forced convection mechanism created by the entering propellant, and b) the fluid flow mechanism is the same for model tests as for the prototype configuration. To satisfy the former condition we must verify that the heat transfer coefficient given by equation (6-9) exceeds that free convection coefficient created by a normal gravity environment. Fquation (6-9) is applicable to heat exchange between a propellant tank surface and its contained vapor.

Prechill Process. The primary concern with normal gravity prechill tests is that inflow conditions required for similarity on the basis of tank scale (Table 6-4) will not be sufficient to guarantee the dominance of forced convection heat transfer. This point is illustrated by Figure 6-13 which shows how prechill test tank pressures will be affected a) by the influence of a normal gravity environment, and b) by the increased inflow conditions necessary to provide a forced convection dominated environment. Curve 1 is the predicted 45.7 cm (18 inch) diameter test tank pressure at the indicated flow conditions in a zero gravity environment. Obviously, the heat exchange mechanism is forced convection dominated. Curve 2 shows the predicted tank pressure if the same test is performed in a normal gravity environment. The tank wall-ullage heat exchange mechanism is free convection dominated. However, a forced convection dominated environment is necessary for a valid model test. This will require a factor of four increase in entering flowrate and velocity. The outcome is that a greater difference results between the zero-g and one-g test tank pressure profiles, Curves 3.

It is questionable that results of one-g tests that meet the criterion for forced convection heat transfer dominance can be employed to scale the zero-g prechill process.

6.2.5 PRECHILL SUMMARY. It is concluded from the computer simulations that the 45.7 cm (18 in.) diameter test tank cannot be employed to obtain results that are directly extrapolated to a prototype OTV vehicle. This is so even if tests are conducted with LH₂ in a near zero-g environment. Direct extrapolation becomes even less likely

NOTES:

- Curves are "HYPRES" computer runs simulating prech 'l tests conducted with a 0.108 scale LH₂ tank.
- 2. Prechill tests simulated the full scale conditions given in Table 6-4.
- 3. Predicted full scale tank prechill duration is "model test" time divided by Table 3 time scale factor.
- 4. Case (1), (2) and (3) conditions are identified in Table 6-4.
- 5. Case (1) results exactly simulate the full scale prechill process.
- Model Test Tank Configuration Scale ~ 45.7 cm (18 inch) V*/M* ~ 0.108

	$V^{+}/M^{+} \sim 0.108$			Flowrate
2.	Curve 1 is forced convoction domain to the	Curve	<u>G-Level</u>	kg/sec (lb-sec)
	Curve 2 is free convection dominated	1	0	.0045 (.01)
	Curve 3 is forged as	2	1	.0045 (.01)
	Curve 3 is forced convection dominated	3	1	.0180 (.04)



Figure 6-12. Predicted LH₂ OTV Tank Pressure Histories from "HYPRES" Computer Simulation of 0.108 Scale Model Prechill Tests

	Incoming Flow Rate, kg/sec (lb/sec)	Incoming Velocity m/sec (ft/sec)	Tank Scale	V*/M*	Time Scale
Prototype Condition	0.71 (1.56)	6.7 (22)	1.0	1.0	1.0
Case (1)	0.045 (0.100)	62. 2 (204)	0.108	1.0	0.0117
Case (2)	0.0045 (0.0100)	6.20 (20.4)	0.108	1.0	0.0021
Case (3)	0.045 (0.100)	62.2 (204)	0.108	0.105	1.28×10^{-4}

Table 6-4. Scale model test variables for simulating OTV LH₂ tank prechill.



Figure 6-13. A normal gravity environment will influence OTV model tank prechill test results

if LN_2 and/or a one-g environment is imposed as a constraint. It is expected, however, that the heat transfer phenomenon involved in the prechill process can be evaluated. Empirical coefficients obtained from such tests can be applied to an analytical model, such as HYPRES. This model can then serve as a tool for full scale vehicle prechill predictions.

6.3 TANK FILL MODELLING

Tank fill will be initiated after the prechill requirements have been satisfied. The single requirement for tank fill is to maintain an acceptably low pressure during the process. Tank pressures will be at a minimum if thermal equilibrium conditions are maintained during fill.

The intent of the tank fill process will be to create turbulent conditions within the tank. These conditions will be achieved by introducing liquid into the tank at high velocities (and perhaps through a spray nozzle) to provide the high heat-transfer rates needed to attain near-thermal equilibrium. As tank fill continues, the internal tank fluid environment changes from liquid droplets in the ullage volume to vapor bubbles entrained within a liquid bulk. The transition from heat transfer dominated by liquid droplets to heat transfer dominated by vapor bubbles is expected to occur at about the 40% to 60% liquid fill. This latter mechanism is the only mechanism that will influence tank pressures toward the completion of tank fill.

Tank pressure near the end of fill is more critical than during the early stage (since pressure does not become excessive in the interim), because the end state must reside within an acceptable thermodynamic range to satisfy mission and propulsion system requirements. For this reason, an evaluation is made only of the bubble dominant tank fill process.

6.3.1 VAPOR BUBBLE DOMINANT HEAT EXCHANGE PROCESS. Scaling parameters for this process can be developed by assuming tank fill to be equivalent to the transient heat conduction process of a lumped-mass system. Consider the heat exchange condition described by Figure 6-14 at some instant in time during fill.

- 1. Vapor is dispersed throughout the luquid bulk and resides at a higher temperature than the surrounding liquid.
- 2. Vapor dispersal and heat transfer is caused by fluid agitation, created either by a mechanical mixer or the entering liquid.
- 3. The temperature-time history of the liquid, as indicated by Figure 6-14, may be characterized by

$$\frac{T - T_{\infty}}{T_{o} - T_{\infty}} = e^{-(hA/mC_{p})\tau}$$
(6-37)



Figure 6-14. (hA τ/mCp) is applicable as time scale parameter for tank fill process.

where

Т	= liquid temperature at time, t
To	= initial fluid temperature
-T∞	= equilibrium temperature of vapor and liquid
h	= bubble-liquid heat transfer coefficient
A	= total surface area of the dispersed vapor phase
m	= liquid mass
Ср	= liquid heat capacity
τ	= time
θ	= $(hA/mCp)\tau$, dimensionless time

Note that the development of a time scaling parameter for the fill process begins identically to the approach selected for prechill. This parallel to the prechill analysis will be maintained throughout this evaluation.

The intent of a model fill test is to simulate the thermodynamic exchange between liquid and vapor during fill. Such a simulation would enable one to relate the model fluid temperatures to the prototype (or full scale) conditions. Tank pressure will be

influenced by fluid temperature during fill. And, if thermal equilibrium conditions can be attained, tank pressure will be a direct function of liquid temperature. Consequently, tank pressures observed during a model fill experiment can be directly related to the prototype only if near-thermal equilibrium conditions exist.

An inspection of equation (6-37) indicates that model fluid temperatures will be similar to those of the prototype if the dimensionless time parameter, θ , is the same for each condition. That is, if $\theta^* = 1$.

By taking the ratio of model-to-prototype for equation (6-37) we have

$$(T - T_{\infty})^{*}/(T_{0} - T_{\infty})^{*} = e^{-\theta_{m}}/e^{-\theta_{p}} = e^{-(\theta_{m} - \theta_{p})} = e^{-(\theta^{*}-1)\theta_{p}}$$
 (6-38)

For $\theta^* = 1$, equation (6-38) becomes

$$(T - T_{\infty})^{*}/(T_{0} - T_{\infty})^{*} = 1$$

(6-39)

Now, if in addition to $\theta^* = 1$ conditions can be selected such that $T^* = T_{\infty}^* = 1$, then $T^* = 1$ (from equation 6-39). This means that fluid temperatures will be identical for the model and prototype conditions at the same dimensionless time, θ . The remainder of this discussion is devoted to identifying model test conditions that will satisfy $T^* = T_{\infty}^* * = \theta * = 1$. Furthermore, all analysis is based upon the following assumptions:

- 1. Model tank geometry is identical to the prototype tank geometry.
- 2. Aluminum is material for both model and prototype.
- 3. Liquid hydrogen is fluid for both model and prototype.

6.3.1.1 <u>Initial fluid temperature</u>. T_0^* can be maintained at unity simply by providing the same propellant supply temperature for model and prototype tank resupply.

6.3.1.2 <u>Equilibrium temperature</u>. The following development will show that equilibrium temperature at any time during the tank fill process will be a function of entering liquid temperature, initial propellant tank stored energy and the percent liquid fill:

From the First Law

$$m_{W} (u_{\infty} - u_{i})_{W} + (u_{L_{\infty}} m_{L}) + (u_{V_{\infty}} m_{i}) - (u_{L} m_{L})_{i} - (u_{V} m_{V})_{i} = h_{L} (m - m_{i})$$

$$m_{I} = m_{I} + m_{I} (at time, \tau)$$

$$(6-40)$$

$$L = v (at the t, t)$$
 (6-41)

$$m_i = (m_L + m_v)_i$$
 (6-42)

where

- u = internal energy h = enthalpy
- m = mass

subscript,

L	=	liquid
v	=	vapor
w	z	wall
i	=	conditions at tank fill initiation
9 0	=	equilibrium property conditions at time $ au$

The solution to equation (6-40) can be readily obtained by recognizing that m_{L_i} , m_{V_i} and $u_{W_{\infty}}$ are either zero or insignificant. Fluid mass at tank fill initiation will be essentially zero for filling an evacuated tank and $u_{W_{\infty}}$ will be near zero throughout much of the tank fill because tank wall temperatures will be at or very near liquid temperature. Thus, combining (6-40), (6-41) and (6-42) results in

$${}^{u}L_{\infty} {}^{m}L {}^{+u}v_{\infty} {}^{m}v {}^{=}h_{L} {}^{(m}L {}^{+m}v) {}^{+m}w {}^{u}w_{i}$$
(6-43)

Since $m_w u_{w_i}$ represents the initial tank stored energy, ΔQ_i , we can make this substitution in (6-43)

$$\mathbf{u}_{\mathbf{L}_{\infty}}\mathbf{L} + \mathbf{u}_{\mathbf{V}_{\infty}}\mathbf{m}_{\mathbf{V}} = \mathbf{h}_{\mathbf{L}}(\mathbf{m}_{\mathbf{L}} + \mathbf{m}_{\mathbf{V}}) + \Delta \mathbf{Q}_{\mathbf{i}}$$
(6-44)

Dividing (6-44) by m_L gives

$$u_{L_{\infty}} + u_{V_{\infty}} (m_{V}/m_{L}) = h_{L} (1 + m_{V}/m_{L}) + \Delta Q_{i}/m_{L}$$
 (6-45)

The left hand side of (6-45) is a function only of equilibrium temperature and vapor to liquid mass ratio, m_V/m_L . The right hand side of the equation is a function of entering fluid temperature ($h_L = f(T_0)$), $\Delta Q_i/m_L$, and m_V/m_L .

It is concluded from (6-45) that $T_{\infty} = f(T_0 \text{ and } \Delta Q_i/m_L)$ for a given m_v/m_L . Therefore, $T_{\infty} * = 1$ when $T_0^* = (\Delta Q_i/m_L)^* = (m_v/m_L)^* = 1$. As stated previously T_0^* can be selected by controlling supply temperature. $(\Delta Q_i/m_L)^*$ can be selected by varying model tank mass and/or initial tank temperature. The variable $(m_v/m_L)^*$ imposes no restraint other than to stipulate the obvious, which is that a comparison of model to prototype tank fill behavior is applicable only at the same tank fill condition. However, this requirement does provide the following relationship between time and tank scale

$$\mathbf{w}^{*} = \mathbf{m}^{*} = \mathbf{m}^{*} = \mathbf{1} = (\mathbf{m}_{m}/\mathbf{m}_{p}) = (\rho V_{T})_{m}/(\rho V_{T})_{p} = V_{T}^{*} = L^{*3}$$
(6-46)

where

V_T = tank volume L = characteristics tank dimension

$$\rho$$
 = fluid density

$$\rho_{\rm m} = \rho_{\rm p}$$
 (same fluid)

Time can be introduced by recognizing that

$$m_{L} = m\tau \text{ or } \tau = m_{L}/\dot{m}$$

where

 $\dot{\mathbf{m}}$ = entering flow rate τ = flow duration

 m_{I} = liquid mass in roduced to tank during τ

Dividing model variables by prototype variables gives

$$\tau * = \mathbf{m}_{\mathbf{L}} * / \dot{\mathbf{m}} * \tag{6-48}$$

Substituting (6-46) into (6-48) results in

$$\tau * = L *^3 / \dot{m} *$$
 (6-48a)

6.3.1.3 <u>Dimensionless time parameter</u>. It is now necessary to identify conditions under which $\theta^* = 1$.

All variables of the dimensionless time parameter (θ), with the exception of h and A, can be described. Equation 3-11, Reference 3-3, relates h to fluid properties and tank inflow conditions. This relationship is given as

$$\frac{h}{\rho_{\rm Cp}} \left(N_{\rm PR}\right)^{2/3} = C_1 \left[\frac{(\dot{\rm mv}^2/V_{\rm L})^{\mu}}{\rho^2} \right]^{1/4}$$
(6-49)

where all variables have been previously identified except:

$$V_{L}$$
 = fluid volume

Normalizing equation 6-49 provides the following expression

h* =
$$(\dot{m} * v *^2 / V_L^*)^{1/4}$$
 (6-50)

The total surface area, A, of the dispersed phase is

$$A = n A_{h}$$
(6-51a)

$$n = V_{\rm u}/V_{\rm b} \tag{6-52}$$

$$V_{\rm b} = \pi \, {\rm d}^3/6$$
 (6-53)

$$A_{b} = \pi d^{2} \qquad (6-54)$$

Therefore

А

$$= \frac{6V_{u}}{d}$$
(6-51b)

or

$$A^* = V_{11}^*/d^*$$
 (6-51c)

where

n	=	total number of bubbles immersed in liquid
A _b	=	bubble surface area (assumed spherical)
$v_{\mathbf{v}}$	=	total vapor volume
d	æ	bubble diameter

Now, equation 3-36 can be written as

d =
$$F_1 \epsilon^{-5} (\dot{m}v^2 / V_L)^{0.4} + C$$
 (6-55)

where

$$\epsilon = V_u / V_1$$

$$F_{1} = 1.134 \sigma^{0.6} / \gamma^{0.2} = \text{constant based upon fluid properties}$$

C = empirical constant = 0.09

By assuming that C is insignificant,

$$d^{*} = \epsilon^{*} \cdot \frac{5}{(m^{*}v^{*}/V_{L}^{*})}^{0.40}$$
(6-56)

Also since tank null condition is the same for model and prototype, $V_L^* = V_V^* = V_T^*$

$$d^* = (\dot{m}^* v^{*2} / V_L^*)^{-0.40}$$
(6-57)

This, of course, assumes that the same proportion of vapor is entrained in liquid for both the model and prototype.

If one assumes that C is the dominant term in (6-55) the result is

$$d^* = 1$$
 (6-58)

Both 6-57 and 6-58 will be considered in evaluating experiment modelling require-

Normalizing the dimensionless time parameter, we have

$$\theta^* = h^* A^* \tau^*/m^* (Cp^* = 1 \text{ for same fluid})$$
 (6-59)

Substituting (6-48) into (6-59) gives

$$\theta^* = h^* A^* / \dot{m}^*$$
 (6-60)

Consider first the condition where bubble diameter is primarily influenced by (mv²/V₁). Substituting equations 6-50, 6-51 and 6-57 into 6-60 provides

$$\theta^* = (\dot{m}^* v^{*2} / V_L^*)^{0.25} (V_u^* / \dot{m}^*) / (\dot{m}^* v^{*2} / V_L^*)^{-0.40}$$
(6-61a)

$$= (\dot{\mathbf{m}}^* \mathbf{v}^2 / V_L^*)^{0.65} (V_u^* / \dot{\mathbf{m}}^*)$$
(6-61)

Now, from continuity

<u>^*</u>

$$\dot{\mathbf{m}} = \boldsymbol{\rho}_{\mathbf{A}_{\mathbf{V}}} \tag{6-62}$$

where $A_n = nozzle$ inlet diameter.

Normalizing (6-62) results in

$$\dot{\mathbf{m}}^* = A_{\mathbf{n}}^* \mathbf{v}^*$$
 (where $P^* = 1$ for same fluid) (6-63)

Substituting (6-63) into (6-61) and recognizing that

$$V_{u}^{*} = V_{L}^{*} = L^{*3}, \text{ and } A_{n}^{*} = L^{*2},$$

$$\theta^{*} = (\tilde{m}^{*3}/A_{n}^{*2} V_{L}^{*})^{0.65} (L^{*3}/\tilde{m}^{*})$$

$$= (\tilde{m}^{*3}/L^{*7})^{0.65} (L^{*3}/\tilde{m}^{*}) = (\tilde{m}^{*0.95}/L^{*1.55})$$
(6-64)

By setting $\theta^* = 1$, (6-64) becomes

$$\dot{\mathbf{m}}^* = \mathbf{L}^* \mathbf{1}^* \mathbf{63}$$
 (6-65)

Combining 6-63 with 6-65 and solving for v*, we have

$$\mathbf{v}^{*} = \mathbf{L}^{*} / \mathbf{A}_{n}^{*} = \mathbf{L}^{*} / \mathbf{A}_{n}^{*} = \mathbf{L}^{*} / \mathbf{L}^{*} = \mathbf{L}^{*}$$
(6-66)

Also, combining (6-48a) with (6-65) and solving for $\tau^* = L^{*1.37}$ (6-67)

Equations 6-65, 6-66 and 6-67 relate the primary test variables of flowrate, velocity and time to model tank scale for the condition where bubble diameter, d, is a function of fluid power input (or mechanical power). These relationships will exist at low power input levels. Results are given in Table 6-5.

Consider now the second condition where bubble diameter is independent of $(\dot{m}v^2/V_L)$.

Substituting equations (6-50), (6-51c), and (6-58) into (6-60) gives

$$\theta^* = (\dot{m}^* v^* / V_L^*)^{0.25} (V_u^* / \dot{m}^*)$$
 (6-68)

Substituting (6-63) into (6-68) and recognizing that

$$V_{u}^{*} = V_{L}^{*} = L^{*3}$$
, and $A_{u}^{*} = L^{*2}$

results in

$$\theta^* = (\dot{m}^* / L^*)^{0.25} (L^* / \dot{m}^*) = L^*^{1.25} / \dot{m}^*^{0.25}$$
 (6-69)

By setting $\theta^* = 1$, (6-69) becomes

$$\dot{\mathbf{m}}^* = \mathbf{L}^*$$

Combining (6-63) with (6-70) and solving for v^* we have

$$\mathbf{v}^* = \mathbf{L}^{*5} / \mathbf{L}^{*2} = \mathbf{L}^{*3}$$
 (6-71)

Also, combining (6-49) with (6-70) and solving for *,

$$= L^{*-2}$$
 (6-72)

Results of equations (6-70), (6-71), and (6-72) are given in Table 6-6.

	<u></u>				
	Model Tank Scale, L*				
Model Test Variable	$0.108^{(1)}$	0.2	0.3		
m*	0.0266	0.07	0.14		
v *	2. 278	1.81	1.56		
Γ.*	0.0474	0.11	0, 19		

Table 6-5. Model tank scale influence upon fill tank variables. $(d^* = f(\dot{m}^* v^{*2}/V_T^*))$

(1) Tank model for Spacelab experiment.

Table 6-6. Model tank scale influence upon fill tank variables. $(d^* = 1)$

Model Test Variable	Model Tank Scale, L*		
	0.108 ⁽¹⁾	0.2	0.3
m *	1.47×10^{-5}	0.0003	0.002
v *	1.26×10^{-4}	0.008	0. 027
T *	85.7	25	11.11

(1) Tank model for Spacelab experiment

• Prototype Tank Dimension, $I_p = 4.29m$ (169 inches) (diameter)

θ* = 1

Assumed Test Conditions

- o Same Propellant (LH₂)
- o Same Initial Tank Temperature, 200K, (360R)

The question to be resolved is which set of scaling equations should be satisfied during a model fill experiment: the equations based upon $d^* = 1$, or upon $d^* = f(\dot{m} * v *^2 / V_L)$? The answer depends upon determining the following:

- 1. What are the vapor to liquid heat transfer rates needed to guarantee thermal equilibrium during tank fill?
- 2. What fluid power (or equivalent mixer power) requirements will provide the
- 3. Will propellant tank fill durations be acceptable at the identified fluid power
- 4. Is the calculated bubble diameter a function of or independent of fluid power (or mixer power) over the range of anticipated power input conditions?

First, Figure 6-15 gives the average vapor to liquid heating rate required during tank fill to assure near thermal equilibrium conditions. Note that heating rate requirements decrease as tank fill duration increases. This occurs because the total ullage energy removal requirement is independent of fill duration, and a longer fill duration means that the everage heat removal rate can be decreased. A fill duration of 3 hours was selected as acceptable in section 3-3, and results in a required average heating rate of about 5 kw (> 5 Btu/sec).



Figure 6-15. Average vapor to liquid heating rate needed to achieve thermal equilibrium.



•

Figure 6-16. Mixer Power Influence Upon Entrained Vapor to Liquid Hydrogen Heat Transfer Rate



Figure 6-17. Mixer Power Influence Upon Entrained Vapor to Liquid Oxygen Heat Transfer Rate

achieved for a mixer power input less than 4 watts. Note that input power requirements to achieve a given heating rate are a strong function of vapor hold-up (percent vapor entrained in liquid), and a very weak function of the percent liquid fill condition. According to Figures 6-18 and 6-19, vapor bubbles generated within the OTV propellant tanks by a mixer (or its equivalent in fluid power) are a function of fluid power for power levels less than about 12 watts. Thus it appears that the model test parameters of Table 6-5 should be employed for tank fill experimentation. It is encouraging to note from Table 6-5 that the required model test flowrate and velocity requirement for the Spacelab experiment can be readily achieved.

6.3.1.4 <u>Mixer power/fluid power relationship</u>. Additional information applicable to orbital refill is presented in Figures 6-20 and 6-21. These curves show the equivalence between fluid power and mixer power, but using variables of fill duration and transfer line nozzle inlet diameter to describe fluid power entering the propellant tanks during fill. Equivalence was established in the following manner:

Fluid Power =
$$\dot{m} v^2 = \dot{m}^3 \rho_L^2 A_n^2 = C_1 (\dot{m}^3 / \rho_L^2 D_n^4)$$
 (6-72)

where

 $D_{n} = nozzle diameter$ $\dot{m} = tanking flowrate$ v = inlet velocity $\rho_{L} = propellant density$ $C_{1} = constant$

But

m

$$= \mathbf{m}_{\mathrm{T}}^{\prime} \tau = \mathbf{\rho}_{\mathrm{L}} \mathbf{V}_{\mathrm{T}}^{\prime} \tau \tag{6-73}$$

Therefore

$$\dot{m}v^2 = C_1 \rho_L / D_n^4 (V_T / \tau)^3$$
(6-74)

where

 m_{T} = tanked propellant mass V_{T} = propellant tank volume τ = tanking duration









6-41



Figure 6-20. Fluid Power Input Equivalence to Mixer Power During OTV LH2 Tank Fill



Figure 6-21. Fluid Power Input Equivalence to Mixer Power During OTV LO₂ Tank Fill

Finally, we can show that

Fluid Power = Mixer Power Output

= Efficiency × Mixer Power Input

Note that fluid power requirements of greater than 4 watts can be achieved with a reasonable selection of fill duration and nozzle diameter.

6.3.1.5 <u>Model tank V*/M* influence</u>. The inability to exactly scale POTV on the bases of tank volume and tank mass will have a significant influence on tank fill, as it will have upon the prechill process. Higher model tank pressures will occur during the initial fill transient than for the prototype LH₂ tank, as previously discussed and indicated by Figure 6-7. This peak pressure could be excessive, especially for the 0.108 scale tank which has V*/M* = 0.105. Equally significant is the fact that equilibrium pressures at the end of tank fill can also be heavily influenced by V*/M*. Figure 6-22 gives final tank pressure as a function of entering LH₂ vapor pressure and V*/M*. It is seen that final tank pressure could increase by about 69 kN/m² (10 psid) greater at a V*/M* of 0.108 than for the prototype tank. Conversely, if the goal is to achieve a given final pressure, the model test tank will require an entering LH₂ vapor pressure that is about 69 kN/m² (10 psid) lower than for the prototype tank.



Figure 6-22. V*/M* influence upon POTV LH₂ tank pressure following thermal equilibrium fill process.
3.3.2 FLUID SUBSTITUTE. A determination was made in Section 6.3.1 that a propellant tank fill process can be simulated if the model-to-prototype dimensionless time parameter, θ^* , can be established as unity for experimentation. Model test parameters of flowrate, velocity and time scaling are identified in Table 6-5 which establish $\theta^* = 1$ for model tests conducted with liquid hydrogen. The purpose of this analysis is to quantify these same test parameters for the case where liquid nitrogen is substituted for liquid hydrogen. The following equations employed in Section 6.3.1 are also applicable to this analysis:

$$\theta = (hA/mCp) \tau$$

$$\frac{h}{\rho Cp} (N_{PR})^{2/3} = C_1 \left[\frac{(\dot{m}v^2/V_L)}{\rho^2} \right]^{1/4}$$
(6-10)
(6-10)
(6-49)

Normalizing equation (6-10) by taking the ratio of model-to-prototype conditions results in,

$$\theta * = (h * A * / m * Cp *) * = (h * A * / m * Cp *)$$
 (6-73)

-01

where:

$$\dot{m}^* = m^* / \tau^*$$

and superscript (*) = the ratio of model to prototype (in this case, LN_2 to LH_2 conditions for same tank scale)

Normalizing equation (6-49) and solving for h* gives,

h* =
$$(\dot{m} * v *^2 / V_L)^{25} \mu *^{25} \rho *^{5} Cp * / (N_{PR} *)^{2/3}$$
 (6-74)

$$= (\dot{m}^* v^*)^{\cdot 25} \mu^{\cdot 25} \mu^{\cdot 5} C p^* / (N_{PR}^*)^{2/3}$$
(6-75)

where:

$$V_{L}^{*} = 1$$
 (for same tank scale)

Now, combining (6-73) and (6-75) results in

$$\theta^* = (A^*/\dot{m}^*) (\dot{m}^* v^*)^{25} \mu^{*25} \mu^{*5} / (N_{PR}^*)^{2/3}$$
 (6-76)

= 4.43
$$(\Lambda^*/\dot{m}^*)$$
 $(\dot{m}^*v^*)^{25}$ (6-77)

where

$$\mu^{*} {}^{25} = (11.85)^{\cdot 25} = 1.86$$

$$\rho^{*} {}^{5} = (11.4)^{\cdot 5} = 3.38$$

$$(N_{PR}^{*})^{2/3} = (1.7)^{2/3} = 1.42$$

The normalized total surface area, A^* , of the dispersed phase is $A^* = Vu^*/d^*$ (6-51c) Now, the expression for bubble diameter, d, was found to be

d =
$$F_1 \epsilon \frac{5}{(mv^2/V_L)^{0.4}} + C$$
 (6-55)

,

where:

$$\epsilon = V_u / V_T$$

$$F_1 = \text{constant based upon fluid properties}$$

$$V_T = \text{tank volume}$$

$$C = \text{empirical constant}$$

It was determined in Section 6.3.1.3 that the expression for d can be simplified by eliminating C. Consequently, we have

d =
$$F_1 \epsilon^{-5} (\dot{m}v^2 / V_L)^{0.4}$$
 (6-78)

Normalizing equation (6-78) and recognizing that $V_L^* = V_u^* = V_T^*$,

$$\epsilon^{*} = 1 \text{ and},$$

$$d^{*} = F_{1}^{*} / (\dot{m}^{*} v^{*})^{0.4}$$
(6-79)
Now $F_{1}^{*} = .\sigma^{*} \cdot {}^{6} / \rho^{*} = 1.26$
where: σ = liquid vapor surface tension
and $\sigma^{*} \cdot {}^{6} = (3.3)^{\cdot 6} = 2.05$
and $\rho^{*} \cdot {}^{2} = (11.4)^{\cdot 2} = 1.63$

Combining equations (6-51c) and (6-79) and recognizing that

$$V_u^* = 1$$
 we have

A* =
$$1/d* = (\dot{m}*v*^2)^{0.4}/1.26$$
 (6-80)

Substituting equation (6-80) into (6-77),

$$\theta^* = 3.52 (\dot{m}^* v^*)^{.65} / \dot{m}^*$$
 (6-81)

Introducing the continuity equation, $\dot{m} = \rho Av$, and normalizing, we have

$$\mathbf{m}^* = \rho^* \mathbf{v}^* \mathbf{A}^*_{\text{nozzle}} \text{ where } \mathbf{A}^*_{\text{nozzle}} = 1.0$$
 (6-82)

Substituting (6-82) into (6-81) and solving for \dot{m}^* ,

$$\dot{m}^* = .266\rho * {}^{1.368} \theta * {}^{1.053} = 7.42 \theta * {}^{1.053}$$
 (6-83)

Finally, in order to satisfy the requirement for $\theta^* = 1$, we find from equation that

$$m^* = 7.42$$
 (6-84)

$$v^* = .651$$
 (6-85)

$$t^{+} = 1.536$$
 (6-86)

Equations (6-84), (6-85), and (6-86) represent the ratio of nitrogen-to-hydrogen test variables. These factors were applied to the previously determined liquid hydrogen model test variables (of Table 6-5) to arrive at the data given in Table 6-7. An inspection of this data indicates that there appears to be no limitations nor disadvantages to conducting model tests with LN_2 , other than the concern that the fluid properties difference could result in a non-scaleable heat and mass exchange difference between the propellants.

6.3.3 ONE-G TEST ENVIRONMENT LIMITATIONS. Ground-based tests for determining the validity of the tank fill process will be applicable only if a) the normal gravity free convection heat transfer process is dominated by the forced convection mechanism created by the entering propellant, and b) the fluid flow mechanism is the same for model tests as for the prototype configuration. To satisfy the former condition we must verify that the heat transfer coefficient given by equation (6-9) exceeds that free convection coefficient created by a normal gravity environment. Equation (6-9) is applicable to heat exchange between a propellant tank surface and its contained vapor, and to heat exchange between a liquid and entrained vapor bubbles. The latter condition applies to the similarity of fluid flow regimes such as laminar or turbulent flow in a flow inertia

	Model Ta	nk Scale, L*	·····
Model Test Variable	. 108 ⁽¹⁾	. 2	. 3
m*	.15	. 52	1.04
v*	1.52	1.18	1.02
au *	. 06	.17	. 29

Table 6-7. Model tank scale and fluid substitute influence upon fill test variables $d^* = f(\mathbf{m}^* \mathbf{v}^* / V_L^*)$, for LN₂.

(1) Tank model selected for Spacelab experiment.

- Reference data is from Table 6–5.
- Test fluid is LN₂

dominated environment.

It is expected that propellant tank fill in space will cause the propellant and its vapor to be intimately mixed due to the absence of gravity. Thus the fluid flow mechanism will be that of an inertia dominated process. Furthermore, this intimate mixture of liquid and vapor should serve to provide near-thermal equilibrium conditions during fill.

This same uniform mixing of liquid and vapor will not be possible for tests conducted in a normal gravity environment because gravity will tend to maintain the liquid phase separated from vapor. Analyses have indicated that incoming liquid velocities, for a 0.108 scale model tank, may have to be increased by a factor of five to ten greater than scaling would indicate just to provide liquid-vapor mixing. A velocity increase of this magnitude could invalidate the experiment because of a greatly increased vapor bubb'e-to-liquid heat exchange. As with the prechill process, it is questionable that useful data can be obtained for predicting the full-scale process.

6.3.4 START BASKET REFILL TEST CONSIDERATIONS. Refill of the propellant start baskets is an important requirement during the tank fill process. The start baskets must be free of vapor prior to first main start so that 100 percent liquid flow to the OTV main engines can be assured. It was determined in the analysis of Section 3.3.6 that an unknown quantity of propellant vapor may reside within these screen devices at tank fill completion. An extremely conservative assumption was made that the largest spherical bubble possible would reside within each device. Collapse times of about five minutes and four hours were calculated for the oxygen and hydrogen bubbles, respectively. It is obvious from the predicted collapse times that oxygen start basket refill will be assured. The long time to collapse hydrogen bubbles, however, resulted in a recommendation to actively assist hydrogen start basket refill. It was proposed that a small diameter line be routed from the tank fill port to the basket so that liquid would be sprayed into the volume during the fill process. Calculations, for a conservative fill model, indicated that all entrapped vapor would be condensed before tank fill completion.

The mechanism for removing vapor from the start basket is identical to that described for tank fill. That is, the incoming liquid momentum will serve the dual function of creating small bubbles and a high heat exchange environment between liquid and vapor, both of which enhance condensation. The strategy to assure start basket refill will be to create sufficient agitation within the screen volume to assure complete vapor condensation. It appears that there will be fewer restraints imposed upon start basket refill than tank refill, consequently, a propellant flow split can be selected to assure basket refill before tank fill completion. As an extreme example, tank fill could be performed by flowing all propellant through the screen device.

Since it appears that basket refill can be assured, and because tank fill experiments will also be useful in assessing condensation conditions within the start basket, a detailed test program is not recommended. Rather, it may be sufficient to provide a range of start basket flowrates as part of the tank fill experiment. There would be an advantage to performing several tank fill experiments without flowing propellant to the start basket. These tests would identify if an active means of start basket refill is necessary.

6.4 SPACELAB EXPERIMENT INTEGRATION

The Spacelab provides facilities for investigating fluid behavior in a low gravity environment. The experiment payload accommodated within the pressurized Spacelab, however, must adhere to strict safety regulations regarding the crew, the mission, and the payloads. Specifically, Reference 6-1 restricts the use of cryogens to the exterior of the Spacelab.

To confirm the Reference 6-1 guidelines, Merle Slayden, a NASA/MSFC Safety Engineer was contacted to discuss current directives regarding the use of cryogens. The proposed receiver tank experiment, employing a 457 mm (18 inch) diameter tank mounted within a double rack, was discussed and he stated such LN_2 quantities would require a waiver for Spacelab. Unfortunately a waiver can only be obtained during the experiment integration safety review period, which occurs after payload design and fabrication. As a guideline, Mr. Slayden offered the information that LN_2 quantities of about 3.6 kg (8 lb) would be acceptable. The proposed test tank would contain a maximum of 120 kg (264 lb) LN_2 . Liquid hydrogen is unacceptable under any condition.

There will be additional restraints imposed upon the receiver tank experiment design other than the use of LN_2 . These Spacelab general experiment accommodation factors are given in Appendix B and have been extracted from the "Life Sciences Guide to the Space Shuttle and Spacelab'' experimenters handbook, and the Reference 6-1 handbook

6.5 MODELLING OF LTL REFUELING OPERATIONS

A mission scenario was selected in Section 5 for the purpose of analyzing orbital refill operations. A low thrust liquid (LTL) vehicle configuration was selected from existing vehicle hardware; principally, the OMS tanks were selected as the main propellant tanks, and the shuttle RCS bottles were selected for the same function on the LTL. N_2O_4 and MMH were the propellants selected for the main propulsion and RCS systems. Each of the storage tanks and bottles contained a screen acquisition device.

The analysis of Section 5 identified the refill requirements listed in Table 6-8 . It was determined that all potential problems, such as liquid venting and helium entering the screen devices, resulted from the need to expel helium from the storage vessels before initiating refill. Furthermore, it was found that refill could be performed in a very straightforward manner; venting would not be required, and tank pressures would not exceed normal operating pressures during refill.

Table 6-8 . LTL vehicle refueling requirements.

- Propellant Tank Vonting Must Not Damage LTL or Orbiter
 Liquid Venting Potentially More Damaging Than Vapor Venting
- Helium Must Not Enter Partial Screen Device Galleries
- Vehicle Control Must Be Maintained During Refill Operations
 Liquid Venting is Undesirable
- Simplified Operations Are a Necessity
- Must Have Ability to Refill to $\sim 95\%$ Level

It was concluded from the refill analysis that virtually all potential problems would be eliminated through the use of procedures, if certain vehicle modifications were incorporated. These modifications included adding plumbing between storage vessels (see Figure 5-5) to enable propellant transfer between common tanks. With these changes, procedures were established so that propellants could be transferred from one tank to another before initiating a tank vent.

The single concern of the selected refueling procedure was that propellant contained within the screen devices might boil during tank vent to expel helium. The solution would be to maintain sufficient liquid residual that it would replenish any screen device liquid lost by evaporation. This solution will be effective only if correlation is maintained between liquid pool and screen device. Thus, the purpose of all orbital experiment would be to verify that communication is maintained during the tank vent process, and this will be a function of geometry (tank and screen) and acceleration environment.

Laboratory experiments conducted in a normal gravity environment will not be useful because a near zero-g environment is required. The appropriate acceleration environment can be provided by orbital experiments.

Orbital experiments to verify that the screen devices will remain filled with liquid are not recommended. Such tests would not have wide applicability because results would be very sensitive to geometry. It is recommended that experiments of this type not be conducted until an operational vehicle configuration can be established.

7

REFERENCES

- 1-1 Merino, F., et al, "Filling of Orbital Fluid Management Systems," NASA-CR-159404, CASD-NAS-78-010, July 1978.
- 1-2 Blatt, M. H., Bradshaw, et al, "Orbital Transfer of Cryogenic Propellants," CASD-ERR-77-093, December 1977.
- 1-3 Merino, F., Thies, et al. "Orbital Transfer of Cryogenic Propellants," CASD-ERR-78-088, December 1978.
- 3-1 "Orbital Propellant Handling and Storage Systems Definition Study," GDC-ASP- 002, August 1979.
- 3-2 Blatt, M.H., Risberg, J. A. "Study of Liquid and Vapor Flow Into a Centaur Capillary Device," NASA CR-159657, GDC-NAS-79-001, September 1979.
- **3-3** Merino, F., et al, "Filling of Orbital Fluid Management Systems," NASA-CR-159404, CASD-NAS-78-010, July 1978.
- 3-4 Uhl, V. W. and Gray, J. B., "Mixing, Theory and Practice, " Volume II, Academic Press, 1967.
- 4-1 "Orbital Propellant Handling and Storage Systems for Large Space Programs," JSC-13967, CASD-ASP-78-001, April 1978.
- 4-2 "Orbital Propellant Handling and Storage System Definition Study," GDC-ASP-79-002, August 1979.
- 6-1 "Payload Accomodations Handbook, " paragraph 8.3.9, Cryogenic Storage, SLP/2104, 30 June 1977.

APPENDIX A

 $\mathcal{L}_{\mathbf{k}}$

IDENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

- TABLE A-1. GENERAL SUMMARY (1975-1990)
- TABLE A-2. ORBITAL TRANSFER VEHICLE (OTV'S)
- TABLE A-3. SPACE PLATFORMS/SPACE STATIONS
- TABLE A-4. AUTOMATED AND MANNED SPACECRAFT
- TABLE A-5. REFERENCES

100 C

1

L

TABLE A-1. IDENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

Page 1 ol Iment/Iteferenc (+3) each (+3) each I. NASA CR 13: I. I. NASA CR 13: I. NASA CR 13: I. I. NASA CR 13: I. I. NASA CR 13: I. I. I	R 134911, Ref.
76-014 76-014 76-014 76-014 7 2 Tanks 7 able A- 7 Ref. 1 7 able A- 8 7 able A- 8 7 able A- 7 1568 8 Ref. 2. 38 7 0 5 4 0 14 1 1 1 25 22 22 - 71. 7 1 34911, Re 1 2. 5 20 20 20 20 20 20 20 20 20 20 20 20 20	ыв А-6, С
	L I
	* *
ERALS ERALS Cryo Cryo Cryo Cryo Cryo Cryo Cryo Cryo	Avg Avg Avg Avg
GENI GENI GENI GENI GENI GENI GENI GENI	
	<u> </u>
	ext x x
	it, (e) =
	(1-3) (-3) (-3) (-3) (-3) (-3) (-3) (-3) (
	+
Press Press Marked Marked	Med Med
	1427 1289 2551 2551
	× × ×
	× ×
Fluid Fluid	9662 4853 (16700) 121 (267) (267)
N204 N204 MMH MMM MMH MMH MMH </td <td>2 2 7</td>	2 2 7
Function Bary H. L. L H. G. C.S. C.S. C.S. C.S. C.S. C.S. C.S.	Aerozh 50 V2H V2H Hal Ende
BFea BFea BFea BFea BFea BFea BFea Codd MPS MPS MPS MPS MPS MPS MPS MPS MPS MPS	ACS
ndidate biole celver e) 56000LJ 5570LB 25 LBS 25 LBS 25 LBS 25 LBS 25 LBS 25 LBS 25 LBS 26 Centaur Centaur	2 ea. ngine coa2:1 Tank + 1
Vertice STS CAMINER CA STS C	anstage 20 lba E P 306 st adrical
	6 21r 6 60 60

.

TABLE A-1. DENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

í. M

(i) = inlet, (c) = exdt

.

----274 MAG

Fluid Size Tank Function Function Type Function Type Renarby Construction MPS Lo2 118600 X Anidizer Lo2 118600 X Anidizer Lo2 118600 X Anidizer Lo2 19700 X X Sphere Conditizer Lo2 Conditizer Lo2 Corold X Adoo X DropTank LH2 43400	Fluid Size Geometry Inction Size Size Geometry 38 LO2 118600 x Large 38 LO2 118600 x Large 38 LO2 118600 x Large 38 LO2 118600 x Smail 38 LO2 19700 x x 300 x x y x 1 LO2 19700 x x 1 LO2 19700 x y 1 MPS (574300) x y	The function The function 112 1118600 Kg (lb) 113600 Kg (lb) Spinetre 113700 X X 113700 X Spinetre 113700 X X 113700 X Spinetre 113700 X X 113700 X X	3 3 </th <th>All arge control of the second second</th> <th>Ilams Ilams Abhere A</th> <th>Alberta Strategy Stra</th> <th>19bnury2 ~ 6 β 19bnury2 ~ 6</th> <th>E Diorof</th> <th></th> <th>Oper 172 186 KN/m²</th> <th>S Caregory c at</th> <th>ku/m² V cai</th> <th>2 4108912</th> <th></th> <th>× × × ×</th> <th>Capillary</th> <th>S C Bladder S</th> <th>Г<mark>Б</mark>Т 111 X 3 03 06</th> <th>C 17 0 Category C</th> <th>K G</th> <th>Category</th> <th>20 20 30 30 40 10 10 10 10 10 10 10 10 10 10 10 10 10</th> <th>× <100 kg/ 6</th> <th>× × × ×</th> <th>Page 3 of 4 Comment/Refurence Tuble A-12, CR-134911, CASD-NAS76-014, Low-G Fluid Transfer Technology Study, 1976, Ref. 1</th>	All arge control of the second	Ilams Ilams Abhere A	Alberta Strategy Stra	19bnury2 ~ 6 β 19bnury2 ~ 6	E Diorof		Oper 172 186 KN/m ²	S Caregory c at	ku/m ² V cai	2 4108912		× × × ×	Capillary	S C Bladder S	Г <mark>Б</mark> Т 111 X 3 03 06	C 17 0 Category C	K G	Category	20 20 30 30 40 10 10 10 10 10 10 10 10 10 10 10 10 10	× <100 kg/ 6	× × × ×	Page 3 of 4 Comment/Refurence Tuble A-12, CR-134911, CASD-NAS76-014, Low-G Fluid Transfer Technology Study, 1976, Ref. 1
lar Spé	Ice RCS	SFuel	હુ	(95700) 275		- <u>*</u>	·			0685 1	ligh		0-0.	<u>, </u>	×	<u> </u>	50 20	20	Crye			112.8		×	
Space	RCS File HIT	hrust N	3H2 12H4	(606) 34 (76) 3190			× ×		<u>N</u> N	062911	figh livh		0-0.1	· · ·			< ×			111 <	Avg		× ×		Table A-15, CR-134911, Ref. 1, SD-71-214-4, Ref. 7.
X-Ray	Low 5 Crb A	Thrust li Maint N	0 ⁻¹	(7000) 204 (450)	× :	×			<u>-</u> ++				0-0.2	77		××	× ×	278 <t 312 74-372</t 	Avg Avg				× ×		Table A-16, CR 134911, MIC G0534, Vol. 1, Bcok 4, page 118-122; Rof. 1, Ref. 8.
-A)	Prop RCS	ellant N	2017 10 10 10 10	50)×4 100 (220)	× ×	××					ow igh		0-0.1	<pre>~ 1 ~ 1</pre>		×	× ×	78 <t 312</t 	A VE	63-	Vide		× ×		NASA Payload Descriptions Vol. I, p. 2-109, July, 1975, Table A-12, CD194011, p. 6
High I atory tic A	Energy Urb J B Prope RCS	Maint N ellant N	2H4 2 (2)	50) × 4 100 (220)	××	××					ow igh		0-0.1 0-0.1	₽		×	× 10	78 < T 12	Avg 2	303 II 63- W 103 IR	flange /ide		×		9. Ref. 1. CAUSAS11; Ref. 9. Ref. 1. VALSAS11; Ref. 1. Payload Descriptions, Vol. 1, 2. 2-109, July 1975, Table 3. 2-109, July 1975, Table 4-17, CR 134911, Ref. 9,
nchron rm Ob - 2500	ous RCS serv. 0 Lbs		2H2	80 (176)	×	×		++	++		MO	++	0-0.1	7		×	57	8 < T 112	29		P		×	- F	
Explo Prope	rer RCS rer RCS	+	H. 6	175 (385) 68600 x 15(000) x	×	× ×				<u>ت</u> •	, M		0-1 0-0.2	v 1		×	x 0	8 < T A	Nr.	┼──┼─			× ,	ZŽĘ	ASA Payload Descriptions, ol. ī, Automated Payloads, 19, 75, p. 4-6, Ref. 9.
	RCS		(31)	× 2 × 2		×		+		E L	2		0-0.1				×		ıyo		-				aute A-22, F.A-14, CK 34911, CASD-NAS76-014, ay '76;S-II Orb. Prop. Study, 07-554. 3-31-72, Ref 1, Bef 10,

GENERAL SUMMARY (1975-1990) Per CASD-NAS76-014

TABLE A-1. IDENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

A-4

(i) = inlet, (e) = exdt

.

Page 4 of 4 Table A-23, p. A-15, CR-134911, Ref. 1 Comnent/Reference GENERAL SUMMARY (1975-1990) Per CASD-NAS76-014 یں 100 نور > 100 نور ^ 100 نور < 100 نور × Flow Rate kt∖m²-8ec Category Gas Temps rature Я Category. Avg I.iquid 278 < T <312 Ж (f) = inlet, (e) = e_{X} it Acquisition System Other Blader × Viellign Acceleration Category. - 1 A. cel. 0-0.5 × Vent Pressurc Calekon _a‱∕w≀ Operating Pressure Category. Med 1724 z^{uu}/NA TABLE A-1. IDENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS DioTo T Geometry Ellipse 12000122 aranda × llam2 × Size mubeli Juri kt (ID) Grentty 106 (234) . Fluid N2H4 $\mathsf{sd}\mathcal{X}_L$ Tank Function x* = Cylindrical Tank + Eilipsoidal Ends MPS + HCS Candidate Vehicke Keceiver Space Tax Identification No. 19

۰.

•

					j ż		<u> </u>			<u> </u>	perati	त्रिः > ∽	cut		Verel.	AC	sinp	ition	- I	Loder	ature	1	E E	10			
ov noteothtrobl	Candidate Vrhicle Receiver	T ank F unction	Ty pe	אל (נף) אישטוננג.	LAIRe	Medium	insme	1 apunta 5	289013	Toroid	c ^{uu} .NM	Saudare 7	- ¹¹¹ / NN	Lionary	к	Veceleration 2	Capillary	Other Bladder	ĸ	ດສະເຮັດນີ້.	́ Я	CHIERCER	kt√m ² -∎ec	/2010015	> 100 141	Conment/Heference	
52	Teleoperator (5 ca 25 LB Thrusters/ 1 ank, 9 Triads X3-23 (old Gas Thrusters	0AS+RC5 MCS	₩2H ₽H2X	6500)X2 (1500)X2 680 1500)X4 45		<u>├</u> }	<u>├</u> }	<u>* *</u>						0-0	v V	-		×	276 < T < 312	Avg		Ave	7.5	× ×		Teleoperator fletrieval System (THS) Proj. Overview to Aero. + Sjace Eigr'g Board (ASEB), by MSFC, 1-12-75, Rel. 11.	1
		c l	7.	1922.24		*	×						+					<		į	< 320	8		<	_		
17	Initial Strigle Stage OTV (545.), Unman- nect, (Also - AFOTV One 15000 LBF Engl	MIS Oxid. IMPS Fuel	L02 LH2	21550 (47506) 3950 (6797)	<u>* *</u>	<u> </u>		- 2	*		.)				·		× ×		56 50	o tryo			< 12.4 <2.3	××		Veh. Data Book, 3-20-10, Boeing D180-20242-3, p. 71, NASA TACX 73394, p. 15-16, A-77, MSFC, p. 37-41, NAS9-	
		RCS	N2H4	98 [] 185[]		×	×			22	N N				v			×	276 < T <312	Avg						230/0 0004, 1-74, Hel 12, Kel 13 (Ref 14), Ref 15, Ref 16,	
N 23	Common Stage OI V (117), /1179, Main- ned dw/fV), 1st Stage 4 ea 2000	MFS Oxid MFS Fuel	5 1 1 1 1	4 4430 (0794u) X2 5255 (18255) X2	* *				ĸ						<u>,</u>		× ×		0 0 5 2	Crye Crye			67. 6 33.5 5.6 6.6	ف	×	Convart Veh. Juta Bork, 4-20-77, Convart CASD-ASD- 79-001, uSC-15907, p. 7-10, -17,-20; JSC-11568, p. VI-E-	
• - •	L.BF. Erg., 2nd Stuge 2 ra 2000 L.F. Erg.	нсs	N2H4	401 (050)		×								I-0	V				276 < 7 < 312							21, 22, F-6, 5-31-76, Ref. 12, Ref. 17, Ref. 2.	
	Medular Controli Stage Of V 2 Stages rea 2 LH2 tailes, 11 to tech	MIS Oxid. MIS Fiel	1762 1.H.2	67300 126400 5340 41750,44	× ×			* *							~ ~				06 07	Crys. Crys			50.6 25.3 8.4 4.2	(a) ×	×	Vehnele Data Book, 4-26-77; Lavus H. P., Solar Power Satellite Demo. Prog. JSC Memo, 3-24-77, Ref. 12, Ref.	
	(trianned)	RCS	×2H4		+	*	×							5					276 < T < 312	AVE						19. 1st Stage 4 ea 15000 1.B ergine, 2nd Stage 2 ea 15000 1.B engine,	
	[0.1V Connor.] Stage (170E/170E), (Marned), 1st Stage 4 va 20006 135	MPS Oxid. MPS Fuel	L/0 ₂ LH ₂	65500 (143310) 12100 (26760)	×			×	×			·		1-0	×		× ×		50 50	С 1 уе С 1 уе			67.6 33.5 11.2 5.6	× ^(t)		Vehicle Lata Book, 4-26-77, Ref. 12.	
	enge, 2nd Stage 2 ea 2000 LBF eng.	RCS	H2N	455 (1022) (101 (101 (101) (101)		ļ	×								\		-	××	278 < 7 312	Avg							
	ylmanical Tank With	Ellipsortal	Bulkhead	ds	-	-	-	1]		-		1	4		-	1	-	0	- tule	ت (= 6X1	-				1

.

RHITAT

TAELE A-2. IDENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

Í.

Ŕ.

ORBITAL TRANSFER VEHICLES (OTV

Š,

L			ļ	ſ		$\left \right $																
.0			ũ.	luid	Site		- more		Operation	× S	nt			Acqui	isition	Ē	emperatu	2	-			Page 2 of 5
v UO1						Ĩ			Pressur	5 0	casure	Ac	cel.	Syste		Liqu		Gas		w Rate		
I TRUCTE	Candidate Vehicle Receiver	T ank Function	aq Y T	Re (ID) Cumutty	Medium	Sphere	Cylinder	Torold	Sategory	rn/ms	Crogets.		Crossie.	cceleration apillan	Ladder Ladder		(108e)		/ح روق دیک	251 00	/201 00 201 - 201 - 201	3 3 ग
	OTV - 1-1/2Stage	MPS	102(4)	85550	-	1-	+ *	+	' (×		8	э.	v	a a	ĸ	c	ж		01>	- 10 - 10	comment/Reference
-	(50A/220K/220K), 5 LH2, 5 LO2	.ppro	LH2(4)	(188500)×4 14250 ₍₎	×							ī	-			<u>3</u>	Cryo		67.			(4 Init. Stage LO ₂ Tanks, 1
	Tanks, (Geo Assy) 4 en. 20000 I BY	Inter I		31420)×4					_							20	Cryo		н. П	•		LH2, 1 Final Stage LO2; 4 Init. Stage LH2, 1 Final Stage LH.).
	Engine	Contd.	2	32500	×		×					1-0	7	-		8	CIVO	+		-	\square	Vehicle Data Book, 4-26-77;
		MPS Fuel	LH2	6060 (13360)	×		*		·							20						D180-20242-4, p. 259; D180- 20242-3, p. 79-83; JSC 11568,
		RCS	N2H4			-	+	+			Ţ	1-0	1	-		1		+		×		D180-2024202, p. 522, SPS Concepts, Mid-Term Briefing
1	OTV-475T (520K/	MPS	ą	196800		1	-		-				 		_	<312						Boeing 6-7-76; Ref. 12, Ref.
_	520K) (1st + 2nd	Oxdd.	<u>ت</u> ۱	138300)×2			×				_		~	×		90	Cryo		185	5	() ()	20, hel. 14, kef. 21, Ref. 2.
	Stage Tanks Stmilar +1/2 Stage)	MPS	LH ₂	37060 (x	×		•							×		06)	Similar to item 33. Also Ref.
	Stage 1: 11 ea.	1/2 Stage	1-	221001 10	+		Ŧ	+	-				-			04			< 30.	× 8		JSC 11568, Vol. II, p. VI-D-
_	20K LBF Eng. Stare 2: 8 ca	ST N	ro ²		×	×							2	,			+-	-			Ī	2-5; Ref. 12, Ref. 2. Stage 1 of 11 Engines. Stage 2 of 8
-	20K LBF Eng.	WPS.	LH										•	<		20	0 1 1 1 1 1 1 1 1		<67.6	×		Engines, 20000 lbs/each.
	Similar to #33	Fuel				×				_				×		50	Cryo	_	<11.2	e × ®		
		S L	N2H₄	·	×	×						17	1	<u>†</u>	+	278 < T	1 VE	+		Ţ		
_	Lunar Transfer Vehicle (1.74)	Drop Tank	201	15100	×		×	170		T	-		'+- -	+	+	312	, 			_		
-	(32K/40K), 2 en		·H.	2820	,								« 			2 2	0		42.26	e ×		Vehicle Data Book, 4-26-77;
11	20000 LBF Engines			6220)		·	_	140			_		*			20 Ci	06		7.54	× ®		Boeing D180-2024-4, p. 227 Londing Reserved to be 22
		Oxid (2)	02 [13	6100 14401×2	×	×		170				2	×	+	+	0 8	04		42 26	6		2000 kg/m ² /sec) for LO ₂ and
		Fuel (2) 1	.H2	1140	×	×		140					×			20						00 kg/m ² /sec for 1 H ₂ . NAS8- 19676, p. 227; Ref. 12, Ref.
	L	A PS(4) N	12H4		×	×	+	+		+	6		┯			t e	2	_	•c·,	×		0, Ref. 16.
	Mot Plant -	MPS Oxid L		540	+	+				+		\rightarrow	-+	•		 312 						
0	Chemical OTV,		10	00)×20		<u>k</u>					6	~ 	-	×		90 Cr	yo		÷16.9 @	×	3	ehicle Data Book, 4-26-77.
	LTL), 20 ea 1000	Mro Fuel L	H2 181	850 T0)×20	×	×								×		20 Cr			<2.8	×	ũ ở	uture Space Transportation
	b Engines, Pay-	RCS	2H4		×	×	+			+-												Tiefing NAS9-14323, Dec. '76,
I .	- Cylindrical Tank	+ Ellipsoidal	Ends			-				-		_		-	·	312	be					180-20242-4, p. 165, Ref. 12

(1) = inlet, (e) = extt

1

<u>ا</u> "	ABLE A-2. IDENTI	PICATION 6	F CANE	IDATE VE	E I	-		EIVI	I H3	ANNA									GRI	BITAL 1	RANSI	ER VE	HICLES	CTV.	â		Page 3 of 5	
											per-ofi	ing v	ent			~	cqui	ulle	4	c III	neratui	IJ				┝		_
.0N			Ē	pin	<u>s</u>	<u>*</u> +	4	59	<u>i</u>		LChbu	2	TCobut	2	Accel.	<u>s</u>	yate.	Ę		I.iquid	+	Gas	56		1			
Identification	Candidate Vehicle Receiver	T ank Function	ədifi	Quantity Guantity	Larke	Medium	abyete	CT huder	Ellipée	Torold	z ^{₩/NN}	Saudare 3	-	Caugary		Category	Acceletanon Capillan	Bladder	Other		ப்பதில்கப்	K	10 mg-200	754 0015	mg-860	=====================================	Cumment/Heference	
1 3	(m Orbit Assembly	MPS (wid.	1.02	15300	L	H H				×			<u> </u>	<u> </u>	20	-	× -		ت ×	<u>5</u>	o.		0.224	5	×		ASD-AFS-77-005-10, p. 259,	-
<u>,</u>	Chemical ULV. 14606 lbs Payload. 1660 LBF Pamo	MFS Fuel	2H.1	2556 2556 62561		×		×						÷	65		×		Ń ×	<u>- </u>	yu		0.032		×	N	72, Hef. 22.	
	Fed Ct 'ogenic Eng	kCS	N2H2		<u> </u>	<u> </u>	ļ	1	<u> </u>	+			+	3	1¥ 30		+	×	270	- T - A			33.8	2	×	1		
- Pr	SPS Geo Abs'y OTV Common	MPS (201d.	1:02	1±7396 (413096)	×				×	+	+	+		6-1	17	+	×		3	5	0 0	+	202.8 1(*	×	-16(20242-4, p.353, 1st	
	Stage, Payload 50000 lbs, 50000	Mrs Fuel	1.H ₂	34500 (17000)	- <u>×</u>												×		Ñ.	<u> </u>	30		33.5	- 5 -	×	5 5 5	lage - 4 each, 60000 lb Eng. , ad Stage - 2 each, 60060 LBF 	
	LBF Ergunes	RCS	л ₂ н 4		├	*	×								V	-		×	276 -	CT AN	مد	-		+	+	۵ 	ugine, net. 20.	
Ξ	STS Corbiter (Space Shuttle, (OTV Ser.)	OMS	570 ⁴	3520 77 ⁻ 91×2		×		×		17,	N PS	<u>i</u>		1-5	⊽ 		×		7 867	10 A1			10.6-1	1.1			able A-1, RASA CR134911;	
	2 ca OMS Eng of Greet has for Hr S		HWW	2130		*		×		12	Se Me	ž					×			<u>5 7 (</u>			5.4-6	 ;		<u>.</u>	ASD-ASD-75-601 (JCS-13967) . 7. Vol. I, Vol. II, p. 4-41	
	Eng of Efu lbs. 6 hCS Eng of 25 lbs.	HCS	*07x	520 1360,43	+	×	×	+	+	16	59 Mc		<u> </u>	1-0	v v	-		×	∓ 1 62	10 C	<u> </u>		-			=	aru 4-14; Ref. 1, Ref. 17.	
)		IIWW	065 6× 105:		×	×			<u> </u>	Ne.	, d	<u> </u>		• <u></u> .u.			<u> </u>		A C	4 <i>2</i> 2						<u> </u>	
2	SIS Orbiter Exter-	Oxidizer	- ⁷ 01	1331000	×			×	+			15	13	K (]		×		Ê	б х	<u>ں ہے</u>	ri yu						able A-2, CH134911; Ref. 1	
!	Monfred W Add'l	Prop	H H	222000 1459400	×	<u> </u>		×	+			23				×	·	+	Z.	<u>5</u>	2.0	_						
53	Cargo Orb. Transf. Vehicle (COTV)2-	MIS (mud	1.02	195000	×		[Ĺ_	×			ļ	 	1-0	-		×	<u> </u>	5	5	50	+-	< 185.		+-	<u> </u>	CS-12973, p. VI-C-32, 33;	
	1/2 Stage 1/62/ L.H., (225 1 of Pav	MPS Fuel	H _z	36300	×			×			<u> </u>		-12-1	<u></u>	·		×		51	5	yo.		< 30.		·	a 2 :	180242-3, Vol.3, Sec 6.3, 2/31/76, Ref. 23, Ref. 14.	
	Liverb 1st Stage 11 to a 20K erze. 2nd	A IS/RCS	707			×	×		 					5	v v	-	×	1	2	5	×0	-	<67.1	× cc	+	≍ <u>ज</u> 	S 11569, 3-31-76, p. VI-D-2- Grunnian ECON-100, 3-77,	
	stage F ea. 20ff Endine		7H.1			×	×										×		50	CL	٨u		чи.	26 ×		<u>. </u>	lov;Ret. 2, Ref. 24.	
																		· · · · ·			 		 					
	(Almderical Tank + F	L Threaddal E	nds –		-			-			_		-	_		_		-	_	-	-			-	4	-		

D - inlet, @ - exit

•

- Cylindrical Tank • Ethpsotdal Ends

. L									ĺ													•	ì			
										ð	crating	Vent V				lound	Litio	9	Tomer						1416 4 01 0	
	N W					-	4	۶Ľ	ŝ	<u>د</u>	CESSURG	Pres	ssure	Acci	el.	iysture	ε	Ц	Liqui		Gae	Flow	Rate			
	Identification R C	Tank Function	aqYT	ice (ip) Guantity	Large	Medium	Sphere	Tabauty	Ellipse	KN/W5	Category	۲ ^۳ /۳	Calegory	k	Category.	Cepillery	Tabbalg	Techo Y			A103018;	£∕.w ₅ -≋€C	/2 001	/201001	Ecc.	
ń	4 Personnel Orbit Transfer Vahicly	N IS	°.	40530	×				×	 				1	× 	-		3		*		M	>@	<	E Comment/Reference	T
	(POTV), (75 Passengers + 20	K PS	LH2	1550 1	м			×							<u> </u>			20	Cryo			9.1 4 5	× •		JSC 11568, 8-31-78, p. VI-E 16, VI-E-17, JSC 12973, Juby	
	tons UP), Stage 4 ea. 15000 lb en	1 - APS K.	N2H4	8		1×	×	\pm	+			T		17	× 	1		278 <	T AVE						Ref. 23.	
	Stage 2 - 2 ca 15000 LBF eng.			2610	·	×	×		<u> </u>									< 31	C						17, VI-E-18, Ref. 2	
35	Crew Rotation	M FS Chald	102	40530 x		\vdash		+-	+					+-	-		-+-	06	Cree				e			
	4 ca 15K Eng. Stage 2 = 2 ca	MPS Fuel	LH2	7550 x				•							H			20	Cryo		/ <u> </u>		×)		Grumman Econ-100, Mar 77, P109, Ref. 24, JSC-11568,	
	ISK Eng.	APS	N2H4	2610 (5750)		H	×	+	+			1		-+			- <u>-</u>	228 <	T Avg				:T		p. vi-r-10, 8-31-76, Itef. 2	
8	SPS GEO Ase'y	MIN Cald	671	185200				┥-'	<u> </u>			+-	+	+-	Ŧ		+-	16 >	Con			T	T			
<	OTV, Common Steel 10-71 H		-	08300)×2								<u></u>	1	4.	× -			3	Cryo			75.8	×		Boeing D180-20242-2, Ref. 21	
	500,000 lb paylon		<u> </u>	37050 x			×	e.							×			20	Cryo			12.6P	×		P 577; D180-20242-4, p. 355, Ref. 20, JSC 12973, p. VI-C-	
	capability/alao COTV _G) 1 ea 2000t lb F crigine	S I V	N2H4	1304 2875)×4	······································	×	×					<u> </u>	<u> </u> 	-	-		+-	276 <1 312	Avg.		1			\top	30, 31, Ref. 23; D180-20242-3 NAS9-14323, Vol. 3, Sec. 2.1, R.4. 20	n .:
2	Space Based Com- mon State Const-	- MFS Oxid		784400			┼──	*		13790	High	+	+	+	-	+	+	8					1	-	Boeine Report Disa. 22276-2	
	ruction, 400000 Ki pay load capability,	G MPS Fuel	CH3	130700 x			<u>×</u>	•		13790	High		Ĵ	-5	-			20				57.89		×	Dec. 1977, Vol. II, pp 137, Nef. 29.	
	lat stage of 4 caid: 1.BF engines, 2nd	WPS Cadd	1.02	784400		<u>†</u>	╂	×		13796 4	1 kgh	+	+			+	-	6		1	+-	72 66				
	LBF engines. LBF engines.	MPS Fuel	L.H2	130700 x			<u> </u>			106781	ţ.		• •	<u>.</u>				20				28.94	×		Boeing Report D180-22876-5, Nef. 3u. Dec. 1977, Vol. V, D. 48. D. 265 (h.m. 270)	
	2	AFS Thruster	ۍ دري			×				189.51	NOT	+		-0	+_	+	×	100-320						- <u>-</u>		
		(400 Ist)	GH2			·····				1 3.68	N		9-6	ۍ د				30-20								
	x* Cylindrica	I Tank + Ellip	with En	1	1	-	-]	-	1	+	E) = 1	nlet,	= =	exit -						-		\neg			

•

TABLE A-2. IDENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

ORBITAL TDA

. ľ

TABLE A-2. IDENTIFICATION OF CANDIDATE VEHICLE NECEIVER TANKS

OHBITAL TRANSFER VEHICLES (OTV'B)

			9		July July		Concept	0 -01/9, 1-15-76 216, Fry	et. 26.	 ,			5; hel.		1 - V 171	. 2-7,		
Page 6 of 1			cat/l'efervad	Solar Power	anceja Evaluz ailed report '77 n. VILC		Н. Гротав, (→7701-77-C= -77-605-10, - 255-265. (2-26,2-29, H			-73394. eat	stems, M5F(2-45, Hef. 19 4 b f. 20	11 HEL 23.	STS Orbiter	r 134911), ('c PLN 78-029,		
			Comm	JSC 12973,	Vol. 2, del 76 to July	kef. 24.	Convarr, H	Review, Fi CASD/AFS Ref. 25. 1	75-029, p.			NASA UNX	transfer by p. 15-21, 4 27 - 25 -	Fr4	OMS from 5	Iron GDC	Kel, 2t.	
			سی-∎حد > 100 ا⊄/												1			
		Ŀ				1-			×	×		×	×		×	×	1	
		Flow Ra	кк\ш ₅ -вес		*				12.5	6. 1 5		16.9	5. B		J. 6-12.9	5.6-6.45		
ľ			ر در در منه					_										
	ure	3	к															
	a fur fat		Category.				20	i i	341	ting		1			Avs			
	5 Te	1 Iquite	к	36	50		101 ± 762	<u> </u>	01 = 267			50	20		L > 422	< 312		
Ì	5		Other			×	×	×	×	×							1	
	i i i l	=	Bladder				ļ				×			×		_		
	nb:	y's fe	Veculeration			-	×	×	×			×	×		×	×	×	*
	<u> </u>	<u>.</u>	Category.	*	<u> </u>		-				-			-	 		 	
		Acce	لا	- 6.5	-0.5		P . 7		0.05		20.0		N 1	N 1	<u> </u>		<u> </u>	
		212	و عرولامن).		->		<u> </u>				<u>ن</u> ـــــ	<u> </u>	~~~~					
	Vent	Pres	_{لالا} /س ²															
	rating	Sure	Catogory													Med		Med
	ŝ	Ъrе: Г	K2€\ ^m 5												1766	1760	16.50	14.90
		2	Torold														[
		-	13008175	×					¥.			×		ļ			¦	·
		3	alayda		×		×	×					<u>×</u>	•	×	×		<u> </u>
		┥	Ilams		······	×		*	<u> </u>				t t	 	×	×	***	
ļ		2	Medium				ж.		×	×		×	×					
		<u> </u>	Lange	×	×													
		uld	kg (lb) Quantity	243000 5357001×2	40500 153001-2		4670 103001/3	2310 5100/×3	31000 (65300)	1 +000 (35300)		1+540 - 40-70	3464 1630		3524 12210,-Z	2130	119	1 (022) E - (022)
		Ξ	ad KL	7071	1.H2	RCS	524	HIVEN	N204	MMM	₩SN	71	ίH.		N204	HMM	N_04	HWW
			T ank Function	MI'S Cand.	MHS Fuel		MPS (And	MIS Fuel	PDC) SIW	MFS Fuel	kes	Mr. wid.	MIS Fuel	KCS	MIP Cand	I'm I SI W	ы К С	
			C andidate Vehicle Receiver	JSC Cargo (rthit Transfer Vehicle,	CUTV6		Low Thrust Liq UTV Press. Fed Mur-	able 6000 LBF Erg.			Low Thrust Liquid O.I.V., Pump Fed Storable Too LBF Engtro, Payload - Moto LL	Aeromaneteering	Shuttle (AMCAD), Fas load 10060 Lbs	the 20000 La have.	1.1.1. orbital Veb. Based on UMS	Competionia, 2 na OMS Engines (Guen LBF, 35 ca. RC 5 Enginee of 870	1155, partized 6000 Lo 24000 1105
		.0	N aottastittesbi	ສ ບ			5				2				n;			

(i) = inlet, (e) = exi:

,

 $\mathbf{x}^{\bullet} = C_{\mathbf{y}}$ limitical L tak + Ellipsoidul Ends

TABLE A-1. EDENTIPICATION OF CANDIDATE VEHICLE RECEIVEN TANKS

•

 Candidate Vehicle Identificate Vehicle Receiver Plots SFS Plant (15M W), 347 X 374 M, 340,000 K/ (750,000 LBS) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1986) (198	- Tank Function Punction RCS RCS RCS RCS RCS RCS RCS RCS RCS RCS	A H2 C LAT 1.1	huld huld 23000 /rr 23000 /rr 23000 /rr 1235/20 (1550) (1650) (1650) (1650) (1650) (1650) (1650) (1650) (1650) (1650) (1650) (1650) (1750) (1650) (1750) (1650) (1750) (1650) (1750)			P(010] 2.2.4/NJ 2.2.4			Accel 2 2 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2			1140 I I I I I I I I I I I I I I I I I I I	Category	R C	Cutegory	× <100 kg/ kg	\\$\ 2998-2m ₩ ∞ ₩ 0 ₩ 0 ₩ 2 ₩ 0 ₩ 2	Page 2 of 6 Comment/itefere ocing D180-20242-4, 47, Ref. 20. 9 3.2-4; M 7 3.2-4; M 8 3.2-4; M 8 3.2-4; M 9 3.2-4; M 1970, Ref. 21. B 1977, Ref. 33	P245, P255, P245,
GSS-Geosyn. Sp. Sta. (8 Man, Earth Applic. + Science), (283, 000 - 348, 000 lb) 1988) 1988) CVlindrical Tank + pr	Crbit. Position Adjust RCS N	PH4	1840					0-0.1			5	8 k T 312				 	1 G 7	80-20242-2, P 107, 1	Ref.

* 64.6 anariye V

(i) = inlet, (e) = exdt

x* = Cylindrical Tank + Ellipsoidal Ends

Ì

 Γ

TABLE A-3. IDENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

ŧ.

TABLE A-3. IDENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

Ż

SPACE PLATFOHMS/SPACE STATIONS

8	
63	1
ž	
Δ,	

P

	Comment/Heference	160-20242-4, P 194, Ref. 20,	180-20242-2, Ref. 21.	180-22876-5, P150-155, ec 1977, Ref. 30.		D-77-AD-0054, Ref. 34.)-78-AD-0055-2, 4-14-78, ef. 35	160-20242-2, Р 362; гчилтал Есси Е104, 3-77, Р 3, (LC ₂ /LH2 & 800 KG/ day),	tef. 2, Ref. 24). 160-22876-5, 12-77, P 34, 35 81 <u>6</u> , 2-34). P120. Ref. 30 .	180-22676-2, Vol. II, 12-77, 59, Ref. 29.	180-22876-5, Vol. V, 12-77, . 59, Ref. 30.	160-20242-2, PC"3, 374, 380, 16, Ref. 21.
	228- ⁵ m	<u> </u>	<u> </u>	<u>ađ</u>		<u></u>	보포	<u> </u>	<u>= 0 3</u>	<u> </u>	<u></u>	<u> </u>
	/\$b(001< >30€-∋u											
tate	/\$1001>											
Flow F	100-2000 ydd										•	
	CureBoth											
Gas Gas	к											
nperatui	Calegory											
Ter Liquid	к		278 < T < 312					278 <t < 312</t 	278 < T <312		% 97 7	
15	ાગ્યા()											
iisili cm	Capillary											
Acqu	uopnaolooov										·	
	Calekori.		< 1	1,	1 1	1 >	-				l v	v
ق ا	K		-ù.2	-0.2	-0.2	0.2	<u>-0.2</u>	-0.2		0-0.2	9-0.2	0-0.2
c.	Caugan's		a	a	¥	¥				- -		
Vent	z ^{ur/NM}											
ating	Currenty.											
Opera	7 ¹¹¹ /57											
j z	Edinse											
t of t										32		
8	arouds										н н	
	IIam2								06)		N N	
Siz	yieqimu a		ļ									
	Quantity kg (lb)		1 43 0 (3150)					1 60000 352700) 9000 (19800	31 × 10 ⁶ 53 × 10 ⁶) 1. 4 × 10 ⁶ (3.	45900 (90200)	1250 (2519, 214	1000 1000 2200) 2200) 11000//Yr
Piel	adX_L		N2H4					N2H4 of Arreon	N2H4 or Argon	LAF	LG ₂ LH ₂	LH2 102
	Tank Function	Orbit Position Adiust	RCS	Orbit Maint.	RCS	Crthit Maint.	RCS	Orbit Keeping (ter vr)	RCS (per yr)	OAS + BCS	Equinoctal Occultation Control	Equinoctal Osciliation Control
	Candidate Candidate Vehicle Receiver) Orthiting Lumar Sta. (12 Man), (277,000	(1992)	 Low Earth Staging Depot/Laboratory (72 man) (1967) 		I Low Earth Manu- facturing Facility	(1969)	2 Low Earth 300 NM Const. Base (Fac.) (1.6/10 ⁶ lbs to	20.66×10 ⁶ lbe) (1967)	3 Geosyn, Space Part Satellite, 10,000	MW, 86×10 ⁶ KG (190×10 ⁶ lb) to 97 × 10 ⁵ KG	I I200/S00 MW Syace Fower Satellite Sta.), 19.5 to 22 Million Lb (Prototype Satellite) (1550-1595)

A-13

(1) = inlet, (e) = exit

•

 $\mathbf{x}^* = Cyindrical Tank + Ellipsoidal Ends$

TABLE A-3. IDENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

SPACE PLATFORMS/SPACE STATIONS

Paga 4 of 6

1

-

Fluid	Cuantity Type Type	nous OAS Inten-	-1 Men, RCB 710 KG	Dowered Drag Adj GO2 16 Proces- + AC8 (3520)	Flant Or 012 1000 Plant Drw Adj Excess 1030 + RCS CO2 22700	lilte RCS chous A Bta. ba)	orm RCS ^N 2 ^H 4)	Loop International Contraction	acus Orbit n Base, Keeping N2H4 or 3 N00 KG RCS N2H4 or 3 15-67 Argon (7	
Size	ارد (1b) ۲۰۰۱ عدمی ۳۰-۵۵			0000 × 00/yr)	000 x 000 yr				1001	
Geometry	liams Sphere Sph			и	х н					
Operating Pressure	Catogory KX/m2									
Vent Pressure	Culogory KV/m ²									
A cel.	כמניצסד <i>ז</i> . רמניצסד	0-0.2 < 1	0-0.2 < 1	0-0.2 ~ 1	-0.2 < 1	-0.2 < 1		-0.2 < 1	-0.2 < 1 -0.2 < 1	
Arquisition System	Acceleration Capitizer Mader									
Tempera	נ ערפע סד <i>י</i> א									•
dure Gas	Criegory K			Wide Range	Wide Range		}			
Plow Rate	> 100 F8/ < 100 F8/ < 100 F8/ F8/ F8/ F8/ F8/ F8/ F8/ F8/ F8/ F8/									
	Comment/Reference	D180-20242-2 (NAS9-14323).	Vol. 2, 31 Dec. 1976, pp. 73- Ref. 21.	CASD-ASD-78-001, JSC 139, p.7-14, 7-56, JSC Orbital	Prop. Handling Pi5-18, Ref. 17.		CASD-AFS-77-005-10, 1-17-78, Ref. 25.	Eo-8A, LANDEAT D, NASA Prel. Payload Descriptions, Vol. 1, Automated Payloads, July 19°5, P 5-1, Ref. 9.	Di 50-22876-5, Dec. 1977, Figure 2-33, P 33, P 145, Re 30.	

(i) = Inlet, (e) = Exit

	Candidate Vehicle . Tank Receiver Function	eosynchronous Orbit Total Antenna Position arm (OAF) Adjust	RCS		-us kw rower Backup khule (Small RCS, wer Module) Moment- wen Dump	KW Power Backup dule RCS	tenna Farm RCS astruction uform	Ktal Antenna Th	dstics Tank, Gross Wt.	54 Lbs (Used b CTV's)	4	
4	adil			į	GN2				102	LH ₂	N2H4	
luid	kt (10) Gasutty								20000	3600	181	(400)
Size	Larke Larke		<u>† </u>						*	×	×	
	lleinZ		+								×	
i-rij	Sphere		+						×	×		
a trant	Ellipse										+ 	-
 ;	Toroid		<u> </u>									
C De	am'ny								 		1	
suite.	כ שוטלטובי. 											
- C	Faxm ²	ļ										
nt	(united in the second										+-	•
	K 2	ļ	ļ									
<u> </u>	A conforment of a		+						×	×	-	
n de											- ×	
silie	=		-								-	
	Cifict .								96	20	278<7	
Temucr	ດ ຊີ ຊີ	1										
ature	×											
	Category											
	5 008- m/2h	۱ 							<8000(1	< 600(1)		
	2 / 2	<u>}</u>										
	100 pm /	< u			<u> </u>				×	*		
-	535-2t	0 Å Å	553	156	ž 2 ä	De E An		28 A	NN	P 4 025	Han 1	
		Comment/Itelerence .E. Stationary Platform ublic Service/Antenna F 20 of OTV (Concepts and	uestions). DC Briefing 1502 8060JC Inc 1976, D. A. Heald, I	6, Ref. 36, pgs 20-29, 1 5, No. 9, Astro. 4 Aer.		udilary Payload Power udy, DPD 533, Configura finition. Ref. 39. AAP	nau rite, MUK 1446, 2-2 ef. 40. decagon Structure Assy necept; Viewgraph, P 20. ef. 41.	ge 20 of Ref. 41. ge 19 of Ref. 36.	SA/Conval r CR-134911	4-43, 5-76, CASD-NAS7 5, 12-77, P 1-2, Hef. 1	A. Heald, 'Orb. Prop. adling & Stor, Sve. for	

.

TABLE A-1. IDENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

SPACE PLATFORMS/SPACE STATIONS

£

March 1

Рид Зала Солонали Констан Кон		Г						<u> </u>	3				
н н	Page 6 of 6				Comment/Reference	COUVER VERCER DELA BOOK, 4-26-77, Ref. 1,		GDC CM78-007, Orbital	Service Module (OSM) System Analysis, 12-22-77, Ref. 43.		PUC ISSUE, CABU-ABU PUC Ref. 42.		
Солональной Солональной Солональной Солональной Солональной Солональной Солональной				100 102	,	< ×			1	╈			_
Слование Слование Слование Слование Слование <				398-20	u –				+-	-+-			
Сонит Сонита Сонита<			킕	/ = 1001	<u>}</u>				\downarrow				
Вание Соневсий Соневсий 1 Соневсий Соневсий С			Plow	tk∖m ² -sec		1)009 >							1
Вать Delivery К К К К Солотись Солотись Солотись Солотись Солотись Солотись			g	Cutegory	<u>}</u>								-
Зали Саневал устание Саневал устание Саневал устание 1 Саневал устание Саневал устание Саневал устание Саневал устание 1 Саневал устание Саневал устание Саневал устание Саневал устание 1 Саневал устание Саневал устание Саневал устание Саневал устание 1 Саневал устание Саневал устание Саневал устание Саневал устание 1 Саневал устание Саневал устание Саневал устание Саневал устание 1 Саневал устание Саневал устание Саневал устание Саневал устание 1 Саневал устание Саневал устание Саневал устание Саневал устание 1 Саневал устание Саневал устание Саневал устание Саневал устание 1 Саневал устание Саневал устание Саневал устание Саневал устание 1 Саневал устание Саневал устание Саневал устание Саневал устание 1 Саневал устание Саневал устание Саневал устание Саневал устание 1		ure		К	1		Í						-
Солональной		Crat	1	Category	+		+		+	╉			
93 64 94 144 144 144 144 144 144 144 144 144 144 144 144 144 144 144 144 144 145 144 145 144 145 144 145 144 145 144 145 144 145 144 145 144 145 144 145 144 145 145 145 144 145 145 145<		emp	ᆋ					-					
Вани ранков Сониндовнии мо. Сониндовнии мо. 1000000000000000000000000000000000000				ĸ	95	58				274	-370		
30 60 60 60 7 Identification No. 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000		Ę	Ļ	Other						\top			
30 00 <		usit	≡ŀ	Bladder	 		_			×			- ;
3 3 <td></td> <td>Ъ.</td> <td></td> <td>Acceleration</td> <td>·+</td> <td></td> <td></td> <td></td> <td><u> </u></td> <td></td> <td></td> <td></td> <td>- 8</td>		Ъ.		Acceleration	·+				<u> </u>				- 8
Generation 1000 Generation 100 Signal Generation 100 Sign	Ì	<u>,</u>	;	Category	+					+			10
60 68 68 7 Identification No. 100K 100K 100K 100K 100K 110K 100K 100K 100K 100K 110K 100K 100K 100K 100K 110K 100K 100K 100K 100K		•	N.A.	8						+-			
Size Cambra Cambra Cambra No. Cambra Cambr	.[2	Category.	1				┢──	+			- =
60 <			LTCbs	rn/m ₅			-+-			-			-
Filid Size Canification No. Filid Filid Size Gi Contrait Contrait Filid Filid Size Filid Filid Size Filid Filid Filid Filid Filid Filid <	ather		2112	Category.									-
63 Water Delivery Delivery Levin 10, 2 36700 Moretton No. 1000 Moretton 10, 2 36700 Large Convert 1000 Moretton 10, 2 36700 Large Convert 1000 Martino Text, 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	(1.44	5 6		لايار <mark>س</mark> 2					-				-
63 Water Delivery Water Delivery Track 10000 Water Delivery 10000 Water Delivery 100			╞┝	DioroI							<u>£</u>		-
65 1 Logiation Yo. Candidate Vehicle Vehicle Vehicle Ruccelver Ruccelve		1.111		Japunta			\bot						-
6 1 Identification No. Candidate		Ĵ	\$ -	obuete	—	_ <u>*</u>	-+-						-
65 Logistics Task Cambridge Cam	h	-	+	llame			-+-			 *			1
63 Verhicher Camhidate Camhidat		1.10		umper	×	×					·		
67 Logiatica Tank, Tank Relice Tank, Candidate Tank, Tank Relice Tank, Lugistica Tank, Ressing Robin Bervice Station Module Rocs Reston Ressing Robin Bervice Bervice Station Rocs Reston		0	"	LAIRC			-+-]
61 Identification No. Cambidate Vehicle Vehic		but		kt (1p) Grantfo	36700	(00010) 6710 (14800)			<u> </u>	4.25×106	9.36×106		
67 Identification No. Cambidate Vehicle Riccelver R				adı´l	102	LH ₂			GN2	0 ² H			
67 Identification No. Candidate Vehicle Kocelver Bis Orthual Service Module Taak, Taak				. Tank Function			Station	Keeying	RCS	Delivened	BCS		lipsoldal End
S Identification No.				Candi date Vehic le Keelver	Logistics Tank, 100K		Orbital Service	Module		Water Delivery	1		/lindrical Tank + El
	L	۰0	N U	Identificatio	67		68			59			ິ ເ ເ

6-49 57

TABLE A-4. DENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

Å.

70 MSR-MARS Sample Tank Receiver 70 MSR-MARS Sample Crbit Adj. 71 MSR-MARS Sample Orbit Adj. 71 MSR-MARS Sample Orbit Adj. 71 Automated Construction 71 Automated Geosyn- Orbit Adj. 71 Automated Geosyn- Orbit Adj. 71 Automated Geosyn- Orbit Adj. 72 Lurgey Return 73 Automated Geosyn- Orbit Adj. 74 Automated Geosyn- Orbit Adj. 75 Low Zarth Orbit Orbit Adj. 73 Large High Energy Mailor RCS 74 Ussell Orbit Adj. 75 Low Zarth Orbit Orbit Adj. 76 Orbit Adj. RCS 77 Large High Energy Mailor N2 78 No N2 790 Stone Stone 700 Bayload) RCS	Aunutic Small			THE REAL PROPERTY IN		1000	isition	Temps	rature				2 5 V DB -
70 MSR-MARS Sample Function FN 70 MSR-MARS Sample Orbit Adj. Function 70 MSR-MARS Sample Orbit Adj. Function 71 Automated Function FN 71 Automated Orbit Adj. Function 71 Automated Geosyn- Orbit Adj. FN 72 Low Earth Orbit NCS FN 73 Low Earth Orbit Crothit Adj. FN 7300 Bayesal Vehicle Orbit FN 7300 Nuclear Waste Orbit 730 Nuclear Waste <th>Small</th> <th>1</th> <th></th> <th>Fressure</th> <th>Accel.</th> <th>Syste</th> <th>ε</th> <th>Liquid</th> <th>Ľ</th> <th>28</th> <th>Flow Rut</th> <th></th> <th></th>	Small	1		Fressure	Accel.	Syste	ε	Liquid	Ľ	28	Flow Rut		
0 MSR-MARS Sample Orbit Adj. Return (Automated Planetary) Spacecth RCS I 1 Automated Geosyn- (1989) Orbit Adj I 1 Automated Geosyn- Chronous Satellite Orbit Adj I (Ibasater Comm, Vehr RCS I I (Ibasater Comm, Vehr RCS I II (Ibasater Comm, Vehr RCS II II 2 Low Earth Orbit Orbit Alt III 2 Low Earth Orbit Orbit Alt III 2 Low Earth Orbit Orbit Alt III 113 Large High Energy Maint. III 2 Low Earth Orbit Orbit Alt III 1090-1985) RCS N2 91 1090-1985) Orbit RCS II 1090-1985) Orbit RCS II 1090-1985) Orbit RCS II 1090-1985) Orbit II II 1090-1985) Orbit RCS II		Sphere Cylinde Ellipsc Toroid	Catekoly KV/m2	Jogon V	A108018	cceleration	inder inder		61092	regory	, m ² -86C		
Planetary (ubmated Planetary Spacecfi RCS Planetary Spacecfi RCS Planetary Spacecfi RCS Planetary Spacecfi RCS I Automated Geosyn- Orbit Adj I Automated Geosyn- Orbit Adj I Automated Comm. RCS Veh. Tracking. Energy Monitory Large High Energy Maint. Large High Energy Maint. BOO Ib (1980-1985) Nuclear Waste Orbit Adj RCS No Starge High Energy Maint. RCS Starge High Energy Maint. RCS Starge Lorbit Adj				2	C H	o V	B		K P	Ca	/54	01 <	Comment/Reference
1 Automated Geosyn- I. Automated Geosyn- (Disaster Comm, Veh. Tracking, Energy Monitor) Orbit Adj (Disaster Comm, Veh. Tracking, Energy Monitor) BCS [1396] BCS [1396] I. [1396] I. [1300-1965) RCS [1300-1965) N2 [1300-1965) Orbit [1300-1965) Direction [1300-1965) Direction [1300-1965) Direction													PP 309-322, D180-20242-2, Ref. 21.
(Disaster Comm, Veh. Tracking, Energy Monitor) BCS 13 (1986) Large High Energy Large High Energy Maint. 0.13 1. Low Earth Orbit Orbit Alt H ₂ H ₄ 113 1. Low Earth Orbit I 0.7bit Alt H ₂ H ₄ 1. Low Earth Orbit I 0.7bit Alt H ₂ H ₄ 1. Low Earth Orbit I 0.7bit Alt H ₂ H ₄ 1. Low Earth Orbit I 0.7bit Alt H ₂ H ₄ 1. Low Earth Orbit I 0.7bit Alt H ₂ H ₄ 1. Low Earth Orbit I 0.7bit Alt H ₂ H ₄ 1. Low Earth Orbit I 0.7bit Alt H ₂ H ₄ 1. Low Earth Orbit I B. 1. Low Earth Orbit I 0.7bit Alt H ₂ H ₄ 1. Low Earth Orbit I B.													
Ellergy Monitor) Large High Energy Maint. Large High Energy Maint. Descretory - RCS N2 91 (1980-1985) Nuclear Waste Orbit Disposal Vehicle Direction 0.00 LB Payload) RCS	-+												Aerospace Co HTR-76(7365)- PP 46,47,48, 4-76, Ref. 44
Low Earth Orbit Orbit Alt H ₂ H ₄ 113 Large High Energy Maint. (250)×. Observatory - RCS N ₂ 91 3000 [1980-1985) (290) (1980-1985) (1980-1985) (200) Nuclear Waste Orbit Disposal Vehicle Direction (200) (30 KLB Payload) RCS													
3000 lb RC3 N2 91 (1980-1985) RC3 N2 91 Nuclear Waste Orbit (200) Disposal Vehicle Direction (300) (30 KLB Payload) RC3 RC3	а 4 ж		Low	+	0.0.1 <1	+	+	278-312 Avg					Pavlnad Docentry
Nuclear Waste Orbit Disposal Vehicle Direction (30 KLB Payload) RCS			High		1 > 10'								
(30 KLB Payload) RCS						-+-				-+			
(1991)					+	+	-+-	-					Aerospace Co, ATR-76-(7365) 1, PP 84,85, 4-76. Ref. 44
Advanced TV Orbit Broadcaat (Geo-Adjust, struch) Satellita Do													
14000 Lb				+		+	+			+	╉	Ī	леговрасе Со. АТR-76(7365)- Р 53, Артіі 76, Ref. 44.
High Resolution Orbit Earth Mapping Maint. (-200 NMG Orbit),				+				-					verospace Co. ATR-76(7365).
Antenna/Satellite RCS Wt 110,000 I.b. 16×100 ft Antenna				-									. Vol. П, Р 58, April 76, tef. 44.
Adv. Resources/ Orbit Polution Obs., Wr Maint. 30000 Lb, 500 NMI												<	erospace Co ATR-76/73651-1
Orbit (Sun Sync) Size: 30 × 60 ft					+							> #	ol. II. P 57, April 76, ef. 44.
Vlintricel Task 4 succession										_			

.

(i) = fnlet, (e) = exit

and a second s

TABLE A-4. DENTIFICATION OF CANDIDATE VEHICLE RECEIVER TANKS

ŝ

AUTOMATED SPACECRAFT

をすいここ

I																										
••				616A	د 					8	erating	Vent				Acc	Juisi	tion		empera	lure				Γ	r 20 2 01 2
DN U					1	äÈ	╧╋			£	CENTE	Pres	sure	Accel	_	sys	ten	F	Liq	P	g		Flow R	ate		
ldentification	Candidate Vehic le Receiver	Tank Function	Lype	kg (lb) Quantity	Large	Medium	[lam2	Svlinder Tebritv2	Ellipse	Loroid MV/m2	Category	Ku/m ²	VIOZUTY	8	Category	Acceleration	Capillary	Bladder	к	Category	>	Aregory	298-2m/3	15-86C 100 kg/	5-86C	
11	Coastal Anti- Collision Radar.	Orthit												+		1	+	-				<u>,</u>	*	ш >	u <	Comment/Reference Aerospace Co. ATR-767765
	1000×1000 Ft, Synch Equat. Orbit	RCS L				+	+	-			-				1	+						+				Vol. II, P. 46, April 1976, Ref. 44.
18	Intentionally Left Blank						ļ						<u> </u>		1	1	+					+				
5	Intentionally Left Blank					+			-		_			_	-											
2	Geosynchronous Manned Mission Module				<u> </u>	+			+						+	+		+					-			Adv. Study Prog Meeting, R. Piland et al, JSC, 8/30-31/1
=	Manned Utility Spacecraft (MUS)				1	+			+			1	+		+						+-		+	-+		Ref. 45. Same as above.
2	Orbital Construct- ion Demonstration	ACS (per 6 mo)	N2H4	7580 (16700)	—	<u> </u>			+	1400		1		0.2 <	-	+			278 < T	BAV			+		- <u> </u> -	Orbital Construction Demon-
	Article Mass 37073 KG (81790 Lb)	Pressur-	GHe	88		H	×		+			+	Jā	0.2 <1	+	+	+	+	/312		+	+-	+	+-		stration Study, Final Report, N55-DC-RP-008, 12-1-76,
		Ion RCS Thrusters (per 6 mo)	٨٢	523 (1152)		M	×					+	<u> </u>	0.2 <1		+	<u> </u>			+				+	<u> </u>	Grumman, Ref. 46.
	Cylindrical Tank +	Ellipeotdal 1	Ende		-		_	-	-				÷	= inlet,	- e		- =	-		-	\dashv					

антона 1

TABLE A-5.REFERENCES

1. GDC

Low-G Fluid Transfer Technology Study Final Report, NASA CR134911, CASD-NAS-76-014, J. A. Stark, etal, May 1976.

2. JSC

Initial Technical, Environmental and Economic Evaluation of Space Solar Concepts -Volume II - Detailed Report, JSC-11568, 31 August 1976.

- RI-SD Prephase A Study for an Analysis of a Reusable Space Tug, Final Report, SD 71-292-4, 22 March 1977.
- 4. GDC Memo ER-78-004, 13 January 1978.
- 5. GDC

Space Tug/Shuttle Interface Compatibility, Interface Compatibility Study, NAS8-31012, CASD-NAS-75-, 017, June 1975. E. H. Bock, et al.

6. RI-SD

S-II Stage Interorbital Shuttle Capability Analysis, North American Rockwell - Space Division SD-71-145-1, 21 April 1971.

7. RI-SD

Modular Space Station, Phase B Extension SD71-214-4, Rockwell International, January 1972.

- 8. MDC Leo Space Station, Vol. 1, Book IV MDC GO 634, July 1970.
- 9. MSFC Payload Descriptions, Level B Data, Vol. I, Automated Payloads, General Dynamics-Convair Division, July 1975.

10. RI-SD

S-II Stage Orbital Propellant Storage System Feasibility Study, SD70-554, N.A. Rockwell - Space Division, 31 March 1971.

11. MM

Teleoperator Retrieval System (By R. J. Molloy, J. R. Tewell and R. A. Spencer), (for 1978 Goddard Memorial Symposium (AAS) March 8, 9, 10).

4

12. GDC

GD/C Vehicle Data Book, R. Drowns and Associates, 26 April 1977.

13. Boeing

Future Space Transportation Systems Analysis NAS9-14323, Boeing D180-20242, -2, -3, -4 Boeing Aerospace Company, G. R. Wookcook, et al, 31 December 1976.

14. JSC

Future Space Transportation Systems Analysis Study D180-20242-3 Transportation Systems Reference Data Boeing (NAS9-14323), 31 December 1976.

15. MSFC

NASA Technical Memorandum - Orbit Transfer Systems With Emphasis on Shuttle Applications - 1986-1991 NASA TM X-73394 - Prepared by NASA, April 1977.

16. GDC

Space Tug Systems Study (Cryogenic), Final Report, Vol. II - Comprendium NAS9-29676, January 1974.

17. GDC

Orbital Propellant Handling and Storage Systems for Large Space Programs, Vol. II, Final Report CASD-ASD-78-001, JSC 13967, D. A. Heald, M. H. Blatt, et al, 14 April 1978.

18. Intentionally deleted.

19. JSC

Orbital Construction Demonstration Study, Mid-Term Briefing Ald-On Study -NAS-OC-RP010, Grumman Aerospace (NAS9-14916), March 1977. 20. JSC

موجد وملادي الروار

R.

Future Space Transportation Systems Analysis Study D180-20242-4 - Final Briefing, Boeing (NAS9-14323), December 1976.

21. JSC

Future Space Transportation Systems Analysis Study D180-20242-2 Final Report, Boeing (NAS9-14323), 31 December 1976.

22. DOD/STS On-Orbit Assembly Concept Design Study (Preliminary Concept for Analysis)
 CASD-AFS-66-005-10
 R. H. Thomas and Team (GDC), 18 January 1978.

23. JSC

Solar Power Satellite - Concept Evaluation Activities Report, July 1976 to June 1977 (Vol. 2, Detailed Report) JSC 12973, July 1977.

24. GDC

(Proposal) Advanced Composites Design Data for Spacecraft Structural Application - Volume 1 - Technical GDC-PIN77-337, 31 October 1977.

25. GDC

DOD/STS On-Orbit Assembly Concept Design Study (Preliminary Concept for Analysis) CASD-AFS-77-005-10 R. H. Thomas and Team, 18 January 1978.

26. Orbital Refill of Propulsion Vehicle Tankage RFP-3833354, GDC-PIN 78-029, M. H. Blatt and associates, 1978.

27. MSFC

Extended Applications Study of AMOOS and AMRS - Final Report LMSC-HREC TR D497150, Lockheed Missiles and Engineering Center Cummings Research Park, 4800 Bradford Drive, Huntsville, Alabama

 Feasibility and Tradeoff Study of an Aeromaneuvering Orbit-to-Orbit Shuttle LMSC - D390272, July 1974.

- 29. Solar Power Satellite, System Definition Study Part I and Part II, Volume II Technical Summary D180-22876-2 (NAS9-15196), December 1977.
- Solar Power Satellite, System Definition Study, Volume V Space Operations (Construction and Transportation) D 180-22876-5 (NAS9-15196), December 1977.

•

- 31. AIAA/MSFC Symposium on Space Industrialization Proceedings 26, 27 May 1978.
- Space Station Systems Analysis Study, Grumman Program Review Presentation 17 June 1976.

33. MSFC

in the second

Space Construction Base Preliminary Study by Program Development Preliminary Design Office February 1977.

34. MSFC

Industries in Space to Benefit Mankind - A View Over the Next 30 Years (RIC Contract NASS-32198) Managed by Dr. Drafft Ehricke. SD 77-AP-0094, Joint Project of MSFC and Rockwell International, 30 March 1978.

35. MSFC

Auxiliary Payload Power System Study - Final Report Volume IIIA - Task I, Establish Design Constraints for the APPS MDC E1466, (Copy 71), 29 February 1976.

36. GDC

Near-Term Applications for Beam Fabricators 14 June 1978.

37. COMSAT

Orbital Antenna Farms 77 CLR37 (Reprint from <u>Astronautics and Aeronautics</u>, Vol. 15, No. 9, Pages 20-29. B. I. Edelson and W. L. Morgan, Sept. 1977.

38. JSC

Small Power Module - Status Review With Dr. Kraft NASA-S-77-5339, September 1977.

39. MSFC

Auxiliary Payload Power System Study, Phase II Thermal Control DPD No. 533 McDonnell Douglas Corp., 31 July 1976.

40. MSFC

-

Auxiliary Payload Power System Study - Final Report Volume II - Technical Report MDC E1466, Copy 14 MDC, 29 February 1976.

41. GDC Orbital Transfer Vehicles (Briefing) GDC 15028000 JU 604A D. Heald, June 1978.

42. Intentionally Deleted

43. GDC

Orbital Service Module (OSM), Systems Analysis GDC CM78-007 22 December 1977.

44. Aerospace Corporation

Mission Requirements for Orbit Transfer Operations Contract No. NASW 3141 "Coordination Meeting With Dr. R. W. Johnson," by M. G. Wolfe) 26 January 1978.

45. JSC

Manned Geosynchronous Mission Systems Analysis Study (R. O. Piland) Encl. M - Memo from HDQ Dated 14 September 1977 - Advanced Planning Meeting of August 20-31, 1977. 30-31 August 1977.

46. Grumman

Orbital Construction Demonstration Study, Final Report NSS-DC-RP-008, 1 December 1976.

APPENDIX B

SPACECRAFT ACCOMMODATIONS

Equipment Mounting and Software

- The Spacelab double racks (38'') can hold up to 580 Kg (1276 lbs) of equipment with a maximum volume of 1.75 m³ (62 ft³)
- Equipment mounted in racks may be as large as:
 - Width (double rack) = 94 cm (36.7 in)
 - Height = 149 cm (58.1 in)
 - Depth = 61 cm (23.8 in)
- Rack accessories include: interfaces with Spacelab data collection/processing/ transmission systems; power outlets; connectors for forced-air cooling of flight experiment equipment.
- Equipment unsuitable for rack mounting can be mounted directly to the floor of the Spacelab.

Electric Power

Spacelab electrical power is routed to flight experiments through experiment switching panels which can be mounted to individual racks, or under the Spacelab floor to service floor-mounted equipment.

- Electrical power: 28 v DC; 115/200v AC @ 400 H_/3 phase
- Power conditioning equipment
- Power available for payload and mission-dependent equipment in the Module is payload dependent

Ascent /Descent: 1 kw (continuous) 1.5 kw (peak)

On-orbit (all module configurations): 2.6 to 3.5 kw (continuous) 7.2 to 8.1 kw (peak) 160 to 300 kwh of energy available/mission

B-1

Thermal Control

Ð

The Spacelab Thermal Control Subsystem (TCS) consists of four thermal control loops:

- The module cabin air loop provides conditioned air, within established comfort criteria, for the crew within the module
- The module avionics loop provides forced-air cooling for equipment mounted in racks.
- The freon loop supplies cooling to cold plates mounted on any pallets that might be carried on the flight
- The module water loop exchanges heat with the three loops already discussed and, in turn, transfers Spacelab heat into the Orbiter coolant loop. The water loop also provides cold plate cooling for Spacelab subsystems, and can accommodate an Experiment Heat Exchanger and one Experiment Cold Plate.

The TCS transfer 8.5 kw, maximum, continuously to the Orbiter. It can also accommodate peak heat loads of up to 12.4 kw for 15 minutes every 3 hours. However, for Spacelab configurations incorporating a Module, part of the water loop heat rejection capability is used to cool Module subsystems. The remaining capability may be used by experiments and other mission dependent equipment. The water loop heat rejection available to such equipment is 3.7 to 5.2 kw, continuous; and 7.6 to 9.0 kw peak. This range of values covers all possible Spacelab Module configurations. Within these limits, the remaining thermal control loops have their individual capacities to provide experiment cooling.

- Cabin air-loop 0.4 to 1.6 kw, continuous; 2.4 to 3.6 kw, peak
- Avionics loop 3.3 kw, continuous; 8.7 kw, peak
- Freen loop

 8.0 kw, continuous; 11.9 kw, peak (Note: In palletonly configurations, the freen loop exchanges heat directly with the Orbiter loop. In this case the Spacelab total heat rejection capability is 7.0 kw, continuous and 10.9 kw, peak).

The Experiment Heat Exchanger can be mounted only in experiment double rack number 4 in the Spacelab Core segment. Its nominal heat transfer capability is 4kw. However, its actual performance depends on where it is connected into the water loop on the other loads along the water loop.

The Experiment Cold Plate must also be mounted in rack 4. When both the Heat Exchanger and the Cold Plate are flown, they must be connected in series and must share the same connection into the water loop. The cold plate is $500 \text{ mm} \times 390 \text{ mm} \times 4.4 \text{ mm}$ and has a $70 \times 70 \text{ mm}$ hole pattern. When filler is used to improve the contact between the experiment and plate, the conductance will be 0.07 watts per cm² per "C. Without filler the conductance is approximately 1.0 watt per °C per bolt area

<u>Di gital Data</u>

The Spacelab Modules's data management system enables the collection, processing, recording, on-board display, and transmission of low-rate digital data (including digitized analog) received from the flight experiments at 1 Mb/sec or less. Experiment produced data with rates between 1 Mb/sec and 16 Mb/sec can be stored and/or transmitted to the ground, but these data cannot be processed or displayed on-board the spacecraft.

Depending on the needs of the experiment, low-rate digital data can be:

- Processed and analyzed on-board the spacecraft using experiment supplied software
- Formatted and displayed for review and analysis by the Payload Specialists on-board the spacecraft.
- Annotated with voice recordings and/or time marks.
- Recorded on data tapes or transmitted to the ground.

High-rate digital data can be recorded on tapes or transmitted to the ground. Additional information about low-rate data and high-rate data follows:

Low rate ($\leq 1 \text{ Mb/sec}$)

• Remote Acquisition Units (RAU's) receive experiment data and deliver them to the Spacelab data management system for computer processing display, storage and/or transmission to the ground. RAU's may be mounted in equipment racks, under the Spacelab floor, and inside an Airlock.

- RAU's provide an 8 bit resolution, analog-to-digital conversion capability, and can accept serial digital data.
- The RAU data acquisition function is under experiment-supplied software control
- Signal conditioning equipment will be available, as required, to interface experiment hardware with the RAU
- A data processing computer (Mitra 125 S, 64K core, 16 bitwords, 3.5×10^5 operations/sec) which can analyze experiment data onboard the spacecraft, format data for on-board display, and format data for transmission to the ground will be on-board.
- A mass memory unit for storage of software will be on-board
- A data display unit and keyboard which permit on-board review of experiment data will be available on-board.
- Data may be stored on tape or transmitted to the ground at up to 64 Kb/sec.

High rate (62.5 Kb/sec to 16 Mb/sec)

- On-board data recording and transmission to the ground will be provided.
- Experiment data can be automatically annotated with time references, and digitized voice signals via the high data rate system.

Voice Data and Communications

The spacecraft provides voice recording, voice communications between the spacecraft and the ground, and an intercom system within the spacecraft.

Analog Data

Investigations which produce analog data will have two basic options. If feasible the analog data should be digitized, to take full advantage of the Spacelab Module's extensive digital data recording processing, and transmission capabilities.

Analog data, that cannot be digitized, may be recorded on-board and/or transmitted to the ground. Planned capabilities include:

- Analog to digital conversion. Data may be stored, transmitted to the ground, or processed and displayed on-board the spacecraft. Conversion rates are programmable at 1,10, and 100 samples/sec.
- Analog-down-link transmission for up to 85% of the mission. Data may be real-time or tape records $(3 \text{ H}_{\mathbb{Z}} \text{ to } 4.2 \text{ Mh}_{\mathbb{Z}})$.

Video Data

An Orbiter-to-Spacelab interface has been provided that would enable the collection monitoring and recording of black-and-white and color video signals. Also, blackand-white video data could be transmitted to the ground, one channel at a time, through this interface. Spacelab video capabilities currently under consideration include:

- Black-and-white video cameras
- Black-and-white video monitor
- Black-and-white video tape recorder plus tapes
- Camera Control Provides automatic and manual camera/recorder/ monitor switching and tags video records with time and experiment identification.
- Camera Timer Provides automatic time-controlled activation/ deactivation of cameras and recorders.
- Accessories Mounting brackets, lens assortments, cabling, and remote controls.
- Color video camera
- Color video montior
- Color video tape recorders plus tapes
Table B-1 summarizes the Spacelab module environment that the fluid behavior experiment designs will consider.

```
Table B-1 Spacelab Module Interior Environment
 Gravity
    On-orbit operations: 10^{-4}_{-3g} to 10^{-5} (spacecraft drifting)
10^{-3}g (spacecraft maneuvering)
    Other phase_. 3g (maximum, launch); 1.5g (maximum, reentry)
 Acoustic
    On-orbit operations: < NC-50 curve; overall noise level < 75 db
                              Incidental equipment-produced noise is dependent on equipment
                                 included in the final payload complement.
    Other phases: 137 db maximum - during launch (ref: 20 uN/m<sup>2</sup>)
 Vibration
    On-orbit operations: Incidental equipment produced vibration is dependent on
                                equipment included in the final payload complement.
    Other phases: Launch and ascent

    Sinusodial + 0.25g (5-35 Hz) - System level
    Random - 3.3g RMS (composite) - Equipment racks-4.77g RMS

                                     (composite) - floor mounted. For ~6 seconds after
                                     engine ignition.
Temperature
    On-orbit: (Adjustable) Min = 18^{\circ} \pm 1^{\circ}C (~ 64.6^{\circ}F)
Max = 27^{\circ} \pm 1^{\circ}C (~ 80.6^{\circ}F)
    Other phases: Within the on-orbit range.
Humidity
    On-orbit: 25% to 70% RH (not adjustable).
    Other phases: Within the on-orbit range.
Atmospheric pressure
    On-orbit: Total pressures = 14.7 psia (02/N2)
                 Composition: 0_2 = 21 x
N<sub>2</sub> = 79 x
    Other phases: Approximately the same as on-orbit
Cleanliness
    On-orbit: Maintained by 280 micron filters
   Other phases: Most NASA operations - class 100,000
Radiated Emissions
                Narrow band - 0.1 V/m (peak at 1 to 10 MHz & S-band)
   On-orbit:
                 Broad Band - 90db µV/m/MHz (peak in 100 MHz range)
                   This is the estimated upper limit.
   Other phases: Not presently available.
Magnetic Environment (AC)
   On-orbit: 146 db above a pico-tesla at 30 Hz, decreasing linearly to 80 db above
                   a pico-tesla at 50 KHz.
   Other phases: Within the on-orbit range.
Radiation
   On-orbit: Exposure to cosmic and trapped radiation, and solar flare particle fluxes through a minimum shielding of 0.45 g/cm<sup>2</sup>. Actual shielding depends on location and configuration of equipment.
Lighting
   On-orbit: 200-300 lumens/meter<sup>2</sup>, but increases to 400-600 lumen/meter<sup>2</sup> at Spacelab
                   workbenches.
```

Other phases: Lights will be turned off.

National Aeronautics and Space Administration

Participation of the second second

1. A. A.

Lewis Research Center 21000 Brookpark Road Cleveland, Ohio 44135 M.S. Copies Atm: L. E. Light 500-306 1 501-5 5 E. A. Bourke Tech. Utilization Office 7-3 1 Tech. Report Control Office 5-5 1 AFSC Liaison Office 501-3 2 2 60 - 3Library Office of Rel. & Qual. Assur. 500-211 1 N. T. Musial 500-318 1 E. P. Symons, Proj. Mgr. 501-7 20 T. H. Cochran 501-7 1 J. C. Aydelott 501 - 71 500-109 1 R. F. Lacovic L. J. Ross 500-113 1 E. J. Domino 501 - 41 National Aeronautics and Space Administration Headquarters Washington, DC 20546 Attn: RS/Director, Manned Space Technology 1 RP/Director, Space Propulsion & Power 1 RT/Director, Tech. Utilization Div. 1 RTP-6/F. W. Stephenson 1 MHE/P. N. Herr 1 National Aeronautics and Space Administration Goddard Space Flight Center Greenbelt, MD 20771 Attn: Library 1 National Aeronautics and Space Administration John F. Kennedy Space Center Kennedy Space Center, FL 32899 Atm: /Library 1 DD-MED-41/F. S. Howard 1 DE-A/W. H. Boggs 1 National Aeronautics and Space Administration Ames Research Center Moffett Field, CA 94035 Attn: Library 1 J. Vorreiter 244-7 1 National Aeronautics and Space Administration Langley Research Center Hampton, VA 23365 Attn: Library 1 249A C. T. D'Aiutolo 1 National Aeronautics and Space Administration

Johnson Space Center Houston, TX 77001 Attn: Library 1 EP2/Z. D. Kirkland 1 EP5/W. Chandler 1 ER/Hugh Davis 1 EP4/Dale Connelly 1 ER/M. Jones 1 PD13/James Thompson 1

National Aeronautics and Space Administration George C. Marshall Space Flight Center Huntsville, AL 35812 Copies <u>M.S.</u> 1 Attn: Library 1 EP43/L. Hastings EP43/G. Young 1 1 EP45/Dr. Wayne Littles 1 EP24/K. B. Chandler ES24/E. W. Urban 1 1 EP43/Eric Hyde Jet Propulsion Laboratory 4800 Oak Grove Drive Pasadena, CA 91103 Attn: Library 1 125-224 Don Young 1 NASA Scientific & Technical Information Facility P.O. Box 8757 Balt/Wash. International Airport Attn: Accessioning Department 10 Defense Documentation Center Cameron Station - Bldg. 5 5010 Duke Street Alexandria, VA 22314 Attn: TISIA 1 National Aeronautics and Space Administration Flight Research Center P.O. Box 273 Edwards, CA 93523 1 Attn: Library Air Force Rocket Propulsion Laboratory Edwards, CA 93523 Attn: LKCC/J. E. Brannigan 1 LKDS/R. L. Wiswell 1 Aeronautical Systems Division Air Force Systems Command Wright Patterson Air Force Base Dayton, OH Attn: Library 1 Air Force Office of Scientific Research Washington, DC 20333 1 Attn: Library Aerospace Corporation 2400 E. El Segundo Blvd. Los Angeles, CA 90045 Attn: Library - Documents 1 Arthur D. Little, Inc. 20 Acorn Park Cambridge, MA 02140 Attn: Library 1 Beech Aircraft Corporation Boulder Facility Box 631 Boulder, CO Attn: Library 1

Bel	ll Aerosystems, Inc.			Space Division	
Boj				Rockwell International Corporation	
BU	1810, NY 14240	<u>M.S.</u>	Copies	12214 Lakewood Blvd.	
Ate	n: Library		1	Downey, CA 90241 MS	Contes
	L. Inompson		1	Attn: Library	1
Boe	ing Company			J. Nichols	1
Sna	ce Division			A. Jones	1
P.C	D. Boy 868				•
Sea	ttle. WA 98124			Northrop Research and Technology Center	
Attr	1: Library			1 Research Park	
			L	Palos Verdes Peninsula, CA 90274	
Chr	vsler Corporation			Attn: Library	1
Spac	ce Division				
P.0	. Box 29200			TRW Systems, Inc.	
New	Orleans, LA 70129			1 Space Park	
Attn	: Library		1	Redondo Beach, CA 90278	
	·		•	Attn: Tech. Lib. Doc. Acquisitions	1
McE	Donnell Douglas Astronau	tics Co.			
5301	Balsa Avenue			National Science Foundation, Engr. Div.	
Hunt	tington Beach, CA 92647	,		1800 G. Street, NW	
Attn	: Library		1	washington, DC 20540	
	E. C. Cady		1	Aton: Library	1
			-	The set do to the set of the set of the	
Gene	eral Dynamics Convair U	Vvision		Florida institute of Technology	
P.O.	. Box 80847			Me Department	
San 1	Diego, CA 92138			Atta De T B B	
Attn	: Library	40-6540	1	Aun: Dr. T. E. Bowman	1
	R. D. Bradshaw	22-6960	1		
				BO Bon 200	
Missiles and Space Systems Center				Princeton NY 00540	
General Electric Company				Attra Ma Dantal Dat	
Valle	y Forge Space Technolo	gy Center		Aun: Mr. Daniel Balzer	1
P.O.	Box 8555			Southwest Personal Tradition	
Phila	delphia, PA 19101			Dont of Marken institute	
Attn:	Library		1	P.O. Drewer Solito	
				F.O. Drawer 28510	
Grumman Aircraft Engineering Corporation				Atta H Name the	
Bethp	bage, Long Island, NY			Franklin Deduc	1
Atta:	Library		1	Frankfin Docge	1
				Tufta University	
III R	esearch Institute			Mechanical Engineering Dept	
Techr	lology Center			Medord, MA 02155	
Cnica	go, 1L 60616			Atin: Dr. Lloyd Trefethen	
Attn:	Library		1		1
T caleb		_		McDonnell Douglas Astronautics Co Front	
DO	Bow 504	Company		P.O. Box 516	
F. U.	DUX 040			St. Louis. MO 63166	
Atta	Tibuom			Attn: G. Orton	
MP444	C D Planell		1	W. Regnier	1
	S. G. DaBrock		1		T
	S. G. DEDIOCK		1	Xerox Electro Optical Systems	
Linde, Div. of Union Carbida			300 North Halstead		
P.O. Box 44				Pasadena, CA 91107	
Tonew	anda. NV 11450			Attn: Robert Richter	1
Attn:	G. Nies				-
			1	Science Applications, Inc.	
Denver Division			1200 Prospect Street		
Martin-Marietta Cornovation				P.O. Box 2351	
P.O. 1	Box 179			La Jolla, CA 92037	1
Denver	. CO 80201				-
Attn	Library			Heat Transfer & Thermodynamic Lab.	
•	D. Fester		1	University of Michigan	
	J. Tegart		1	Ann Arbor, MI 48107	
	A. Villara		1	Attn: Dr. John A Clark	1
			L		-

•

3

. . Heat Transfer and Thermodynamic Lab. University of Michigan Copies M.S. Ann Arbor, MI 48107 1 Attn: Dr. John A. Clark Dr. Edward B. Igenbergs 8000 Muchen 80 Rauchstrasse 3 1 Germany Office National D'Etudes Et De **Recherches** Aerospatiales 29, Avenue de la Division LeClerc 92 Chatillon France 1 Par Jean Maulard Michel Delattre 16827 Oak View Dr. Encino, CA 91436 1 Attn: Dr. Marvin Adelberg Boeing Acrospace P.O. Box 3999 Seattle, QA 98124 1 Attn: C. L. Wilkensen 8K/31

J

A ORACINA STREET

and the second

1